Einstein-like geometric structures on surfaces

DANIEL J. F. FOX

Abstract. An AH (affine hypersurface) structure is a pair comprising a projective equivalence class of torsion-free connections and a conformal structure satisfying a compatibility condition which is automatic in two dimensions. They generalize Weyl structures, and a pair of AH structures is induced on a co-oriented nondegenerate immersed hypersurface in flat affine space. The author has defined for AH structures Einstein equations, which specialize on the one hand to the usual Einstein-Weyl equations and, on the other hand, to the equations for affine hyperspheres. Here these equations are solved for Riemannian signature AH structures on compact orientable surfaces, the deformation spaces of solutions are described, and some aspects of the geometry of these structures are related. Every such structure is either Einstein-Weyl (in the sense defined for surfaces by Calderbank) or is determined by a pair comprising a conformal structure and a cubic holomorphic differential, and so by a convex flat real projective structure. In the latter case it can be identified with a solution of the Abelian vortex equations on an appropriate power of the canonical bundle. On the cone over a surface of genus at least two carrying an Einstein AH structure there are Monge-Ampère metrics of Lorentzian and Riemannian signature and a Riemannian Einstein Kähler affine metric. A mean curvature zero space-like immersed Lagrangian submanifold of a para-Kähler four-manifold with constant para-holomorphic sectional curvature inherits an Einstein AH structure, and this is used to deduce some restrictions on such immersions.

Mathematics Subject Classification (2010): 53A15 (primary); 53C25, 53A30, 57N16, 57M50 (secondary).

1. Introduction

In [30] the author defined a class of geometric structures, called **AH** (*affine hypersurface*) structures, which generalize both Weyl structures and the structures induced on a co-oriented non-degenerate immersed hypersurface in flat affine space by its second fundamental form, affine normal and co-normal Gauß map, and defined for these AH structures equations, called **Einstein**, specializing on the one hand to the usual Einstein-Weyl equations and, on the other hand, to the equations for an affine hypersphere. In the present paper these equations are solved on

Received January 10, 2011; accepted in revised form August 30, 2011.

compact orientable surfaces and some aspects of their geometry in this case are described. It is hoped that, aside from their interest as such, the results provide motivation for studying higher dimensional AH structures.

1.1. An **AH structure** on an *n*-manifold is a pair $([\nabla], [h])$ comprising a projective equivalence class $[\nabla]$ of torsion-free affine connections and a conformal structure [h] such that for each $\nabla \in [\nabla]$ and each $h \in [h]$ there is a one-form σ_i such that $\nabla_{[i}h_{j]k} = 2\sigma_{[i}h_{j]k}$, or, what is the same, the completely trace-free part of $\nabla_{ik} h_{ik}$ vanishes (most notational and terminological conventions can be found in Sections 2 and 4). When n = 2 this compatibility condition is automatic; any pair ($[\nabla], [h]$) is AH. On the one hand, an AH structure for which $\nabla_i h_{jk}$ is pure trace for any $\nabla \in [\nabla]$ and any $h \in [h]$ is simply a Weyl structure (what is usually called the Weyl connection is the **aligned** representative $\nabla \in [\nabla]$ distinguished by the requirement that $h^{pq} \nabla_p h_{iq} = n h^{pq} \nabla_i h_{pq}$ for any $h \in [h]$). On the other hand, there is induced on any non-degenerate co-orientable immersed hypersurface in flat affine space a pair of AH structures. Namely, for both, [h] is generated by the second fundamental form, while the projective structures are those induced via the affine normal bundle and the conormal Gauß map. There is a canonical duality associating to each AH structure ($[\nabla], [h]$) a conjugate AH structure ($[\nabla], [h]$) having the same underlying conformal structure. The Weyl structures are exactly the self-conjugate AH structures, and the two AH structure induced on an affine hypersurface are conjugate in this sense. It may be helpful to think of the formalism of AH structures as giving an instrinsically formulated generalization of the geometry of hypersurfaces in flat affine space in a manner similar to how CR structures abstract the geometry of a pseudoconvex real hypersurface in flat complex Euclidean space. As for CR structures, the generalization is genuine; there are local obstructions to realizing a given AH structure as that induced on a hypersurface in flat affine space.

The specialization to surfaces of the notion of Einstein AH structure defined in [30] says that an AH structure on a surface is **Einstein** if there vanishes the tracefree symmetric part of the Ricci curvature of its aligned representative ∇ , while the scalar trace of its Ricci curvature satisfies the condition (6.1), generalizing the constancy of the scalar curvature of an Einstein metric, and taken by D. Calderbank in [13–15] as the definition of a two-dimensional Einstein-Weyl structure. Calderbank's point of view in the two-dimensional case was motivating for the definition of Einstein AH equations in general. In all dimensions the notion of Einstein AH structure has the following properties: for Weyl structures it specializes to the usual Einstein-Weyl equations; the conjugate of an Einstein AH structure is again Einstein AH; and the AH structures induced on a hypersurface in affine space are Einstein if and only if the hypersurface is an affine hypersphere.

There seem to be two principal reasons for interest in these equations. On the one hand, in Section 7 of [30] or [32] there are given examples of Einstein AH structures in dimension 4 and higher which are neither Weyl nor locally immersable as non-degenerate hypersurfaces in flat affine space, though otherwise as nice as possible (in the terminology of [30], they are exact, with self-conjugate curvature). This means the Einstein AH equations are a genuine generalization of the Einstein-Weyl and affine hypersphere equations. On the other hand, via the theorem of Cheng-Yau associating an affine hypersphere to the universal cover of a manifold with strictly convex flat projective structure, methods for solving the Einstein AH equations should lead to analytic methods for producing convex flat projective structures. The two-dimensional case studied here illustrates that class of AH structures is rich yet sufficiently limited as to be amenable to characterization.

The primary purpose of the present paper is to describe the classification up to the action of the group of orientation-preserving diffeomorphisms isotopic to the identity of the Riemannian signature Einstein AH structures on a compact orientable surface. It turns out that these all are either Einstein-Weyl structures or have underlying projective structure which is flat and convex. Such a structure is determined in the former case by a conformal structure and a holomorphic vector field the real part of which is Killing for some metric in the conformal class, and in the latter case by a conformal structure and cubic holomorphic differential, and, consequently, by results going back to C. P. Wang's [75] and E. Calabi's [12] and completed by F. Labourie's [49] and J. Loftin's [51–53], by a convex flat real projective structure. In the latter case it can be viewed as a solution of the Abelian vortex equations. Precise statements are given later in the introduction.

For either Einstein-Weyl structure or convex flat projective structures on surfaces the relevant classifications were already understood. The description of the deformation space of strictly convex flat projective structures is due independently to F. Labourie and J. Loftin, while the description of Einstein-Weyl structures on surfaces is mostly completed in the papers [13, 15] of Calderbank. They are recounted here in part to show concretely how they fit into the formalism used here, and in part to highlight the relation with the Abelian vortex equations, which appears not to have been noted previously. The main novelty is the point of view, that there is a common framework encompassing both kinds of structures. The picture that emerges, and is suggestive of what might be true in higher dimensions, is roughly that for Einstein AH structures, positive curvature implies Weyl while negative curvature implies flatness and convexity of the underlying projective structure. The situation recalls that for extremal Kähler metrics. If there are no holomorphic vector fields then the scalar curvature of an extremal Kähler metric must be constant; the analogous statement here is that an Einstein AH structure on a compact Riemann surface admitting no holomorphic vector fields must be exact, and, as a consequence, have underlying flat projective structure which is strictly convex. On the other hand, if there are holomorphic vector fields, then there can be extremal Kähler metrics with nonconstant scalar curvature, as was shown by E. Calabi in [10]; the analogous examples here are just the Einstein-Weyl structures on spheres and tori.

The final sections show that Einstein AH structures arise naturally in at least two other contexts, namely in the construction of Hessian metrics, and on mean curvature zero Lagrangian submanifolds of para-Kähler manifolds. In the penultimate section it is shown that the trivial real line bundle over a surface admitting an Einstein AH structure with parallel negative scalar curvature admits several interesting Hessian metrics of Riemannian and Lorentzian signatures which satisfy various Einstein type conditions. In the final section it is shown that an Einstein AH structure is induced on a mean curvature zero Lagrangian submanifold of a para-Kähler space form.

The remainder of the introduction describes the contents in more detail.

1.2. Section 2 describes background needed in the remainder of the paper. The reader is advised to read it enough to be familiar with the notational and terminological conventions employed throughout. Sections 4 and 5 describes the basic properties and local curvature invariants of AH structures on surfaces. As for a Riemann surface, the geometric structures considered admit equivalent descriptions in one-dimensional holomorphic terms or two-dimensional smooth real terms. The complex description generally leads to a more efficient and more transparent description, while the real description is more natural for comparing the two-dimensional results to the higher-dimensional case. Here both, and the transition from one to the other, are described, relying on material recounted in Section 3 relating holomorphic differentials with conformal Killing and Codazzi tensors.

1.3. In Section 6 the Einstein AH equations are defined and their most basic properties are noted. The defining conditions are given several reformulations. In particular, Lemma 6.6 shows that a Riemannian AH structure on an oriented surface is Einstein if and only if its Ricci curvature has type (1, 1) and its complex weighted scalar curvature (defined in Section 5.2) is holomorphic.

1.4. In order to state the main result of Section 7 it is necessary to recall some definitions. An open subset of the projective sphere is **convex** if its intersection with any projective line is connected. It is **properly convex** if its closure contains no pair of antipodal points. A properly convex domain is **strictly convex** if its boundary contains no open segment of a projective line. For example the positive orthant in \mathbb{R}^n (which is projectively equivalent to a standard simplex) is properly convex if its developing map is a diffeomorphism onto a (strictly) properly convex subset of the projective sphere.

Perhaps the principal technical result in the paper is Theorem 7.7 which describes what are the possible types of Einstein AH structure on a compact orientable surface in terms of the genus of the surface, the sign of the weighted scalar curvature, the exactness or not of the AH structure, and whether the AH structure is Weyl or not. The precise statement is long, so is not repeated here, but roughly the result is that a Riemannian Einstein AH structure ($[\nabla], [h]$), with conjugate ($[\nabla], [h]$), on a compact orientable surface M of genus g satisfies one of the following mutually exclusive possibilities:

- (1) ($[\nabla]$, [h]) is exact and Weyl, or, what is the same, $[\nabla]$ is generated by the Levi-Civita connection of a constant curvature metric representative of [h].
- (2) The genus is $g \ge 2$. ($[\nabla], [h]$) is exact and not Weyl with parallel negative

scalar curvature, $[\nabla]$ and $[\overline{\nabla}]$ are strictly convex flat real projective structures, the cubic torsion is the real part of a holomorphic cubic differential, and a distinguished metric has negative scalar curvature.

- (3) *M* is a torus. ([∇], [*h*]) is exact and not Weyl with parallel negative scalar curvature, [∇] and [∇̄] are flat real projective structures which are convex but not strictly convex, the cubic torsion is the real part of a holomorphic cubic differential, and a distinguished metric is flat.
- (4) *M* is a torus, ([∇], [g]) is Weyl and closed but not exact, the scalar curvature is zero, and the (1, 0) part of the aligned representative ∇ ∈ [∇] is a holomorphic affine connection.
- (5) *M* is a torus, $([\nabla], [h])$ is Weyl and not closed, and the scalar curvature changes signs.
- (6) *M* is a sphere, $([\nabla], [h])$ is Weyl and not closed, and the scalar curvature is somewhere positive.

In particular an Einstein AH structure on M which is not simply that generated by a conformal structure is either Weyl or exact, but not both. The proof of Theorem 7.7 is based on Theorem 7.1 which shows that for a distinguished metric representing the conformal structure the Einstein equations have a technically convenient form. Although the existence of this distinguished metric is a consequence of the Hodge decomposition, it is said to be **Gauduchon** because in the higher-dimensional setting the corresponding construction in the context of Einstein-Weyl structures is due to P. Gauduchon; see [34,35].

Theorem 7.5 shows that for an Einstein-Weyl structure on a sphere or torus the integral curves of the metric dual of the Faraday primitive of a Gauduchon metric h are magnetic geodesics for the magnetic flow generated by h and a scalar multiple of the Faraday two-form. This interpretation of Einstein-Weyl structures seems to be new.

1.5. In Section 8 it is shown that an exact Einstein AH structure on a compact orientable surface is equivalent to a special sort of solution of the Abelian vortex equations. Recall that a triple (∇, g, s) comprising a Hermitian structure g on a complex line bundle \mathcal{V} , a Hermitian connection ∇ on \mathcal{V} , and a smooth section s of \mathcal{V} solves the abelian vortex equations if ∇ induces a holomorphic structure on \mathcal{V} with respect to which s is holomorphic, and there is satisfied a third equation relating the curvature of ∇ and the Hermitian norm of s (see (7.12)). This extension to compact Kähler manifolds of equations proposed by Landau and Ginzburg to model superconductivity was first studied by M. Noguchi in [61] and S. Bradlow in [5]. A special class of solutions, the *p*-canonical solutions, is distinguished by the requirement that the resulting s be a section of the pth power of the canonical bundle with respect to the underlying complex structure, and that ∇ be the Hermitian connection induced by the underlying Kähler structure. The Abelian vortices corresponding to exact Einstein AH structures are restricted in the sense that they correspond to 3-canonical solutions. This gives a geometric interpretation to 3canonical solutions of the Abelian vortex equations which appears not to have been

previously observed. It should be interesting to understand exactly how this plays out at the level of moduli spaces, though note that diffeomorphism inequivalent Einstein AH structures can determine gauge equivalent Abelian vortices, so at this level the correspondence is neither injective nor surjective. See the final paragraphs of Section 8 for a brief discussion.

1.6. Section 9 is devoted mainly to showing constructively that all of the possibilities identified in Theorem 7.7 are realized and to describing the moduli/deformation spaces of solutions.

Combining Theorems 7.7 and 9.7 leads to the following theorem, the proof of which is completed in Section 11.8.

Theorem 1.1. On a compact orientable surface M of genus at least 2, the following spaces are in canonical bijection:

- (1) The fiber bundle over Teichmüller space of M the fibers of which comprise the cubic holomorphic differentials.
- (2) The deformation space of convex flat real projective structures.
- (3) The deformation space of Einstein AH structures.

The same equivalences are true for M of genus 1 provided that (3) is replaced by

(4) The deformation space of exact Einstein AH structures.

Aside from its interest as such, Theorem 1.1 is suggestive of what can be expected to be true about the corresponding structures in higher dimensions. It plays for Einstein AH structures a role something like the uniformization theorem plays in the theory of higher-dimensional Einstein metrics.

The main point of Section 9 is to prove directly the implication $(1) \Rightarrow (3)$ of Theorem 1.1, which is Theorem 9.7. The equivalence (1) \iff (2) (and also, implicitly, the implication $(2) \Rightarrow (3)$) of Theorem 1.1 was known already, having been proved independently by F. Labourie (see [49]) and J. Loftin (see [51]). The proof given here of Theorem 9.7 is not in essential points different from Loftin's proof of the implication $(1) \Rightarrow (2)$, but is set in a slightly more general context so as to yield Corollary 9.5, which shows how to construct certain p-canonical solutions of the Abelian vortex equations. Also there are extracted some bounds on the solutions which yield bounds on the volume and curvature of distinguished metrics of Einstein AH structures. These represent very preliminary steps in the direction of understanding the analogues in the present context of Teichmüller curves. As is explained in Section 3.6 there is a natural diffeomorphism equivariant action of $GL^+(2,\mathbb{R})$ on the space of cubic holomorphic differentials coming from its action on the singular flat Euclidean structure determined by such a differential on the complement of its zero locus. Theorem 1.1 means that the orbits of this action on the space of cubic holomorphic differentials determine disks in the deformation space of Einstein AH structures. An analysis of the structure of these disks awaits, but in Section 9 there are proved some results about the path in the deformation

space corresponding to a ray in the space of cubic holomorphic differentials. Theorem 9.9 shows that with a suitable parameterization of the ray the conformal factor relating the Gauduchon metric at time t to the Gauduchon metric at time 0 is pointwise non-decreasing and Lipschitz in t. This makes possible some statements about the limiting behavior along the ray of the volume and curvature of suitably scaled Gauduchon metrics.

In order to say a bit more about the background to Theorem 1.1, some context is recalled. In [25], S. Choi and W. M. Goldman showed that the deformation space of convex flat real projective structures on a two-dimensional orbifold M of negative orbifold characteristic $\chi(M)$ is homeomorphic to a cell of a dimension equal to $-8\chi(M) + k$, where k is a number expressible in terms of the orders of the stabilizers of the singular points; this generalizes an earlier theorem of Goldman in [36] for the manifold case. It follows from a theorem of Thurston that a compact 2-orbifold of non-positive orbifold Euler characteristic is a quotient of a manifold by a finite group; see the end of [25, Section 1.2]. In [75], C. P. Wang shows how a hyperbolic affine hypersphere gives rise to a conformal structure and a cubic holomorphic differential, and conversely, how given such data on a compact oriented surface there is associated to its universal cover a hyperbolic affine hypersphere. On the affine hypersphere over the universal cover of M the difference tensor of the affine connection induced via the affine normal and the Levi-Civita connection of an equiaffine metric is the real part of a holomorphic cubic differential (it is often called the *Pick form*). This observation underlies the Labourie-Loftin theorem.

The content of the implication $(2) \Rightarrow (3)$ of Theorem 1.1 is the claim that the Einstein AH structure is determined by its underlying projective structure (which is necessarily flat by Lemma 6.4). As is described briefly now, and in more detail in Section 11.8, this can be deduced from various results of S. Y. Cheng and S. T. Yau. By a theorem of Cheng and Yau (see [55] for the full history) resolving a conjecture of E. Calabi, the interior of the cone over a properly convex domain admits a unique foliation by hyperbolic affine hyperspheres asymptotic to the cone. In particular, this can be applied to the cone over the universal cover of a 2-orbifold M carrying a convex flat real projective structure, and in the manifold case the AH structure induced on the affine hypersphere descends to M. Since these AH structures are always exact, this means M carries a canonical homothety class of metrics (those induced by the equiaffine metrics on the affine hyperspheres foliating the interior of the cone) and so a distinguished connection, the Levi-Civita connection of any one of these metrics. Technically, this has two aspects. One is that an immersed hyperbolic affine hypersphere is properly embedded if and only if the induced affine metric is complete; for references to a proof see [71, Section 5]. The other is that a convex flat real projective structure determines a distinguished conformal structure. The latter can be obtained from either of two theorems of Cheng and Yau solving certain Monge-Ampère equations. In [21], it is shown that on a bounded convex domain $\Omega \subset \mathbb{R}^n$ there is a unique smooth, convex solution of the equation u^{n+2} det Hess $u = (-1)^n$ vanishing on the boundary of Ω . The radial graph of uis the desired affine hypersphere (see [52, Theorem 3]). Alternatively, in [23] it is shown that on a convex cone $\Omega \subset \mathbb{R}^{n+1}$ containing no complete affine line there

is a unique smooth function F solving det Hess $F = e^{2F}$, tending to $+\infty$ at the boundary of the cone, and such that Hess F is a complete Riemannian metric on the interior of the cone; the level sets of F are the desired affine hyperspheres asymptotic to the cone (the uniqueness claim follows by passing to the corresponding Kähler metric on the tube domain over Ω and appealing to the Schwarz lemma for volume forms in [60]). Because this approach does not appear to have been written anywhere, it is sketched in Section 11. These theorems of Cheng-Yau should be viewed as real analogues of the theorems of Cheng, Mok and Yau producing complete Kähler Einstein metrics, *e.g.* [22,60]. (In fact the latter theorem follows from the specialization to a tube domain over a pointed convex cone of the theorem of N. Mok and Yau in [60] producing a Kähler Einstein metric on a bounded domain of holomorphy). For further background the reader can consult, in addition to papers already mentioned, [21,24,53,55], and [6,8,9,12]. In [49], Labourie has shown that in two dimensions these results admit more direct and simpler proofs, and has given a variety of other ways of understanding them.

1.7. The remaining Einstein AH structures on compact orientable surfaces are all Weyl and occur on either the torus or the sphere. The existence of such structures in the case of zero scalar curvature (on the torus) is straightforward, and they correspond in a natural way to holomorphic affine connections. This is explained in Section 10.7. The remaining cases are Einstein-Weyl structures which are not closed. Such a structure determines a vector field X which is Killing for the Gauduchon metric. It follows that X is the real part of a holomorphic vector field, though it is not an arbitrary holomorphic vector field, for X preserves some metric and so generates a circle action; in particular on the torus it must generate a rational flow. Using this circle action the equations that need to be solved to construct an Einstein-Weyl structure are reduced to an ODE, which after further reductions admits explicit solutions in terms of elementary functions. In the case of the sphere, the moduli space of Einstein-Weyl structures is essentially parameterized by conjugacy classes of elliptic elements of $PSL(2, \mathbb{C})$, that is by a half-open interval. The precise statement is Theorem 10.3. In the torus case, because of the aforementioned rationality condition, it is not clear how to describe nicely the deformation space.

Much of the description of Einstein-Weyl structures given in Section 10 can be found in some similar form in [13] and [15], although the presentation here is perhaps more elementary, and is made to illustrate the realization of the possibilities stated in Theorem 7.7; also the description of when the solutions found are equivalent is made more explicitly than it is in [13] and [15], although it seems that the results were understood by Calderbank.

1.8. In Section 11 it is shown that an Einstein AH structure $([\nabla], [h])$ on a compact orientable surface M of genus g > 1 gives rise to two metrics f_{IJ} and g_{IJ} on $M \times \mathbb{R}^{\times}$ which together with the flat affine structure induced on $M \times \mathbb{R}^{\times}$ by $[\nabla]$ constitute Hessian metrics. The metric g_{IJ} is a Lorentz signature Monge-Ampère metric, while f_{IJ} is a Riemannian signature Hessian metric which is Einstein when viewed as a Kähler affine metric in the sense of [23]. Precise statements are given in

Theorem 11.2. In particular, it is shown that, with respect to g_{IJ} , M is a smoothly immersed, space-like, umbilic hypersurface of constant mean curvature. In Section 11.6 is explained that the metric g_{IJ} can also be viewed as a solution of 2 + 1 gravitational equations with stress energy tensor of the form corresponding to a pressureless perfect fluid.

In Section 11.7 it is shown that for each C > 0 there is a Riemannian signature Monge-Ampère metric on $M \times [-\log C, \infty)$. Its potential has the form $\Psi(F)$ where F is the potential for the metric f_{IJ} described in the previous paragraph, and Ψ is the function given in (11.16). The precise statement is Theorem 11.3.

1.9. In [1], J. A. Aledo, J. M. Espinar, and J. A. Gálvez have studied a surface equipped with what they call a Codazzi pair, which is a pair comprising a Riemannian metric and a second symmetric covariant tensor satisfying the Codazzi equations with respect to the Levi-Civita connection of the metric. They view this as an abstraction of the geometric structure induced on a hypersurface in Euclidean three space, and have shown that many classical results can be strengthened in this context. Though the setting is different from that considered here, the perspective is similar. One of the motivations of [1] is that such Codazzi pairs arise in other ways, *e.g.* the real part of the Hopf differential of a harmonic mapping yields such a pair. Similarly AH structures on surfaces arise naturally in the study of submanifolds of para-Kähler manifolds. In Section 12 it is explained that a mean curvature zero space-like Lagrangian immersed submanifold of a para-Kähler manifold of constant para-holomorphic sectional curvature inherits an Einstein AH structure; see Theorem 12.2 and the final paragraph of Section 12.2. It is indicated also how to associate to certain Einstein AH structures such an immersion. I thank an anonymous referee for pointing out that essentially equivalent ideas have been worked out independently in R. Hildebrand's papers [38] and [39]. In particular Theorem 12.2 is equivalent to theorems in [38] modulo changes of terminology. Related constructions appeared already in L. Vrancken's [74]. Corollary 12.3 states some restrictions on mean curvature zero Lagrangian immersions in four dimensional para-Kähler space forms resulting from applying Theorem 7.7 to the induced Einstein AH structure. Closely related results about minimal Lagrangian immersions in complex hyperbolic space are the focus of the preprint [50] of J. Loftin and I. McIntosh.

2. Notation and terminology

This section records the notational and terminological conventions in use throughout the paper.

2.1. Throughout *M* is a connected smooth (C^{∞}) manifold. For a vector bundle *E*, $\Gamma(E)$ denotes the space of its smooth sections (even if *E* has a holomorphic structure). Its *k*th symmetric power and top exterior power are written respectively as $S^k(E)$ and Det *E*. For a line bundle *E*, |E| is the tensor product of *E* with

its orientation bundle. Sections of $|\operatorname{Det} T^*M|^{\lambda}$ are called λ -densities. A tensor taking values in $|\operatorname{Det} T^*M|^{\lambda}$ for some $\lambda \in \mathbb{R}^{\times}$ is said to be weighted. If M is compact the integral of a 1-density has sense, and so there is a bilinear pairing $\langle \cdot, \cdot \rangle : \Gamma(|\operatorname{Det} T^*M|^{\lambda}) \times \Gamma(|\operatorname{Det} T^*M|^{1-\lambda}) \to \mathbb{R}$ between densities of weight λ and the complementary weight $(1 - \lambda)$.

2.2. Given a complex structure $J \in \text{End}(\mathbb{V})$ on the real vector space \mathbb{V} , the complexification $\mathbb{V} \otimes_{\mathbb{R}} \mathbb{C}$ decomposes as the direct sum $\mathbb{V} \otimes_{\mathbb{R}} \mathbb{C} = \mathbb{V}^{1,0} \oplus \mathbb{V}^{0,1}$ of the $\pm i$ eigensubbundles of the extension of J to $\mathbb{V} \otimes_{\mathbb{R}} \mathbb{C}$ by complex linearity. The induced action of J on \mathbb{V}^* is defined by $J(\alpha) := \alpha \circ J$, so that (\mathbb{V}^*, J) is an a complex vector space, and the (1, 0) part $\mathbb{V}^{*1,0}$ of \mathbb{V}^* is the \mathbb{C} -dual of $\mathbb{V}^{1,0}$ and annihilates $\mathbb{V}^{0,1}$. A completely symmetric or anti-symmetric tensor decomposes by type. If B is in $S^{p+q}(\mathbb{V}^*)$ or $S^{p+q}(\mathbb{V})$ denote by $B^{(p,q)}$ the (p,q) part of its complex linear extension as an element of $S^{p+q}(\mathbb{V}^* \otimes_{\mathbb{R}} \mathbb{C})$. For example, for $\alpha \in \mathbb{V}^*$, $2\alpha^{(1,0)} = \alpha - iJ(\alpha) = \alpha - i\alpha \circ J$.

The preceeding makes sense fiberwise on a complex vector bundle. If (M, J) is a complex manifold, there is written $TM \otimes_{\mathbb{R}} \mathbb{C} = T^{1,0} \oplus T^{0,1}$, while the complex tangent bundle $T_{\mathbb{C}}M$ is TM viewed as a complex vector bundle; it is identified as a complex vector bundle with $T^{1,0}$. A **holomorphic structure** on a complex vector bundle E over a complex manifold is a linear differential operator \hat{D} sending E-valued (p,q)-forms to E-valued (p,q+1)-forms, satisfying the Leibniz rule $\hat{D}(fs) = \bar{\partial} f \wedge s + f \hat{D} s$ for any $f \in C^{\infty}(M)$ and any smooth section s of E, and such that $\hat{D}^2 = 0$. A smooth local section of E is \hat{D} -holomorphic if it is in ker \hat{D} . If (M, J) is a complex manifold then any bundle of complex tensors (a tensor product of powers of $T_{\mathbb{C}}M$ and its dual) over M is naturally a holomorphic vector bundle, and the corresponding holomorphic structure is denoted $\bar{\partial}$. The exterior differential decomposes by type as $d = \partial + \bar{\partial}$.

For a one-complex dimensional manifold and $p \in \mathbb{Z}$, define \mathcal{K}^p to be the *p*th power of the complex cotangent bundle $T_{\mathbb{C}}M^*$ viewed as a holomorphic line bundle, and viewed also as the (-p)th symmetric power of $T^{1,0}$ (a negative power means the power of the dual). A smooth (holomorphic) section of \mathcal{K}^p is called a complex (holomorphic) *p*-differential. In a local holomorphic coordinate *z*, a holomorphic section *p*-differential σ has the form $\phi(z)dz^p$ for a holomorphic function $\phi(z)$.

2.3. Tensors are usually indicated using the abstract index notation. Ordinary tensors are indicated by lowercase Latin abstract indices, so that, for instance, a_{ij} indicates a covariant two tensor. If a complex structure is given, lowercase Greek indices, *e.g.* α , β , γ , etc., decorate sections of the tensor powers of $T^{1,0}$ and its complex dual, while barred lowercase Greek indices, *e.g.* $\bar{\alpha}$, $\bar{\beta}$, $\bar{\gamma}$, etc., decorate sections of the tensor powers of indices in square brackets (respectively parentheses) indicates complete skew-symmetrization (respectively complete symmetrization), so that for example $a^{ij} = a^{(ij)} + a^{[ij]}$ indicates the decomposition of a contravariant two-tensor into its symmetric and skew-symmetric parts, and $(X \wedge Y)^{ij} = 2X^{[i}Y^{j]}$ for vector fields X and Y. The

summation convention is always in effect in the following form: indices are in either *up* position or *down* and a label appearing as both an up index and a down index indicates the trace pairing. Since polynomials on the vector space \mathbb{V} are tautologically identified with symmetric tensors on the dual vector space \mathbb{V}^* , the index *i* in $\frac{\partial}{\partial y_i}$ has to be regarded as an *up* index. The horizontal position of indices is always maintained when they are raised or lowered. The interior multiplication of a vector field X^i in a covariant tensor $B_{i_1...i_k}$ is defined by $\iota(X)B_{i_1...i_{k-1}} := X^p B_{pi_1...i_{k-1}}$.

2.4. The **curvature** $R_{ijk}^{\ l}$ of a torsion-free affine connection ∇ is defined by $2\nabla_{[i}\nabla_{j]}X^k = R_{ijp}^{\ k}X^p$. The **Ricci curvature** is the trace $R_{ij} := R_{pij}^{\ p}$.

2.5. A non-degenerate weighted covariant two-tensor h_{ij} determines a contravariant two-tensor h^{ij} of complementary weight defined by $h^{ip}h_{jp} = \delta_j^{i}$, in which here, as always, δ_i^{j} is the tautological $\binom{1}{1}$ -tensor determined by the pairing of vectors with covectors. By det *h* is meant the 2-density which satisfies $\langle \det h, E_1 \land \dots \land E_n \rangle = \det h(E_i, E_j)$ for any frame E_1, \dots, E_n .

A **pseudo-Riemannian metric** or, simply, a **metric** means a non-degenerate covariant two-tensor h_{ij} . The metric is **Riemannian** if it is positive definite. A **conformal structure** [h] means a pseudo-Riemannian metric determined up to multiplication by a positive function. A conformal structure is identified with its **normalized representative** $H_{ij} := |\det h|^{-1/\dim M} h_{ij}$ which takes values in the bundle of $-(2/\dim M)$ -densities.

For conformal metrics $\tilde{h}_{ij} = f h_{ij}$ the Levi-Civita connections are written \tilde{D} and D, and their difference tensor is written $\tilde{D} - D = 2\sigma_{(i}\delta_{j)}{}^{k} - h_{ij}h^{kp}\sigma_{p}$, for $2\sigma_{i} = d \log f_{i}$. The curvature of D is written $\mathcal{R}_{ijk}{}^{l}$. Objects corresponding to \tilde{D} are written with the same notations as those corresponding to D, but decorated with a \tilde{R} , although for the scalar curvatures it will be convenient to write $\mathcal{R}_{\tilde{h}}$ and \mathcal{R}_{h} rather than $\tilde{\mathcal{R}}$ and \mathcal{R} . For example, the scalar curvature changes under conformal rescaling by

$$f\mathcal{R}_{\tilde{h}} = \mathcal{R}_h - 2h^{pq}D_p\sigma_q = \mathcal{R}_h - \Delta_h\log f.$$
(2.1)

Here Δ_h is the rough Laplacian $h^{pq}D_pD_q$, which acts on tensors as well as on functions.

Given a metric *h*, for any $X_{i_1...i_k}{}^{j_1...j_l}$ the notations X^{\sharp} and X^{\flat} indicate the tensors obtained by raising and lowering all indices using *h*. That is $X_{i_1...i_{k+l}}^{\flat} = X_{i_1...i_k}{}^{j_1...j_l}h_{j_1i_{k+1}}\dots h_{j_li_{k+l}}$, and similarly for X^{\sharp} . The *h*-norm of a tensor *X* is defined by complete contraction, *e.g.* for $A_{ij}{}^k$, $|A|_h^2 = A_{ijk}^{\flat}A^{\sharp ijk} = A_{ij}{}^k A_{ab}{}^c h^{ia} h^{jb} h_{kc}$.

2.6. A **Riemann surface** is an oriented surface M equipped with a Riemannian signature conformal structure [h]. (Note that *Riemannian surface* and *Riemann surface* are not synonyms; the former indicates there is given a distinguished metric,

while the underlying smooth structure need not be orientable). On a Riemann surface there is a unique almost complex structure J defined in terms of any $h \in [h]$ by the requirements that $J_i {}^p J_j {}^q h_{pq} = h_{ij}$ and that the two-form $\omega_{ij} := J_i {}^p h_{pj}$ determine the given orientation. As these conditions do not depend on the choice of $h \in [h]$, J is determined by the conformal structure [h]. On the other hand, given a complex structure J on a surface M, the orientation determined by $X \wedge JX$ does not depend on the choice of X, and there is a unique Riemannian signature conformal structure [h] such that $J_i {}^p J_j {}^q h_{pq} = h_{ij}$. Any almost complex structure on a surface is integrable, or, equivalently DJ = 0, where D is the Levi-Civita connection of h, and so (h, J, D) is a Kähler structure on M with Kähler form ω_{ij} and associated **Hermitian metric** $h^{(1,1)}$. On any bundle E of complex tensors, hdetermines a Hermitian structure, and D induces the unique Hermitian connection such that $D^{0,1} = \overline{\partial}$.

2.7. For a Hermitian metric $h_{\alpha\bar{\beta}}^{(1,1)}$ it is convenient to omit the superscript (1, 1), and to write instead simply $h_{\alpha\bar{\beta}}$, although it should be kept in mind that h_{ab} and $h_{\alpha\bar{\beta}}$ refer to different objects. The dual bivector $h^{\alpha\bar{\beta}}$ is defined by $h^{\alpha\bar{\beta}}h_{\gamma\bar{\beta}} = \delta_{\alpha}^{\gamma}$. The conventions are such that $\delta_{\alpha}^{\ \beta}$ indicates the tautological endomorphism of $T^{1,0}$, while $\delta_i^{\ j}$ indicates the tautological endomorphism of TM. The convention is that for a section $X^{a_1...a_k}$ of $\otimes^k TM$, the Hermitian norm of its (k, 0) part $X^{\alpha_1...\alpha_k}$ is defined by complete contraction with its complex conjugate using $h_{\alpha\bar{\beta}}$; e.g. for $X \in \Gamma(TM), |X^{(1,0)}|_h^2 = X^{\alpha} X^{\bar{\beta}} h_{\alpha\bar{\beta}} = \frac{1}{2} |X|_h^2$.

2.8. The formal adjoint d_h^* of the exterior differential with respect to the pairing of forms given by *h* is given on *k*-forms by $d_h^* \alpha_{i_1...i_{k-1}} = -D^p \alpha_{pi_1...i_{k-1}}$. Since, for a *k*-form $\alpha_{i_1...i_k}$ there holds $f d_{\tilde{h}}^* \alpha_{i_1...i_{k-1}} = d_h^* \alpha_{i_1...i_{k-1}} + 2(k-1)\sigma^{\sharp p} \alpha_{pi_1...i_{k-1}}$, that a one-form be co-closed (in ker d_h^*) is a conformally invariant condition. For a one-form whether there is written $D^p \gamma_p$ or $-d_h^* \gamma$ will depend on context. On a Riemann surface the action of the Hodge star operator $\star = \star_h$ on one-forms depends only on the conformal class [h], and is given in terms of the complex structure *J* determined by [h] by $(\star \alpha)_i = -\alpha_p J_i^{p}$. On an oriented surface $d_h^* = -\star d\star$.

3. Holomorphic differentials and conformal Killing and Codazzi tensors

This Section 3 records some basic facts about holomorphic differentials on Riemann surfaces. The purely real equations for symmetric tensors characterizing conformal Killing and Codazzi tensors make sense in higher dimensions (see [30, Section 6]), but on surfaces their complex counterparts are easier to work with. Lemma 3.5 states the identification on Riemann surfaces of Codazzi and conformal Killing tensors as the real parts of holomorphic differentials, and Lemma 3.6 shows that such a differential determines a singular flat metric. For quadratic differentials these statements are well known and widely utilized. Although the general statements

are surely also well known, there does not seem to be any convenient reference, and so it seems useful to include them. The correspondence between holomorphic differentials and singular flat metrics is discussed in more detail in Sections 3.5 and 3.6, where it is explained how it yields a diffeomorphism equivariant action of $GL^+(2, \mathbb{R})$ on the space of such differentials. While not much use is later made of this material, it motivates some estimates proved in Section 9 and when coupled with Theorem 1.1 and compared with usual Teichmüller theory, suggests many questions.

3.1. To a Young diagram the boxes of which are labeled with distinct indices corresponds the irreducible $GL(n, \mathbb{R})$ module comprising tensors skew-symmetric in the indices in a given column of the Young diagram and vanishing when skew-symmetrized over the indices in a given column and any index in any box to the right of the given column. The irreducible representations of the subgroup CO(h) of $GL(n, \mathbb{R})$ acting conformally with respect to a fixed metric h on \mathbb{R}^n are described in [76]. The subspace of an irreducible $GL(n, \mathbb{R})$ representation comprising tensors completely trace-free with respect to h is a representation of CO(h). Lemma 3.1 will be invoked repeatedly.

Lemma 3.1 ([76], Theorem 5.7.A). The CO(h)-module of covariant trace-free tensors on \mathbb{R}^n having symmetries corresponding to a Young diagram is trivial if the sum of the lengths of the first two columns of the Young diagram is greater than n.

For instance, Lemma 3.1 implies that the usual conformal Weyl tensor of a Riemannian metric vanishes identically on a manifold of dimension at most 3.

If there is given a fiberwise metric on the vector bundle E then $S_0^k(E)$ denotes the subbundle of $S^k(E)$ comprising elements trace-free with respect to the given metric. The convention is that $S_0^1(E) = E$; this corresponds to regarding the trace as the zero map on vectors.

Lemma 3.2. Let h_{ij} be a constant metric on \mathbb{R}^2 , and for $k \ge 1$ let $A, B \in S_0^k(\mathbb{R}^2)$. *Then*

$$2A_{a_1\dots a_k(i}B_{j)}^{a_1\dots a_k} = A_{a_1\dots a_k}B^{a_1\dots a_k}h_{ij},$$
(3.1)

in which indices are raised and lowered with h_{ij} and its inverse h^{ij} . Let X^i be a vector. Then $2|\iota(X)B|_h^2 = |X|_h^2|B|_h^2$. In particular if h_{ij} has definite signature the equations $|B|_h^2|X|_h^2 = 0$ and $X^p B_{pi_1...i_{k-1}} = 0$ are equivalent.

Proof. The tensor

$$\omega_{ija_1...a_{k+1}} = h_{i(a_1}A_{a_2...a_k)j} + h_{j(a_1}A_{a_2...a_k)i} - h_{ij}A_{a_1...a_k} - h_{(a_1a_2}A_{a_3...a_k)ij}$$

is completely trace-free. As $\omega_{ija_1...a_k} = \omega_{(ij)(a_i...a_k)}$ and $\omega_{i(ja_1...a_k)} = 0$, it follows from Lemma 3.1 that $\omega = 0$. Hence $2A_{a_1...a_k}(iB_j)^{a_1...a_k} - A_{a_1...a_k}B^{a_1...a_k}h_{ij} = B^{a_1...a_k}\omega_{ija_1...a_k} = 0$. By (3.1) there holds $2|\iota(X)B|_h^2 = 2X^p B_{pi_1...i_{k-1}}X^q B_q^{i_1...i_{k-1}} = |X|_h^2|B|_h^2$. **3.2.** On a Riemannian manifold (M, h, D), a symmetric tensor $\sigma \in \Gamma(S^k(T^*M))$ is **Codazzi** if $D\sigma \in \Gamma(S^{k+1}(T^*M))$. From

$$(k+1)\left(D_{i}\sigma_{i_{1}...i_{k}}-D_{(i}\sigma_{i_{1}...i_{k}})\right)=2\sum_{s=1}^{k}D_{[i}\sigma_{i_{s}]i_{1}...i_{s}...i_{k}},$$
(3.2)

(in which a denotes the omission of the index), it is evident that σ is Codazzi if and only if $D_{[i_1\sigma_{i_2}]...i_{k+1}} = 0$. Write div_h (or simply div) for the divergence operator div_h(σ)_{i₁...i_k} := $D^p \sigma_{pi_1...i_k}$ determined by h. For any tensor σ let $f_h(\sigma)$ be its trace-free part. Let $\mathcal{L}_h : \Gamma(S^k(T^*M)) \to \Gamma(S^{k+1}(T^*M))$ be the formal adjoint of the composition div_h \circ f_h with respect to the pairing of sections of $S^{k+1}(T^*M)$ determined by integration. Explicitly, for $\sigma \in \Gamma(S_0^k(T^*M))$ and M a surface,

$$\mathcal{L}_{h}(\sigma)_{i_{1}\dots i_{k+1}} = D_{(i_{1}}\sigma_{i_{2}\dots i_{k+1})} - \frac{1}{2}h_{(i_{1}i_{2}}\operatorname{div}_{h}(\sigma)_{i_{3}\dots i_{k+1})}.$$
(3.3)

Lemma 3.3. On a Riemannian surface (M, h), for $\sigma \in \Gamma(S_0^k(T^*M))$ there hold

$$D_{i}\sigma_{i_{1}...i_{k}} = \mathcal{L}_{h}(\sigma)_{ii_{1}...i_{k}} + h_{i(i_{1}}\operatorname{div}_{h}(\sigma)_{i_{2}...i_{k}}) - \frac{1}{2}h_{(i_{1}i_{2}}\operatorname{div}_{h}(\sigma)_{i_{3}...i_{k}})_{i}, \quad if \, k > 1, \quad (3.4)$$
$$D_{i}\sigma_{j} = \mathcal{L}_{h}(\sigma)_{ij} + \frac{1}{2}d\sigma_{ij} + \frac{1}{2}\operatorname{div}_{h}(\sigma)h_{ij}, \quad if \, k = 1. \quad (3.5)$$

For k > 1, the following are equivalent for $\sigma \in \Gamma(S_0^k(T^*M))$:

1. σ is Codazzi; 2. σ is divergence free; 3. $D\sigma = \mathcal{L}_h(\sigma)$.

Proof. The tensor on the right-hand side of (3.4) is trace-free and its complete symmetrization vanishes, so it lies in the irreducible O(h)-module corresponding to a Young diagram with k - 1 boxes in its first row and 1 box in its second row. By Lemma 3.1 this module is trivial if k > 1. When k > 1, for $\sigma \in \Gamma(S_0^k(T^*M))$ it follows from (3.4) that $D\sigma \in \Gamma(S_0^{k+1})(T^*M)$ if and only if $\operatorname{div}_h(\sigma) = 0$. This also follows by tracing the identity

$$2D_{[a}\sigma_{b]i_{1}...i_{k-1}} = h_{a(i_{1}}\operatorname{div}_{h}(\sigma)_{i_{2}...i_{k-1})b} - h_{b(i_{1}}\operatorname{div}_{h}(\sigma)_{i_{2}...i_{k-1})a},$$
(3.6)

which also follows from Lemma 3.1. From (3.4) there follows $\operatorname{div}_h(\omega) = 0$ if and only if $D\sigma = \mathcal{L}_h(\sigma)$.

On a surface $f \operatorname{div}_{\tilde{h}}(\sigma) = \operatorname{div}_{h}(\sigma)$, so that the space ker $\operatorname{div}_{h} \cap \Gamma(S_{0}^{k}(T^{*}M))$ of trace-free Codazzi tensors depends only on the conformal class of h. As $\mathcal{L}_{e^{\phi}h}(e^{k\phi}\sigma)_{i_{1}\dots i_{k+1}} = e^{k\phi}\mathcal{L}_{h}(\sigma)_{i_{1}\dots i_{k+1}}$, the subspace $\Gamma(S_{0}^{k}(T^{*}M)) \cap \ker \mathcal{L}$ is conformally invariant. The operator $\mathcal{L}_{h}^{\sharp} : \Gamma(S_{0}^{k}(TM)) \to \Gamma(S_{0}^{k+1}(TM))$ defined by $\mathcal{L}_{h}^{\sharp}(X) = \mathcal{L}_{h}(X^{\flat})^{\sharp}$ satisfies $e^{\phi}\mathcal{L}_{e^{\phi}h}^{\sharp}(X) = \mathcal{L}_{h}^{\sharp}(X)$. For a vector field X there holds $2\mathcal{L}_{h}(X^{\flat})_{ij} = \operatorname{tf}(\mathcal{L}_{X}h)_{ij}$, which motivates the notation resembling that for the Lie derivative. Moreover, this shows that $\Gamma(TM) \cap \ker \mathcal{L}_{h}^{\sharp}$ comprises the conformal Killing fields. For this reason the sections of $\ker \mathcal{L}^{\sharp} \cap \Gamma(S_{0}^{k}(TM))$ are called **conformal Killing tensors**.

Lemma 3.5 explains the conformal invariance on surfaces of trace-free Codazzi and conformal Killing tensors by showing that they are exactly the real parts of holomorphic differentials.

3.3. An almost complex structure *J* on a real vector space \mathbb{V} determines an almost complex structure \mathbb{J} on $\otimes^k(\mathbb{V}^*)$ defined by $\mathbb{J}(B)_{i_1...i_k} = J_{i_1} {}^p B_{pi_2...i_k}$.

Lemma 3.4. Fix a two-dimensional real vector space \mathbb{V} with an almost complex structure J. Let k > 1. The map $B \to \mathbb{J}(B)$ is a complex structure on the 2dimensional real vector space $S_0^k(\mathbb{V}^*)$. For $B \in S_0^k(\mathbb{V}^*)$ there holds $2B^{(k,0)} = B - i\mathbb{J}(B)$, so that the (1,0) part of B qua element of $(S_0^k(\mathbb{V}^*), \mathbb{J})$ equals the (k, 0)part of B relative to J. There results a complex linear isomorphism between the complexification $S_0^k(\mathbb{V}^*) \otimes_{\mathbb{R}} \mathbb{C}$ and $\mathcal{K}^k \oplus \overline{\mathcal{K}}^k$ such that the $\pm i$ eigenspaces of \mathbb{J} on $S_0^k(\mathbb{V}^*) \otimes_{\mathbb{R}} \mathbb{C}$ are identified respectively with \mathcal{K}^k and $\overline{\mathcal{K}}^k$.

Proof. Let *h* be a definite signature metric on \mathbb{V} compatible with *J*. Let $B \in S^k(\mathbb{V}^*)$. If $X \in \mathbb{V}$ is non-zero, then $\{X, JX\}$ is an *h*-orthogonal basis of \mathbb{V} . From the evidently equivalent identities

$$\mathbb{J}(B)_{[i_1i_2]i_3...i_k} = J_{[i_1}{}^p B_{i_2]i_3...i_k p} = 0,$$

$$J_{i_1}{}^p J_{i_2}{}^q B_{pqi_3...i_k} = -B_{i_1...i_k}.$$
(3.7)

it follows that $B \in S_0^k(\mathbb{V}^*)$ if and only if $\mathbb{J}(B)$ is completely symmetric. Since $B = -\mathbb{J}(\mathbb{J}(B))$, the same statement with the roles of B and $\mathbb{J}(B)$ interchanged shows that in this case $\mathbb{J}(B) \in S_0^k(\mathbb{V}^*)$. Thus $B \in S_0^k(\mathbb{V}^*)$ if and only if $\mathbb{J}(B) \in S_0^k(\mathbb{V}^*)$. These conditions are obviously equivalent to the vanishing of $B^{(p,k-p)}$ whenever $0 . For <math>\beta \in \mathcal{K}^k$, $\mathbb{J}(\operatorname{Re} \beta) + i\mathbb{J}(\operatorname{Im} \beta) = \mathbb{J}(\beta) = i\beta = -\operatorname{Im} \beta + i\operatorname{Re} \beta$, so $\beta = \operatorname{Re} \beta - i\mathbb{J}(\operatorname{Re} \beta)$. Since β is symmetric so are $\operatorname{Re} \beta$ and $\mathbb{J}(\operatorname{Re} \beta)$, and hence, by the preceeding, $\operatorname{Re} \beta$ and $\operatorname{Im} \beta$ are in $S_0^k(\mathbb{V}^*)$.

Since \mathbb{J} preserves $S_0^k(\mathbb{V}^*)$, it is a complex structure on $S_0^k(\mathbb{V}^*)$. This means that $S_0^k(\mathbb{V}^*) \otimes_{\mathbb{R}} \mathbb{C}$ decomposes into (1, 0) and (0, 1) parts with respect to the action of \mathbb{J} and so for $B \in S_0^k(\mathbb{V}^*)$ it makes sense to speak of its (1, 0) part $\frac{1}{2}(B - i\mathbb{J}(B))$. For $Z \in \mathbb{V}^{0,1}$ there holds $\iota(Z)\mathbb{J}(B) = \iota(J(Z))B = -i\iota(Z)B$, so $\iota(Z)(B - i\mathbb{J}(B)) = 0$. Hence $B - i\mathbb{J}(B) \in \mathcal{K}^k$ if $B \in S_0^k(\mathbb{V}^*)$. The map $B \to B^{(k,0)}$ also sends $S_0^k(\mathbb{V}^*)$ to \mathcal{K}^k , and it is claimed that $2B^{(k,0)} = B - i\mathbb{J}(B)$. By the complete symmetry of B

there holds

$$2^{k}B^{(k,0)}(X,...,X) = B(X - iJX,...,X - iJX)$$

= $\sum_{s=0}^{k} (-i)^{s} {k \choose s} B(X,...,X,\underbrace{JX,...,JX}_{s \text{ times}})$
= $\sum_{s=0}^{\lfloor k/2 \rfloor} (-1)^{s} {k \choose 2s} B(X,...,X,\underbrace{JX,...,JX}_{2s \text{ times}})$
+ $i \sum_{s=1}^{\lfloor k/2 \rceil} (-1)^{s} {k \choose 2s-1} B(X,...,X,\underbrace{JX,...,JX}_{2s-1 \text{ times}}).$ (3.8)

Using the second equation of (3.7) in (3.8) yields

$$2^{k}B^{(k,0)}(X,...,X) = \left(\sum_{s=0}^{\lfloor k/2 \rfloor} {k \choose 2s}\right) B(X,...,X) - i\left(\sum_{s=1}^{\lfloor k/2 \rceil} {k \choose 2s-1}\right) \mathbb{J}(B)(X,...,X)$$
(3.9)
= $2^{k-1}B(X,...,X) - i2^{k-1}\mathbb{J}(B)(X,...,X).$

Polarizing (3.9) shows that for $B \in S_0^k(\mathbb{V}^*)$ there holds $2B^{(k,0)} = B - i\mathbb{J}(B)$, so that *B* is the real part of an element of \mathcal{K}^k , and, similarly, for $\beta \in \mathcal{K}^k$, there holds $\beta = 2(\operatorname{Re} \beta)^{(k,0)}$.

It follows immediately from Lemma 3.4 that on a Riemann surface (M, [h], J), a smooth section *B* of $S^k(T^*M)$ (respectively $S^k(TM)$) is the real part of a smooth section of \mathcal{K}^k (respectively \mathcal{K}^{-k}) if and only if it is completely [h]-trace-free, in which case $2B^{(k,0)} = B - i\mathbb{J}(B)$ and $\mathbb{J}(B)$ is also in $\Gamma(S_0^k(T^*M))$.

3.4. It follows from the identities $(\mathfrak{L}_X J)_{\bar{\alpha}}{}^{\beta} = 2i\bar{\partial}_{\bar{\alpha}}X^{\beta}$ and $(\mathfrak{L}_X J)_{\alpha\beta} = 2iD_{\alpha}X^{\flat}_{\beta}$ that on a Riemann surface (M, J) the following conditions on $X \in \Gamma(TM)$ are equivalent: X is conformal Killing; X is an infinitesimal automorphism of J; and $X^{(1,0)}$ is holomorphic. Since by the Riemann-Roch theorem a compact Riemann surface of genus greater than 1 admits no non-zero holomorphic vector fields, on such a surface every conformal Killing vector field is identically 0. These observations are generalized to sections of $S_0^h(T^*M)$ and $S_0^h(TM)$ by Lemma 3.5.

Lemma 3.5. Let (M, [h], J) be a Riemann surface and k > 0.

(1) If k > 1, a section $B \in \Gamma(S^k(T^*M))$ is the real part of a holomorphic section of \mathcal{K}^k if and only if it is a trace-free Codazzi tensor. In this case $(DB)^{(k+1,0)} = DB^{(k,0)}$.

- (2) A one-form is the real part of a holomorphic section of \mathcal{K}^1 if and only if it is closed and co-closed. In this case $(DB)^{(2,0)} = DB^{(1,0)}$.
- (3) A section $B \in \Gamma(S^k(TM))$ is the real part of a holomorphic section of \mathcal{K}^{-k} if and only if it is a conformal Killing tensor.

Proof of Lemma 3.5. By Lemma 3.4, $B \in \Gamma(S^k(T^*M))$ is the real part of a smooth section of \mathcal{K}^k if and only if $B = 2 \operatorname{Re} B^{(k,0)}$, in which case $B \in \Gamma(S_0^k(T^*M))$. For $B \in \Gamma(S_0^k(T^*M))$, since $\mathcal{L}_h(B)$ and $\operatorname{div}_h(B)$ are [h]-trace-free, there follow from (3.4) and (3.5),

$$D^{1,0}B^{(k,0)} = \mathcal{L}_h(B)^{(k+1,0)}, \quad \bar{\partial}B^{(k,0)} = \overline{h^{(1,1)}} \otimes \operatorname{div}_h(B)^{(k-1,0)}, \quad \text{if } k > 1, \quad (3.10)$$

$$D^{1,0}B^{(1,0)} = \mathcal{L}_h(B)^{(2,0)}, \qquad 2\bar{\partial}B^{(1,0)} = dB + \operatorname{div}_h(B)\overline{h^{(1,1)}}, \qquad \text{if } k = 1.$$
(3.11)

Comparing (3.10) and (3.11) with (3.4) and (3.5) of Lemma 3.3 shows (1) and (2). In the case of (1) or (2) then $DB \in \Gamma(S_0^k(T^*M))$ and it follows from the definition of \mathbb{J} that $\mathbb{J}DB = D\mathbb{J}B$. By Lemma 3.4, $2B^{(k,0)} = B - i\mathbb{J}B$, so $2DB^{(k,0)} = DB - iD\mathbb{J}B = DB - i\mathbb{J}DB = (DB)^{(k+1,0)}$, the last equality also by Lemma 3.4.

Again by Lemma 3.4 if $B \in \Gamma(S^k(TM))$ is to be the real part of a section of \mathcal{K}^{-k} it must be trace-free, in which case $2B^{(k,0)} = B - i\mathbb{J}(B)$, so that B is the real part of a holomorphic section of \mathcal{K}^{-k} if and only if $\bar{\partial}B^{(k,0)} = 0$. Since raising and lowering indices interchanges type, it follows from (3.10) that $\bar{\partial}B^{(k,0)} =$ 0 if and only if $0 = \bar{\partial}B^{\flat(0,k)} = \mathcal{L}_h(B^{\flat})^{(0,k+1)}$. Hence, B is the real part of a holomorphic section if and only if $\mathcal{L}_h(B^{\flat}) = 0$, or, what is by definition the same, B is a conformal Killing tensor.

Lemma 3.6. Let (M, [h], D) be a Riemann surface and σ the real part of a holomorphic section of \mathbb{K}^k which is not identically zero. View σ as a tensor of rank |k|, covariant or contravariant according to whether k is positive or negative. For any $h \in [h]$ there hold

$$2\Delta_h \sigma = k \mathcal{R}_h \sigma, \qquad \Delta_h |\sigma|_h^2 = 2|D\sigma|_h^2 + k \mathcal{R}_h |\sigma|_h^2. \tag{3.12}$$

On the subset $M^* = \{ |\sigma|_h^2 \neq 0 \}$, which is the complement of a discrete set of points, there hold

$$2|d|\sigma||_{h}^{2} = |D\sigma|_{h}^{2}, \qquad \Delta_{h} \log |\sigma|_{h}^{2} = k\mathfrak{R}_{h}. \qquad (3.13)$$

When $k \neq 0$, the metric ${}^{*}h_{ij} := |\sigma|_{h}^{2/k}h_{ij}$ on M^{*} is flat. If M is a torus then $M^{*} = M$.

Proof. Tracing the (Kähler) identity $\Re_{ijp}{}^{l}J_{k}{}^{p} = \Re_{ijk}{}^{p}J_{p}{}^{l}$ and using the algebraic Bianchi identity yields $\omega^{pq}\Re_{pqij} = -2J_{i}{}^{p}\Re_{pj}$, from which follows $\Re_{\alpha\bar{\beta}\gamma}{}^{\gamma} = \Re_{\alpha\bar{\beta}} = (\Re_{h}/2)h_{\alpha\bar{\beta}}$. For $s \in \Gamma(\mathcal{K}^{k})$ there results $2D_{[\alpha}D_{\bar{\beta}]}s = -kR_{\alpha\bar{\beta}\gamma}{}^{\gamma}s =$

 $-(k/2)\mathcal{R}_h h_{\alpha\beta}s$. View σ as the real part of a holomorphic section s of \mathcal{K}^k . That s be holomorphic implies the second equality of

$$2\Delta_h s = 2h^{ij} D_i D_j s = 2h^{\bar{\alpha}\beta} D_{\bar{\alpha}} D_{\beta} s = h^{\bar{\alpha}\beta} \left(2D_{\beta} D_{\bar{\alpha}} s + k \mathcal{R}_h h_{\alpha\bar{\beta}} \right) s = k \mathcal{R}_h s$$

Taking the real part shows the first equation of (3.12), from which the second equation of (3.12) follows. By (1) and (2) of Lemma 3.5, $(D\sigma)^{(k+1,0)} = D^{1,0}s = Ds$, and so $|D\sigma|_h^2 = 2|Ds|_h^2$. From $d|\sigma|_h^2 = 2D(s\bar{s}) = 2(\bar{s}Ds + sD\bar{s})$ there results $|d|\sigma|^2|^2 = 8|s|^2|Ds|^2 = 2|\sigma|^2|D\sigma|^2$, which proves the first equality of (3.13). On M^* there holds by (3.12) and the first equality of (3.13),

$$\Delta_{h} \log |\sigma|_{h}^{2} = 2|\sigma|_{h}^{-2} \left(\Delta_{h} |\sigma|_{h}^{2} - 4|d|\sigma||_{h}^{2} \right) = 2|\sigma|_{h}^{-2} \left(\Delta_{h} |\sigma|_{h}^{2} - 2|D\sigma|_{h}^{2} \right) = k\mathcal{R}_{h}.$$

That **h* is flat follows from (2.1) and the second equality of (3.13). If *M* is a torus, then σ is parallel for a flat representative of [*h*], so has constant norm, which is not zero, as σ is not identically zero, and so $M^* = M$.

Note that **h* does not depend on the choice of $h \in [h]$, and is determined by the requirement that $|\sigma|_{*h}^2 = 1$. Corollary 3.7 is the specialization to Riemann surfaces of the results of [46].

Corollary 3.7. If M is a sphere then there is no non-trivial trace-free Codazzi tensor nor any non-trivial harmonic one-form. If M is a torus then any trace-free Codazzi tensor, harmonic one-form, or conformal Killing tensor is parallel with respect to a flat metric conformal to h. If a Riemann surface (M, [h]) is compact with genus g > 1 then any conformal Killing tensor is identically 0.

Proof. These claims follow either from Riemann Roch together with Lemma 3.5, or from the maximum principle applied to (3.12) for a constant scalar curvature representative $h \in [h]$.

3.5. For an oriented smooth surface M, let Diff(M) be its group of diffeomorphisms viewed as a topological group in the C^{∞} compact-open topology, and let Diff⁺(M) be the subgroup of orientation-preserving diffeomorphisms. The connected component Diff $_0(M)$ of the identity of Diff(M) is evidently contained in Diff⁺(M), and comprises the diffeomorphisms of M smoothly isotopic to the identity (see [3, Corollary 1.2.2]). Let J(M) be the space of complex structures on M inducing the given orientation, with the topology of C^{∞} -convergence. The group Diff⁺(M) acts on J(M) by pull-back and the quotient J(M)/Diff $_0(M)$ is the **Teichmüller space** $\mathcal{T}(M)$. The oriented mapping class group Map⁺(M) := Diff⁺(M)/Diff $_0(M)$ acts on $\mathcal{T}(M)$ with quotient J(M)/Diff $_0(M)$, the **moduli space of complex structures** on M. For background on these spaces from a point of view compatible with that here see [28] and [77].

Let $Q^k(M)$ be the space comprising pairs $(J, B) \in J(M) \times \Gamma(S^k(T^*M))$ such that with respect to the decomposition of tensors by type determined by $J, B^{(k,0)}$

is $\bar{\partial}_J$ -holomorphic. The group Diff⁺(M) acts by pull-back on $Q^k(M)$. Denote the quotient by $Q^k(M) = Q^k(M) / \text{Diff}^+(M)$. The projection $Q^k(M) \to J(M)$ commutes with the Diff⁺(M) action so descends to a projection $\rho : Q^k(M) \to \mathcal{T}(M)$ sending the equivalence class [J, B] to the equivalence class [J]. If a representative $J \in [J]$ is chosen then the fiber $\rho^{-1}([J])$ is identified with the space $H^0(M, \mathcal{K}_J^k)$ of J-holomorphic k-differentials. The space $Q^k(M)$ also will be referred to as the vector bundle over the Teichmüller space of M the fiber of which over [h] comprises the k-holomorphic differentials with respect to the complex structure induced by [h]. (This makes sense for negative k if $S^k(T^*M)$ is replaced by $S^{|k|}(TM)$).

3.6. Here are recalled some aspects of the correspondence between holomorphic differentials and singular flat metrics which are motivating for the discussion in Section 9.6 of the action of $GL^+(2, \mathbb{R})$ on the deformation space of strictly convex flat real projective structures. Related background can be found in many places, *e.g.* [58,69,70], or [72].

For $\tau > 0$, the metric $dr^2 + r^2 dt^2$ on the cone $V_{\tau} = \{(r, t) : r \ge 0, t \in [0, \tau)\}$ is Euclidean away from the vertex (where r = 0), where it is said to have a **conical singularity** of **cone angle** τ . The change of variables $z = ((b+1)r)^{1/(b+1)}e^{it/(b+1)}$ identifies the cone $V_{2\pi(b+1)}$ isometrically with the singular metric $|z|^{2b}|dz|^2$ on \mathbb{C} . If there are integers β and k such that $b = \beta/k > -1$, this metric is that determined as in Lemma 3.6 by the holomorphic k-differential $z^{\beta}dz^{k}$. Conversely, this shows how to associate to a singularity with cone angle $2\pi(\beta/k + 1)$ a holomorphic kdifferential with a zero of order β/k at the singularity.

By a **flat Euclidean structure** is meant an atlas of charts in which the transition functions are restrictions of Euclidean isometries. Such a structure determines a positive homothety class of flat Riemannian metrics, and an underlying flat real affine structure. If the transition functions can be chosen to be orientation preserving, then it determines also a corresponding complex structure, and flat complex affine structure. For a positive integer k, a 1/k-translation surface is a compact Riemann surface M equipped with a flat Euclidean structure on the complement M^* in M of a finite subset $S(M) \subset M$ generated by an atlas $\{z_i : U_i \to \mathbb{C}\}$ for which the transition functions have the form $z_i = e^{2\pi i m_{ij}/k} z_j + u_{ij}$ for some integers $0 \le m_{ii} \le k - 1$ and some complex numbers u_{ii} . In particular, the linear part of the affine holonomy around each $p \in S(M)$ is contained in the finite subgroup $\mathbb{Z}/k\mathbb{Z} \subset SO(2)$. In the cases k is 1 or 2, such a surface is called a **translation** or half-translation surface, which explains the terminology. The k-differentials dz_i^k and dz_i^k agree on the overlaps $U_i \cap U_j$, so patch together to give a holomorphic k-differential on M^* . A point of M^* will be called a **regular** point of σ , while points in S(M) will be called **singular**. From the form of the transition functions it follows that in a neighborhood of $p \in S(M)$ there is a chart in which a flat metric *h representing the flat Euclidean structure is isometric to a conical singularity of cone angle $2\pi(\beta/k+1) > 0$ for some integer β . Then the holomorphic k-differential constructed on M^* extends to the singular points via the local model described in the preceeding paragraph.

Let (M, [h], J) be a compact Riemann surface of genus g > 1 and, for k > 0, let $\sigma = B^{(k,0)}$ be a non-trivial holomorphic k-differential. By Lemma 3.6, ${}^{*}h = |B|_{h}^{2/k}h$, which does not depend on the choice of $h \in [h]$, is a flat metric on the complement M^{*} of the zero set of M. Around a regular point p_{0} choose a local holomorphic coordinate w (such that w = 0 corresponds to p_{0}) and write $\sigma = \phi(w)dw^{k}$ for a holomorphic function $\phi(w)$. Choose a branch of $\phi^{1/k}(w)$ near w = 0 and define a new coordinate, said to be **adapted** to σ , by $z(p) = \int_{p_{0}}^{p} \phi(w)^{1/k} dw$. In the z coordinate, $\sigma = dz^{k}$. If \tilde{z} is another coordinate constructed in this way in a neighborhood of p_{0} then, since $dz^{k} = d\tilde{z}^{k}$ on the overlap, there are an integer $0 \le m \le k - 1$ and a complex number β such that $\tilde{z} = e^{2\pi i m/k} z + \beta$. Consequently the local charts constructed in this way determine on M^{*} a flat Euclidean structure, which makes M a 1/k-translation surface, and for which the underlying homothety class of flat Riemannian metrics is generated by the metric ${}^{*}h$. In a neighborhood of a singular point p_{0} of σ choose as before a local holomorphic coordinate w and write $\sigma(w) = \phi(w)dw^{k}$, where now it is supposed that $\phi(w)$ is holomorphic with a zero of order β at 0. Define a coordinate z(p) adapted to σ by

$$z(p) = \left(\frac{\beta+k}{k}\int_{p_0}^p \phi(w)^{1/k} \, dw\right)^{\frac{k}{\beta+k}}$$

The coordinate z is determined up to multiplication by a $(\beta + k)$ th root of unity, and $\sigma = \phi(w)dw^k = z^\beta dz^k = (\frac{k}{\beta+k}d(z^{\beta/k+1}))^k$. In an adapted coordinate around a singular point p of σ of order β the flat metric *h has the form $|z|^{2\beta/k}|dz|^2$, and so p is a cone point of angle $2\pi(\beta/k+1)$, and the linear holonomy of *h around p is a rotation of angle $2\pi\beta/k$. By Riemann Roch the sum of the orders $\beta(p)$ of the zeroes p of σ is 2k(g-1), and so the cone angles $\tau(p) = 2\pi(\beta(p)/k+1)$ of *h satisfy the relation $\sum_{p \in M} (\tau(p)/2\pi - 1) = 2(g-1)$. The same conclusion follows from a version of the Gauß-Bonnet theorem for metrics with conic singularities (see [70]).

The preceding establishes a bijective correspondence between 1/k-translation surfaces and Riemann surfaces with a holomorphic k-differential.

A pair (J, σ) for which σ is a *J*-holomorphic *k*-differential generates a complex curve in Teichmüller space, and, moreover, in $\Omega^k(M)$, as follows. The curve in $\mathcal{T}(M)$ comprises those conformal structures generated by singular flat metrics real affinely equivalent to the flat metric **h* determined by (J, σ) . The evolution of the *k*-differential σ along this curve in $\mathcal{T}(M)$ is determined tautologically by the requirement that the singular flat metric associated to the evolved *k*-differential be a member of a conformal structure representing the point in the curve over which it lies. These curves should be relevant to understanding compactifications of the space of strictly convex flat real projective structures on a compact surface of genus g > 1 (see Section 9.6).

These curves can be described analytically in terms of an action of the group $GL^+(2, \mathbb{R})$ on $Q^k(M)$, with respect to which they are images of the hyperbolic disk $GL^+(2, \mathbb{R})/C^+O(2)$. The structure of this $GL^+(2, \mathbb{R})$ action on $Q^k(M)$ has

been intensively studied in the k = 1, 2 cases; see [45,58], and [37] for background and references. An identification of real vector spaces $\mathbb{C} \simeq \mathbb{R}^2$ is fixed so that $g \in$ $GL^+(2,\mathbb{R})$ acts on $z = x + iy \in \mathbb{C}$ real linearly by $g \cdot z = (ax + by) + i(cx + dy)$. The complex field \mathbb{C}^{\times} acting by multiplication is identified with the oriented conformal subgroup $C^+O(2)$. That is $z = re^{i\theta}$ corresponds to $a = r\cos\theta = d, -b = d$ $r\sin\theta = c$. Given $(J, \sigma) \in Q^k(M)$, let $\{z_i\}$ be an atlas on M^* of J-holomorphic charts adapted to σ . This atlas makes M^* a flat Euclidean manifold with linear holonomy in $\mathbb{Z}/k\mathbb{Z}$. For $g \in GL^+(2,\mathbb{R})$ the collection $\{\tilde{z}_i = gz_i\}$ determines a flat Euclidean structure which has also holonomy in $\mathbb{Z}/k\mathbb{Z}$, so a structure of 1/ktranslation surface on M, the cone angles of which are same as those of the original structure determined by (J, σ) . This 1/k-translation surface corresponds to a pair $g \cdot (J, \sigma) = (g \cdot J, g \cdot \sigma) \in Q^k(M)$, and this defines the desired $GL^+(2, \mathbb{R})$ action on $Q^k(M)$. In the case $k = 1, g \cdot \sigma$ is given in z-coordinates by the real linear action of g on σ ; that is $g \cdot \sigma = (a \operatorname{Re} \sigma + b \operatorname{Im} \sigma) + i(c \operatorname{Re} \sigma + d \operatorname{Im} \sigma)$, but in general, with respect to a coordinate z adapted to σ , the new k-differential $g \cdot \sigma$ is not given by a linear action. The underlying complex structure $g \cdot J$ will equal the original one if $g \in C^+O(2) = \mathbb{C}^{\times}$. In this case $g \cdot \sigma$ is expressible in adapted coordinates in terms of a linear representation of $C^+O(2)$; namely, it is given by the product by $z^k \sigma = z^k B^{(k,0)} = r^k (\cos(k\theta)B - \sin(k\theta)\mathbb{J}(B))^{(k,0)}$, where z is the complex number corresponding to g. This action of $C^+O(2)$ is considered in Section 9.7. The more interesting actions of hyperbolic and parabolic elements of $GL^+(2,\mathbb{R})$ are more difficult to describe; their study is left for the future.

If $\Phi \in \text{Diff}(M)$ then $z_i \circ \Phi$ are $\Phi^*(\sigma)$ adapted coordinates with respect to the conformal structure $\Phi^*([h])$, and since $g(z_i \circ F) = (gz_i) \circ F$, it follows that the action of $GL^+(2, \mathbb{R})$ on $Q^k(M)$ is diffeormorphism equivariant in the sense that $\Phi^*(g \cdot (J, \sigma)) = g \cdot \Phi^*(J, \sigma)$, so descends to an action on $Q^k(M)$ which commutes with Map⁺(M).

4. AH structures on surfaces

In this section are given the basic definitions related to AH structures. Though adapted to the peculiarities of the two-dimensional case, the exposition is consistent with the conventions of [30].

4.1. Two affine connections are **projectively equivalent** if they have the same unparameterized geodesics in the sense that the image of any geodesic of one connection is the image of a geodesic of the other connection. This is the case if and only if the symmetric part of their difference tensor is pure trace. A **projective structure** $[\nabla]$ is an equivalence class of projectively equivalent affine connections. For a torsion-free affine connection ∇ on a surface there vanishes the usual projective Weyl tensor, or, what is the same,

$$R_{ijk}^{\ \ l} = \delta_i^{\ \ l} R_{(jk)} - \delta_j^{\ \ l} R_{(ik)} - R_{[ij]} \delta_k^{\ \ l}.$$
(4.1)

The projective Cotton tensor $C_{ijk} := -\nabla_i R_{(jk)} + \nabla_j R_{(ik)} + \frac{1}{3} \nabla_k R_{[ij]}$ does not depend on the choice of $\nabla \in [\nabla]$. On a surface, since a 3-form vanishes, there hold $C_{[i\,ik]} = 0$ and $\nabla_{[i}C_{ik]l} = 0$.

4.2. The definition of an AH structure $([\nabla], [h])$ on an *n*-manifold was given in the first paragraph of Section 1.1 of the introduction. Let H_{ii} be the normalized representative of [h]. An equivalent definition is that for each $\nabla \in [\nabla]$ there be a one-form σ_i such that $\nabla_{[i}H_{i]k} = 2\sigma_{[i}H_{i]k}$. In two dimensions, by Lemma 3.1 the tensor $\nabla_{i} H_{ik}$ is pure trace, so there is always such a one-form. Hence in two dimensions an AH structure could simply be defined to be a pair $([\nabla], [h])$, the necessary compatibility being automatic. If M is oriented, an alternative way to see this is the following. On a surface with Kähler structure (h, J, ω) , the Hodge star operator on one-forms is $(\star \sigma)_i = -\sigma_p J_i^p$, while on two-forms it is $\star \beta =$ $\frac{1}{2}\omega^{\sharp ij}\beta_{ij}$. Because any two-form is a multiple of any other, given $\nabla \in [\nabla]$ there is a one-form σ_i such that $\nabla_{[i}h_{i]k} = \omega_{ij}\sigma_k$. An easy computation shows

$$\omega_{ij}\sigma_k = -2(\star\sigma)_{[i}h_{j]k},\tag{4.2}$$

so $\nabla_{[i}h_{i]k} = -2(\star\sigma)_{[i}h_{i]k}$.

4.3. Given an AH structure $([\nabla], [h])$, there is a unique torsion-free $\nabla \in [\nabla]$ such that $\nabla_i H_{ik} = 0$; given any torsion-free $\overline{\nabla} \in [\nabla]$ with $\overline{\nabla}_{[i} H_{i]k} = 2\sigma_{[i} H_{i]k}$ it is given by $\nabla = \tilde{\nabla} - 2\sigma_{(i}\delta_{j)}^{k}$. This distinguished representative of $[\nabla]$ is called the aligned representative of the AH structure.

From now on, except where stated otherwise, indices and raised and lowered using H_{ii} and the dual bivector H^{ij} . Because det $H_{ii} = 1$ there holds $H^{pq} \bar{\nabla}_i H_{pq} =$ 0 for any $\overline{\nabla} \in [\nabla]$. Alternative characterizations of the alignment condition are given in Lemma 4.1, the proof of which is straightfoward, using the identity

$$A_{ijk} = A_{i(jk)} + A_{j(ik)} - A_{k(ij)} + A_{[ij]k} + A_{[ki]j} - A_{[jk]i}, \qquad (4.3)$$

valid for any covariant 3-tensor.

Lemma 4.1. Let $[\nabla]$ be a projective structure and [h] a conformal structure on a surface M. There is a unique torsion-free representative $\nabla \in [\nabla]$ satisfying any one of the following equivalent conditions

(1)
$$\nabla_{i} H_{ik} = 0.$$

- (2) $\nabla_i H_{jk} = \nabla_{(i} H_{jk)}.$
- (3) $\nabla_p H^{ip} = 0.$
- (4) $H^{pq} \nabla_p H_{qi} = 0$. That is, $\nabla_i H_{jk}$ is completely trace-free.
- (5) For any $h \in [h]$ there holds $2h^{pq} \nabla_p h_{qi} = h^{pq} \nabla_i h_{pq}$. (6) For any $h \in [h]$ there holds $\nabla_{[i}h_{j]k} = 2\gamma_{[i}h_{j]k}$ with $4\gamma_i = h^{pq} \nabla_i h_{pq}$.

Henceforth, except where stated otherwise, ∇ denotes the aligned representative of an AH structure. While it may seem perverse speak of the projective structure $[\nabla]$

if one works only with a distinguished representative $\nabla \in [\nabla]$, later developments will show the utility of the perspective.

4.4. The **cubic torsion** of an AH structure is the tensor \mathcal{L}_{ij}^{k} defined in terms of an arbitrary representative $\nabla \in [\nabla]$ by setting $\mathcal{L}_{ij}^{p} H_{pk}$ equal to the completely trace-free part of $\nabla_{(i}H_{jk)}$. For the aligned representative $\nabla \in [\nabla]$ the cubic torsion is just $\nabla_{i}H_{jk} = \nabla_{(i}H_{jk)} = \mathcal{L}_{ijk}$. An AH structure for which $\mathcal{L}_{ij}^{k} \equiv 0$ is a **Weyl structure**. The aligned representative of a Weyl structure is what is usually called a Weyl connection.

4.5. In the split signature case, an appropriate finite cover \hat{M} of M is orientable and such that the null cone of the lifted metric on \hat{M} is orientable; in this case the null cone of the split signature metric is a pair of transverse nowhere vanishing line fields, and so \hat{M} has Euler characteristic 0, and hence M has as well, and so if M is compact it is a torus or a Klein bottle. The study of Riemannian Einstein AH structures on surfaces makes use of Hodge theory, the associated complex structure, and so forth. While the study of split signature Einstein AH structures on surfaces is also interesting, it requires a different set of techniques and will be mostly ignored here, although it will always be indicated when the Riemannian hypothesis is necessary.

4.6. The basic example of an AH structure is the following. A hypersurface immersion in flat affine space is **non-degenerate** if its second fundamental form (which takes values in the normal bundle) is non-degenerate. If the immersion is also co-oriented the second fundamental form determines a conformal structure on the hypersurface. A choice of subbundle transverse to the immersion induces on the hypersurface a torsion-free affine connection, and there is a unique choice of transverse subbundle such that the induced connection is aligned with respect to the conformal structure determined by the second fundamental form and the co-orientation. This choice of transverse subbundle is the **affine normal subbundle**. That this definition coincides with the customary one is proved in [30, Section 4.2].

The second fundamental form with respect to a particular vector field spanning the affine normal subbundle is a pseudo-Riemannian metric on the hypersurface. Usually an (equi)affine normal vector field W is distinguished by fixing a volume form on the ambient affine space and requiring that the volume density induced by the metric h_{ij} determined by the vector field agree with the volume density induced via interior multiplication of the vector field with the chosen volume form. Though it is often omitted, the prefix *equi* ought to be included because this construction is only invariant under equiaffine transformations of the ambient space (those preserving the given volume form). The metric h_{ij} determined by the affine normal is called the (equi)affine or Blaschke metric. The affine normal field admits the following description (that this is equiaffinely invariant is not self-evident; rather it follows from the definition in the previous paragraph, which is manifestly so). Fix the standard flat Euclidean metric δ_{IJ} on the ambient \mathbb{R}^3 and let $g_{ij} = i^*(\delta)_{ij}$ be the induced metric on the immersed, co-oriented, non-degenerate hypersurface $i : \Sigma \to \mathbb{R}^3$. Let *N* be the unit Euclidean normal consistent with the given coorientation, and let Π_{ij} be the second fundamental form defined with respect to *N*. The Gaussian curvature *K* is the function det $\Pi \otimes (\det g)^{-1}$, and the equiaffine metric h_{ij} has the form $h_{ij} = |K|^{-1/4} \Pi_{ij}$. Let \mathbb{E} be the radial (position) vector field on \mathbb{R}^3 . Along $i(\Sigma)$ the equiaffine vector field W satisfies $\Delta_h \mathbb{E} = 2W$.

For the convenience of a reader not familiar with affine geometry, the translation of these definitions into local coordinates is recalled briefly. For details, consult [11] or the textbook [63]. Locally a non-degenerate hypersurface Σ is given as a graph z = f(x) where $x^i \in \Omega \subset \mathbb{R}^2$. Let ∂ denote the flat connection on Ω with respect to which the dx^i are parallel, and write $f_{i_1...i_k} = \partial_{i_1} \dots \partial_{i_k} f$ and Hess $f = f_{ij}$. Also let f^{ij} be the tensor inverse to f_{ij} . Define H(f) by det Hess $f = H(f)(dx^1 \wedge dx^2)^2$. The normalized representative H_{ij} of the AH structure ([∇], [h]) induced on Σ is given by $H_{ij} = |\mathbf{H}(f)|^{-1/4} f_{ij}$, while the aligned representative $\nabla \in [\nabla]$ is given by $\nabla = \partial + \frac{1}{4} f_{ij} f^{pq} f_{pqr} f^{rk}$. The affine normal is W = $|H(f)|^{1/4} \left(Z - \frac{1}{4} f^{pq} f_{pqr} f^{ri} X_i \right)$ in which $X_i = \frac{\partial}{\partial x^i} - f_i \frac{\partial}{\partial z}$ and $Z = \frac{\partial}{\partial z}$. The Euclidean unit normal is $N = (1 + f^p f_p)^{-1/2} (Z - (1 + f^p f_p)^{-1} f^i X_i)$, in which $f^i = g^{ip} f_p$, and the second fundamental form \prod_{ij} induced by N is $\Pi_{ij} = (1+f^p f_p)^{-1/2} f_{ij}$, while the Gaussian curvature is $K = (1+f^p f_p)^{-2} H(f)$. Along $i(\Sigma)$ the radial field \mathbb{E} is equal to $x^p X_p - f^* Z$, in which $f^* = x^p f_p - f$ is the Legendre transform of f. The hypersurface is an **affine hypersphere** if its affine normal subbundles meet in a point, which may be at infinity. This holds if and only if either W is parallel or there is a constant S such that $W = -S\mathbb{E}$. For non-zero S this holds if and only if f^* solves the equation $(f^*)^4 H(f^*) = S^4$, in which the Hessian is taken with respect to the Legendre transformed variables $y_i := f_i$ on the domain $\Omega^* = df(\Omega)$. Most of the deeper results in the study of affine hyperspheres are obtained by studying this Monge-Ampere equation. In the two-dimensional case under consideration, there is also a Weierstrass-like representation of affine hypersurfaces due to Calabi, [12], and Wang, [75], which allows the use of complex analytic methods.

The simplest affine hyperspheres which are not quadrics are the hypersurfaces

$$\Sigma_{\alpha} = \{ (X, Y, Z) \in \mathbb{R}^3 : (Y^2 Z \cos \alpha - X Y Z \sin \alpha) = 1 \},\$$

for $\alpha \in (0, \pi)$. Writing x = X/Z and y = Y/Z, Σ_{α} is the **radial graph**

$$\{(x/u, y/u, -1/u) \in \mathbb{R}^3 : u(x, y) < 0\}$$

of the function $u(x, y) = (y^2 \cos \alpha - xy \sin \alpha)^{1/3}$, which solves $27u^4$ det Hess $u = \sin^2 \alpha$. This Σ_{α} is asymptotic to the cone over the base triangle $\Omega^* = \{(x, y) : u(x, y) < 0\}$; it can be written as the graph of the Legendre transform of u (that is, u corresponds to f^* in the previous paragraph). The parameter α is not important, as these examples are all equivalent by an affine transformation, but is included to

illustrate that as $\alpha \to \pi$ the function *u* becomes degenerate; as the upper halfspace contains a complete affine line it supports no strictly convex negative function vanishing along its boundary.

The function $v(x, y) = -(x^2 + y^2)^{1/3}$ solves $v^4 H(v) = -(4/27)$ on $\mathbb{R}^2 \setminus \{0\}$. Its radial graph is the hypersurface $\{(X, Y, Z) \in \mathbb{R}^3 : (X^2 + Y^2)Z = 1\}$, which is an affine hypersphere. This does not contradict the preceeding paragraph because the Hessian of v has mixed signature (its eigenvalues are $-(2/3)v^{-2}$ and $(2/9)v^{-2}$), so in this case the equiaffine metric has split signature.

4.7. Given an AH structure $([\nabla], [h])$ the torsion-free connection $\overline{\nabla} := \nabla + \mathcal{L}_{ij}{}^k$ satisfies $\overline{\nabla}_i H_{jk} = -\mathcal{L}_{ijk}$, so is the aligned representative of the AH structure $([\overline{\nabla}], [h])$ formed by the projective structure $[\overline{\nabla}]$ generated by $\overline{\nabla}$ and the given [h]. This $([\overline{\nabla}], [h])$ is said to be **conjugate** to $([\nabla], [h])$. As its cubic torsion is $\mathcal{L}_{ij}{}^k = -\mathcal{L}_{ij}{}^k$, the conjugate of the conjugate is the original AH structure.

The conormal Gauß map of a non-degenerate co-oriented hypersurface immersion in flat affine space sends a point of the hypersurface to the annihilator of the space tangent at the image of the point to the hypersurface. The pull-back of the flat projective structure on the projectivization of the dual to the flat affine space via this conormal Gauß map forms with the conformal structure determined by the second fundamental form and the co-orientation the AH structure conjugate to that determined by the affine normal subbundle.

4.8. The **curvature** and **Ricci curvature** of an AH structure are defined to be the curvature $R_{ijk}{}^l$ and Ricci curvature $R_{ij} := R_{pij}{}^p$ of the aligned representative ∇ . The **scalar curvature** R is the density $R := R_p{}^p = H^{pq}R_{pq}$. Sometimes, for emphasis, the qualifier *weighted* will be added, and R will be called the *weighted* scalar curvature. It does not make sense to speak of the numerical value of R because R takes values in the line bundle $|\det T^*M|$; however it does make sense to speak of the vanishing of R and because $|\det T^*M|$ is oriented, to speak of the positivity or negativity of R. An AH structure is **proper** if its weighted scalar curvature is non-vanishing. When a representative $h \in [h]$ is given there will be written $R_h = |\det h|^{-1/2}R = h^{ij}R_{ij}$.

An AH structure is **projectively flat** if $[\nabla]$ is projectively flat. If the conjugate AH structure $([\bar{\nabla}], [h])$ is projectively flat, then $([\nabla], [h])$ is **conjugate projectively flat**.

4.9. An AH structure is **exact** if there is a representative $h \in [h]$ such that $\nabla_i \det h = 0$ for the aligned representative $\nabla \in [\nabla]$. If there is such an *h* it is determined uniquely up to positive homothety (on each connected component of *M*). Such an *h* will be called a **distinguished representative** of the AH structure. For example, the AH structure induced on a hypersurface in flat affine space is always exact, and the equiaffine metric is a distinguished representative. An AH structure is exact if and only if there is a global ∇ -parallel non-vanishing density of non-trivial weight, for if there is such a density, then some power of it is a non-vanishing density μ

such that $h_{ij} = \mu \otimes H_{ij} \in [h]$ verifies $\nabla |\det h| = \nabla \mu^2 = 0$ (the converse is obvious).

4.10. The **Faraday form** F_{ij} of an AH structure ($[\nabla], [h]$) is the curvature of the covariant derivative induced on the line bundle of -1/2-densities by the aligned representative $\nabla \in [\nabla]$. If R_{ijk}^{l} is the curvature of ∇ , then by definition and the traced algebraic Bianchi identity there hold

$$2F_{ij} = R_{ijp} {}^p = -2R_{[ij]}. ag{4.4}$$

Since F_{ij} is the curvature of a connection on a trivial line bundle it is always exact. The AH structure ($[\nabla], [h]$) is **closed** if $F_{ij} = 0$. The terminology directly extends that introduced for Weyl structures in [14].

If an AH structure ($[\nabla]$, [h]) has parallel weighted scalar curvature then either R vanishes identically or R vanishes nowhere and ($[\nabla]$, [h]) is exact, for if R vanishes nowhere, then $2\sigma_i = -R^{-1}\nabla_i R$ satisfies $d\sigma_{ij} = 2\nabla_{[i}\sigma_{j]} = -R^{-1}\nabla_{[i}\nabla_{j]}R = F_{ij}$.

The **Faraday primitive** γ_i associated to $h \in [h]$ is the one-form γ_i defined by $4\gamma_i = h^{pq}\nabla_i h_{pq}$. Note that the Faraday primitive associated to h depends only on the positive homothety class of h and not on h. From the Ricci identity follows $d\gamma_{ij} = 2\nabla_{[i}\gamma_{j]} = -F_{ij}$, so that γ_i is a primitive for $-F_{ij}$. If $\tilde{h}_{ij} = e^{2f}h_{ij} \in [h]$ then the corresponding one-form $\tilde{\gamma}_i$ differs from γ_i by an exact oneform, $\tilde{\gamma}_i = \gamma_i + df_i$. The equivalence class $\{\gamma\}$ of one-forms so determined is the **equivalence class of Faraday primitives induced by** ($[\nabla], [h]$). The Faraday primitives associated to $h \in [h]$ of the AH structure ($[\nabla], [h]$) and its conjugate ($[\bar{\nabla}], [h]$) are the same, and so these AH structures determine the same equivalence class of Faraday primitives. In particular, an AH structure is closed (respectively exact) if and only if the conjugate AH structure is closed (respectively exact).

Most properties of the Faraday curvature of Weyl structures hold also for AH structures. For example, Lemma 4.2 generalizes (trivially) of [14, Theorem 2.5].

Lemma 4.2. A definite signature AH structure on a surface is closed if and only if $\nabla^p F_{ip} = 0$.

Proof. By (4.4), there holds $\nabla_{[p}\nabla_{q]}F^{pq} = -R_{[pq]}F^{pq} - R_{pqa}{}^{a}F^{pq} = -F_{pq}F^{pq}$. Hence that $\nabla^{p}F_{ip} = 0$ implies F_{ij} is [h]-null; in definite signature this holds if and only if $F_{ij} \equiv 0$.

4.11. Because $f \Box_{\tilde{h}} \alpha_i = \Box_h \alpha_i + 2\sigma^{\sharp p} d\alpha_{pi} - 2d_h^* \alpha \sigma_i$, the **Hodge Laplacian** $\Box_h := dd_h^* + d_h^* d$ is not conformally invariant on one-forms. However, because d is independent of the metric and $f d_{\tilde{h}}^* \alpha = d_h^* \alpha$, and because a form is harmonic if and only if it is closed and co-closed, the Hodge decomposition of one-forms is conformally invariant. On a compact orientable Riemannian surface the Hodge decomposition implies there is a unique representative of $\{\gamma\} = \{\gamma + df : f \in$ $C^{\infty}(M)$ which is co-closed. Consequently, on such a surface there is associated to an AH structure a unique positive homothety class of representative metrics $h \in [h]$ for which the associated Faraday primitive γ_i is co-closed with respect to h. Such a representative metric will be called a **Gauduchon metric**. In higher dimensions the existence for an AH structure ($[\nabla], [h]$) of a representative of [h] distinguished up to positive homothety follows from arguments of P. Gauduchon (*e.g.* [34] and [35]), and, although in two dimensions the existence of such representatives follows from the Hodge decomposition, in this case the terminology *Gauduchon metric* is used also, for consistency. Note that without imposing some further normalization, such as setting the volume equal to a fixed constant, there is no naturally preferred Gauduchon metric. A distinguished metric on an exact AH structure is trivially also a Gauduchon metric.

4.12. If $([\nabla], [h])$ is an AH structure and $h \in [h]$ has corresponding Faraday primitive γ_i then the Levi-Civita connection D of h is related to the aligned representative ∇ by

$$D = \nabla + \frac{1}{2}\mathcal{L}_{ij}^{\ k} + 2\gamma_{(i}\delta_{j)}^{\ k} - h_{ij}\gamma^{\sharp k} = \nabla + \frac{1}{2}\mathcal{L}_{ij}^{\ k} + 2\gamma_{(i}\delta_{j)}^{\ k} - H_{ij}\gamma^{k}.$$
 (4.5)

On the other hand, equation (4.5) shows how to build from a metric and a one-form an AH structure pair with a given cubic torsion. Given a pseudo-Riemannian metric h_{ij} with Levi-Civita connection D, a completely symmetric, completely trace-free covariant 3-form $B_{ijk} = B_{(ijk)}$, and a one-form γ_i , defining $\mathcal{L}_{ij}{}^k = B_{ijp}h^{pk}$ and defining ∇ by (4.5) determines an AH structure with aligned representative ∇ , cubic torsion $\mathcal{L}_{ij}{}^k$, and such that γ_i is the Faraday primitive associated to h.

4.13. On an oriented surface the conformal structure underlying a Riemannian signature AH structure determines a complex structure. Lemma 4.3 describes the compatibility between them.

Lemma 4.3. Let $([\nabla], [h])$ be a Riemannian signature AH structure on an oriented surface and let J_i^{j} be the complex structure determined by [h]. Then

$$\mathcal{L}_{ij}{}^{k} = -J_{p}{}^{k}\nabla_{i}J_{j}{}^{p}, \qquad \nabla_{[i}J_{j]}{}^{k} = 0, \qquad J_{[i}{}^{p}\mathcal{L}_{j]p}{}^{k} = 0.$$
(4.6)

Proof. Fix $h \in [h]$ with Levi-Civita connection D and associated Faraday primitive γ_i . Since the tensor $B_{ijk} := \mathcal{L}_{ij} {}^p h_{pk}$ is completely trace-free, Lemma 3.4 implies $J_{[i} {}^p B_{j]kp} = 0$, which shows the third equation of (4.6). By (4.2) there holds

$$\begin{aligned} &h_{kp} \left(-\gamma_j J_i^{\ p} + h_{ij} (\star \gamma)^{\sharp p} - (\star \gamma)_j \delta_i^{\ p} + \omega_{ij} \gamma^{\sharp p} \right) \\ &= \gamma_j \omega_{ki} + \gamma_k \omega_{ij} - 2 (\star \gamma)_{[j} h_{k]i} = 3 \gamma_{[i} \omega_{jk]} = 0. \end{aligned}$$

$$(4.7)$$

By the last equality of (4.6), $\mathcal{L}_{ip}{}^k J_j{}^p = -\mathcal{L}_{ij}{}^p J_p{}^k$. Using (4.5) and (4.7) there results

$$0 = D_i J_j^{\ k} = \nabla_i J_j^{\ k} - \mathcal{L}_{ij}^{\ p} J_p^{\ k} - \gamma_j J_i^{\ k} + h_{ij} (\star \gamma)^{\sharp k} - (\star \gamma)_j \delta_i^{\ k} + \omega_{ij} \gamma^{\sharp k}$$
$$= \nabla_i J_j^{\ k} - \mathcal{L}_{ij}^{\ p} J_p^{\ k}.$$

This gives the first equation of (4.6), from which the second equation of (4.6) is immediate. \Box

Thus the cubic torsion of an AH structure can be seen as measuring the failure of the aligned representative to preserve the associated complex structure. By Lemma 3.4 the third equation of (4.6) shows that for any $h \in [h]$ the (3,0) part $B^{(3,0)}$ is a smooth section of the bundle of cubic holomorphic differentials.

From (4.6) it follows also that ∇ preserves *J* if and only if the AH structure is Weyl. In [62] (see also the more easily obtained [63]) a torsion-free affine connection on a complex manifold preserving the complex structure is said to be **affine Kähler** if the (2, 0) part of its curvature vanishes. (Though apt, the terminology is problematic because *Kähler affine* has been used in [23] to mean something else, in a somewhat related context; see Section 11.1). This curvature condition is automatic on a Riemann surface, so the aligned representative of a Weyl structure is an affine Kähler connection in this sense.

5. Curvature of an AH structure

In this section there are described the basic local curvature invariants of an AH structure ($[\nabla], [h]$) on a surface. The fundamental invariants are the weighted scalar curvature R, a trace-free symmetric tensor E_{ij} which is a multiple of the trace-free Ricci tensor, as well as the cubic torsion \mathcal{L}_{ij}^{k} and Faraday curvature F_{ij} . Conceptually important are Lemmas 5.1 and 5.2 which show when E_{ij} and \mathcal{L}_{ij}^{k} can be viewed as the real parts of holomorphic differentials. In [12] Calabi develops the geometry of hypersurfaces in flat affine three space; the fundamental geometric invariants are a quadratic differential B and a cubic differential A which is holomorphic when B vanishes; for the AH structure induced on such a hypersurface, constant multiples of the real parts of A and B are identified, respectively, with \mathcal{L}_{ij}^{k} and E_{ij} .

5.1. Let $([\nabla], [h])$ be an AH structure on a surface M. In what follows indices are raised and lowered using the normalized representative $H, \nabla \in [\nabla]$ is the aligned representative, and $h \in [h]$ is a representative with Faraday primitive γ_i and Levi-Civita connection D related to ∇ as in (4.5).

On a manifold of dimension n > 2 it is necessary to consider, in addition to the Ricci trace, the trace $R_{ip}^{p}_{j}$, because in general the trace-free symmetric parts of $R_{ip}^{p}_{j}$ and R_{ij} are independent, but in two dimensions tracing (4.1) shows that

$$R_{ip}{}^{p}{}_{j} + R_{ij} = RH_{ij}, (5.1)$$

so in this case there is no need to speak of $R_{ip} p_j$. It will be convenient to work instead with the trace-free symmetric tensor $E_{ij} = E_{(ij)}$ defined by

$$R_{ij} = -2E_{ij} + \frac{1}{2}RH_{ij} - F_{ij}.$$
(5.2)

The apparently unnatural coefficient -2 is chosen for consistency with the conventions of [30]. Substituting E_{ij} into (4.1) gives $R_{ijkl} = -4H_{l[i}E_{j]k} + RH_{l[i}H_{j]k} + F_{ij}H_{kl}$. However, because $H_{k[i}E_{j]l} - H_{l[i}E_{j]k}$ is trace-free and has symmetries corresponding to the Young diagram of the partition (22) it vanishes by Lemma 3.1, and so

$$R_{ijkl} = -2H_{k[i}E_{j]l} - 2H_{l[i}E_{j]k} + RH_{l[i}H_{j]k} + F_{ij}H_{kl}.$$
(5.3)

The Ricci identity implies

$$2H_{k[i}E_{j]l} + 2H_{l[i}E_{j]k} = -R_{ij(kl)} + F_{ij}H_{kl} = \nabla_{[i}\nabla_{j]}H_{kl} = \nabla_{[i}\mathcal{L}_{j]kl}.$$
 (5.4)

There hold

$$D_{i}\gamma_{j} = \nabla_{i}\gamma_{j} - \frac{1}{2}\mathcal{L}_{ij}^{p}\gamma_{p} - 2\gamma_{i}\gamma_{j} + H_{ij}\gamma_{p}\gamma^{p}, \quad D^{p}\gamma_{p} = \nabla^{p}\gamma_{p} = \gamma_{p}^{p}, \quad (5.5)$$
$$D_{i}\mathcal{L}_{ik}^{l} = \nabla_{i}\mathcal{L}_{ik}^{l} - \frac{1}{2}\mathcal{L}_{ij}^{p}\mathcal{L}_{kp}^{l} - \frac{1}{2}\mathcal{L}\delta_{ij}^{l}H_{ijk}$$

$$\mathcal{L}_{jk} = \nabla_i \mathcal{L}_{jk} - \frac{1}{2} \mathcal{L}_{ij} \mathcal{L}_{kp} - \frac{1}{2} \mathcal{L}_{0[i} H_{j]k} + \delta_i^{\ l} \gamma_p \mathcal{L}_{jk}^{\ p} + 2H_{i(j} \mathcal{L}_{k)p}^{\ l} \gamma^p - 3\gamma_{(i} \mathcal{L}_{jk)}^{\ l} - \mathcal{L}_{ijk} \gamma^l.$$
(5.6)

Tracing (5.4) in *il*, relabeling, and substituting in (5.1) yields the first equality of

$$4E_{ij} = \nabla_p \mathcal{L}_{ij}{}^p - \frac{1}{2}\mathcal{L}H_{ij} = D_p \mathcal{L}_{ij}{}^p - 2\gamma_p \mathcal{L}_{ij}{}^p, \qquad (5.7)$$

while the second follows from the first and tracing (5.6). Equation (5.7) plays an important role in deriving consequences of the Einstein equations; see Lemma 5.1 below. Recall that the curvature of *D* is written $\Re_{ijk}^{\ l} = \Re_h \delta_{[i}^{\ l} h_{j]k}$. Calculating $\Re_{ijkl} - R_{ijkl}$ using (4.5) and simplifying yields

$$\mathcal{R}_{h} = \mathbf{R}_{h} + \frac{1}{4}|\mathcal{L}|_{h}^{2} - 2|\det h|^{-1/2}\nabla^{p}\gamma_{p} = \mathbf{R}_{h} + \frac{1}{4}|\mathcal{L}|_{h}^{2} + 2d_{h}^{*}\gamma.$$
(5.8)

Equation (5.8) will be used repeatedly throughout the remainder of the paper.

5.2. For a Riemannian signature AH structure $([\nabla], [h])$ and a representative $h \in [h]$ with associated Kähler form ω_{ij} define $F_h \in C^{\infty}(M)$ by $2F_{ij} = F_h \omega_{ij}$. Equivalently, $F_h = 2 \star F$. Note that $2|F|_h^2 = F_h^2$, and $\mathcal{L}_{J\gamma^{\sharp}}\omega = d(\iota(J\gamma^{\sharp})\omega) = -d\gamma = F$. Let $F = F_h |\det h|^{1/2}$, which does not depend on the choice of $h \in [h]$. Decomposing (5.2) by parts and substituting (5.8) yields

$$R_{\alpha\beta} = -2E_{\alpha\beta},$$

$$R_{\alpha\bar{\beta}} = \frac{1}{2}(R - iF)H_{\alpha\bar{\beta}} = \frac{1}{2}(R_h - iF_h)h_{\alpha\bar{\beta}} = \frac{1}{2}(\mathcal{R}_h + \frac{1}{4}|\mathcal{L}|_h^2 + 2d_h^*\gamma - iF_h)h_{\alpha\bar{\beta}}.$$
(5.9)

Because of (5.9) it makes sense to refer to S := R - iF as the **complex weighted** scalar curvature of the AH structure ($[\nabla], [h]$). There will be written $S_h = R_h - iF_h$.

Since on a Riemann surface M a (0, 2) form vanishes, for an operator $\hat{\partial}$ on a complex vector bundle E to be a holomorphic structure it suffices that it satisfy

the Leibniz rule. It follows that if (M, J) is a Riemann surface the most general holomorphic structure on \mathcal{K}^k has the form $\hat{\partial} = \bar{\partial} + \sigma^{(0,1)}$ for an arbitrary one-form σ_i on M and the holomorphic structure $\bar{\partial}$ induced on \mathcal{K}^k by J. From (5.9) it is immediate that $R_{\alpha\beta} = 0$ if and only if $E_{ij} = 0$. These conditions mean that the Ricci curvature of ($[\nabla], [h]$) has type (1, 1). Equation (5.7) has the following nice interpretation.

Lemma 5.1. For an AH structure ($[\nabla]$, [h]) with cubic torsion \mathcal{L}_{ij}^{k} on an oriented surface the following are equivalent:

- (1) The curvature R_{ij} has type (1, 1).
- (2) For any $h \in [h]$ the (3,0) part of the tensor $B_{ijk} := \mathcal{L}_{ij} {}^{p}h_{kp}$ is $\hat{\partial}$ -holomorphic for the holomorphic structure $\hat{\partial} := \bar{\partial} - 2\gamma^{(0,1)}$ on \mathcal{K}^{3} , in which $\bar{\partial}$ is the holomorphic structure induced on \mathcal{K}^{3} by the conformal structure [h] and the given orientation, and $4\gamma_{i} = |\det h|^{-1}\nabla_{i}|\det h|$.

Proof. For $\tilde{h}_{ij} = fh_{ij}$ let $\tilde{\partial} = \bar{\partial} - 2\tilde{\gamma}^{(0,1)}$. Since $\tilde{\gamma}_i = \gamma_i + \frac{1}{2}d\log f_i$, $\tilde{\partial} = \hat{\partial} - \bar{\partial}\log f$. It follows that $\tilde{B}_{ijk} = fB_{ijk}$ satisfies $\tilde{\partial}\tilde{B}^{(3,0)} = f\tilde{\partial}B^{(3,0)} + \bar{\partial}f \otimes B^{(3,0)} = f\hat{\partial}B^{(3,0)}$, so that $\tilde{B}^{(3,0)}$ is $\tilde{\partial}$ -holomorphic if and only if $B^{(3,0)}$ is $\hat{\partial}$ -holomorphic. This shows that (5.1) has sense. For any $h \in [h]$ it follows from (3.10) that

$$\hat{\partial}B^{(3,0)} = \bar{\partial}B^{(3,0)} - 2\gamma^{(0,1)} \otimes B^{(3,0)} = \overline{h^{(1,1)}} \otimes \operatorname{div}_h(B)^{(2,0)} - 2\overline{h^{(1,1)}} \otimes (\iota(\gamma^{\sharp})B)^{(2,0)} = 4\overline{h^{(1,1)}} \otimes E^{(2,0)}.$$

the last equality by (5.7). Hence $B^{(3,0)}$ is $\hat{\partial}$ -holomorphic if and only if $E_{ij} \equiv 0$.

For $h \in [h]$ the Chern connection $\widehat{\nabla}$ on \mathcal{K}^3 determined by the Hermitian structure induced on \mathcal{K}^3 by $h^{(1,1)}$ and the holomorphic structure $\hat{\partial}$ is by definition the unique connection on \mathcal{K}^k such that $\widehat{\nabla}^{0,1} = \hat{\partial}$ and for which the induced Hermitian structure is parallel. In terms of the connection induced on \mathcal{K}^3 by the Levi-Civita connection D of h, $\widehat{\nabla}$ is expressible by

$$\widehat{\nabla} = D - 2i\gamma \circ J = D + 2i \star \gamma.$$
(5.10)

It follows that the difference of the curvatures of $\widehat{\nabla}$ and D is $2id \star \gamma = -2id_h^* \gamma \omega$.

5.3. Curvature of the conjugate AH structure

It is convenient to define $|\mathcal{L}|_{H}^{2} = \mathcal{L}_{abc}\mathcal{L}^{abc}$. Evidently $|\mathcal{L}|_{H}^{2}H_{ij} = |\mathcal{L}|_{h}^{2}h_{ij}$ for any $h \in [h]$. By Lemma 3.2, $2\mathcal{L}_{ip}\,{}^{q}\mathcal{L}_{jq}\,{}^{p} = |\mathcal{L}|_{H}^{2}H_{ij}$. Since $2\mathcal{L}_{k[i}\,{}^{p}\mathcal{L}_{j]lp} - |\mathcal{L}|_{H}^{2}H_{l[i}H_{j]k}$ has the algebraic symmetries of a Riemannian curvature tensor and is completely trace-free, Lemma 3.1 implies that it is identically zero. That is $2\mathcal{L}_{k[i}\,{}^{p}\mathcal{L}_{j]lp} = |\mathcal{L}|_{H}^{2}H_{l[i}H_{j]k}$. Let $([\bar{\nabla}], [h])$ be the AH structure conjugate to $([\nabla], [h])$. Decorate with a $\bar{}$ the tensors derived from the curvature \bar{R}_{ijkl} of $([\bar{\nabla}], [h])$. By definition

$$\bar{R}_{ijk}^{\ l} - R_{ijk}^{\ l} = 2\nabla_{[i}\mathcal{L}_{j]k}^{\ l} - 2\mathcal{L}_{k[i}^{\ p}\mathcal{L}_{j]p}^{\ l} = 2\nabla_{[i}\mathcal{L}_{j]k}^{\ l} - |\mathcal{L}|_{H}^{2}\delta_{[i}^{\ l}H_{j]k},$$

so, lowering indices using $2H_{lp}\nabla_{[i}\mathcal{L}_{j]k}^{p} = 2\nabla_{[i}\mathcal{L}_{j]kl} - 2\mathcal{L}_{lpl[i}\mathcal{L}_{j]k}^{p} = 2\nabla_{[i}\mathcal{L}_{j]kl} + \mathcal{L}H_{l[i}H_{j]k}$, and using (5.4) there holds

$$\bar{R}_{ijkl} = R_{ijkl} + 4H_{l[i}E_{j]k} + 4H_{k[i}E_{j]l}.$$
(5.11)

Tracing (5.11) and substituting (5.2) and $\bar{F}_{ij} = F_{ij}$ into the result yields $-4\bar{E}_{ij} + \bar{R}H_{ij} = 4E_{ij} + RH_{ij}$, which when traced shows $\bar{R} = R$ and $\bar{E}_{ij} = -E_{ij}$.

Tensors such as R (respectively E_{ij}) unchanged under conjugacy (multiplied by -1 under conjugacy) will be called **self-conjugate** (respectively **anti-self-conjugate**). Classes of AH structures defined by some condition on the curvatures preserved under conjugacy seem to be of particular interest. By (5.3) and (5.11), the self-conjugate and anti-self-conjugate parts of the curvature tensor are

$$\frac{1}{2}(R_{ijkl} + R_{ijkl}) = RH_{l[i}H_{j]k} + F_{ij}H_{kl},$$

$$\frac{1}{2}(R_{ijkl} - \bar{R}_{ijkl}) = -2H_{l[i}E_{j]k} - 2H_{k[i}E_{j]l}.$$
(5.12)

It follows that $([\nabla], [h])$ has self-conjugate curvature if and only if $E_{ij} = 0$. Also, for a Weyl structure the curvature is simply $R_{ijkl} = RH_{l[i}H_{j]k} + F_{ij}H_{kl}$. However, it will be evident later that on a compact orientable surface of genus at least 2 the class of AH structures with self-conjugate curvature tensor is considerably larger than the class of Weyl structures.

5.4. By Lemma 3.1, the projective Cotton tensor C_{ijk} satisfies $C_{ijk} = 2C_{[i}H_{j]k}$ in which $C_i := C_{ip}^{p}$. From the definition of C_{ijk} there results

$$C_{i} = -\frac{1}{2}\nabla_{i}R - \frac{1}{3}\nabla^{p}F_{ip} - 2\nabla^{p}E_{ip} + 2\mathcal{L}_{i}^{pq}E_{pq}.$$
 (5.13)

From (5.13) and the easily verified identities

$$\bar{\nabla}_i \bar{R} = \nabla_i R, \qquad \bar{\nabla}^p \bar{F}_{ip} = \nabla^p F_{ip}, \qquad \bar{\nabla}^p \bar{E}_{ip} = -\nabla^p E_{ip} + \mathcal{L}_i^{pq} E_{pq},$$

there result

$$\bar{C}_{i} = C_{i} + 4\nabla^{p} E_{ip} - 2\mathcal{L}_{i}^{pq} E_{pq},
C_{i} + \bar{C}_{i} = -\nabla_{i} R - \frac{2}{3} \nabla^{p} F_{ip} + 2\mathcal{L}_{i}^{pq} E_{pq},
C_{i} - \bar{C}_{i} = -4\nabla^{p} E_{ip} + 2\mathcal{L}_{i}^{pq} E_{pq} = -4D^{p} E_{ip},$$
(5.14)

in the last equality of which D is the Levi-Civita connection of any $h \in [h]$.

Lemma 5.2. A Riemannian signature AH structure $([\nabla], [h])$ on an oriented surface M has self-conjugate projective Cotton tensor if and only if E_{ij} is the real part of a holomorphic quadratic differential with respect to the complex structure determined by [h]. In particular, if M is a sphere, then $([\nabla], [h])$ has self-conjugate projective Cotton tensor if and only if $E_{ij} = 0$.

Proof. Since E_{ij} is trace-free, Lemma 3.5 implies that E_{ij} is the real part of a holomorphic quadratic differential if and only if $\operatorname{div}_h(E)_i = 0$, and by the last equality of (5.14) this holds if and only if $([\nabla], [h])$ has self-conjugate projective Cotton tensor.

Theorem 5.3. If a Riemannian signature AH structure $([\nabla], [h])$ on a compact oriented surface has self-conjugate projective Cotton tensor and non-negative weighted scalar curvature, then $E_{ij} = 0$.

Proof. By Lemma 5.2, E_{ij} is the real part of a holomorphic quadratic differential. By (3.12) of Lemma 3.6 and (5.8) for a Gauduchon metric $h \in [h]$ there holds $\Delta_h |E|_h^2 \ge 2|DE|_h^2 + \frac{1}{2}|\mathcal{L}|_h^2|E|_h^2 \ge 0$, and the maximum principle then forces $E_{ij} = 0.$

5.5. Since R and $|\mathcal{L}|_{H}^{2}$ are 1-densities their integrals are defined if M is compact. The L^{2} -norm $||\mathcal{L}||_{h}^{2}$ does not depend on the choice of representative $h \in [h]$ and equals $\int_{M} |\mathcal{L}|_{H}^{2} = \int_{M} |\mathcal{L}|_{h}^{2} d \operatorname{vol}_{h}$.

Theorem 5.4. If $([\nabla], [h])$ is a Riemannian AH stucture on a compact, orientable surface M, then the Euler characteristic $\chi(M)$ satisfies $4\pi\chi(M) \ge \int_M R$, with equality if and only if $([\nabla], [h])$ is Weyl. In particular:

(1) If $\int_M R \ge 0$, then either *M* is a sphere, or *M* is a torus and ([∇], [*h*]) is Weyl. (2) If *M* has genus at least one and ([∇], [*h*]) is not Weyl, then $\int_M R < 0$.

Proof. By the Gauß-Bonnet Theorem, for any $h \in [h]$, integrating (5.8) yields

$$4\pi \chi(M) = \int_{M} \mathcal{R}_{h} \, d \operatorname{vol}_{h} = \frac{1}{4} ||\mathcal{L}||_{h}^{2} + \int_{M} R \ge \int_{M} R.$$
 (5.15)

Equality holds in (5.15) if and only if $\mathcal{L}_{ij}^{k} = 0$. If $\int_{M} R \ge 0$ the Euler characteristic $\chi(M)$ of M must be non-negative, so M must the sphere or the torus. If M is a torus, the Euler characteristic is 0 and so (5.15) forces $||\mathcal{L}||_{h}^{2} = 0$, so that $([\nabla], [h])$ is a Weyl structure. If M has genus at least one and $([\nabla], [h])$ is not Weyl then $4\pi\chi(M) - \frac{1}{4}||\mathcal{L}||_{h}^{2} < 0$, showing the last claim.

6. Einstein equations

In this section the Einstein equations for AH structures are defined and their most basic properties described. **6.1.** By definition of E_{ij} , (5.7), and Lemma 5.1, the following conditions on an AH structure on a surface having cubic torsion \mathcal{L}_{ij}^{k} are equivalent.

- (1) The symmetric part of the Ricci tensor is trace free. That is $E_{ij} = 0$.
- (2) The Ricci tensor has type (1, 1).
- (3) The curvature is self-conjugate.

. . .

(4) For any $h \in [h]$ the (3, 0) part of the tensor $B_{ijk} := \mathcal{L}_{ij} {}^{p}h_{kp}$ is $\hat{\partial}$ -holomorphic for the holomorphic structure $\hat{\partial} := \bar{\partial} - 2\gamma^{(0,1)}$, in which $\bar{\partial}$ is the holomorphic structure induced by the conformal structure [h], and $4\gamma_i = \nabla_i \log |\det h|$.

Definition 6.1. An AH structure on a surface is **naive Einstein** if it satisfies (1)-(4).

The qualifier *naive* is meant to reflect that, while the most obvious generalization of the usual metric Einstein condition is simply to require the vanishing of the symmetric trace-free Ricci tensor, such an approach turns out to be inadequate.

By Lemma 5.2 a Riemannian signature AH structure $([\nabla], [h])$ on the twosphere has self-conjugate projective Cotton tensor if and only if it is naive Einstein. Similarly, by Theorem 5.3, a Riemannian signature AH structure $([\nabla], [h])$ on a compact, oriented surface is naive Einstein if it has non-negative weighted scalar curvature.

Definition 6.2. An AH structure $([\nabla], [h])$ is **Einstein** if it is naive Einstein and satisfies

$$\nabla_i R + 2\nabla^p F_{ip} = 0. \tag{6.1}$$

The condition (6.1) will be referred to as the conservation condition.

Let $h \in [h]$. Using $\nabla_i R = D_i R + 2\gamma_i R$ to expand (6.1), and using (5.5), (5.8), and the Ricci identity yields

$$\begin{aligned} |\det h|^{-1/2} (\nabla_i R + 2\nabla^p F_{ip}) &= D_i R_h + 2\gamma_i R_h + 2h^{pq} D_p F_{iq} + 4\gamma^{\sharp p} F_{ip} \\ &= D_i R_h + 2\gamma_i R_h - 2d_h^* d\gamma_i - 4\gamma^{\sharp p} F_{pi} \\ &= D_i \left(\mathcal{R}_h - \frac{1}{4} |\mathcal{L}|_h^2 \right) - \frac{1}{2} \gamma_i |\mathcal{L}|_h^2 + \mathcal{R}_h \gamma_i + 2\Delta_h \gamma_i - 4\gamma_i d_h^* \gamma - 4\gamma^{\sharp p} F_{pi} \\ &= D_i \left(\mathcal{R}_h - \frac{1}{4} |\mathcal{L}|_h^2 \right) - \frac{1}{2} \gamma_i |\mathcal{L}|_h^2 - 2(\Box_h - \mathcal{R}_h) \gamma_i - 4\gamma_i d_h^* \gamma - 4\gamma^{\sharp p} F_{pi} \end{aligned}$$
(6.2)
$$&= D_i \left(\mathcal{R}_h - \frac{1}{4} |\mathcal{L}|_h^2 - 4|\gamma|_h^2 \right) \\ &- \frac{1}{2} \gamma_i |\mathcal{L}|_h^2 - 2(\Box_h - \mathcal{R}_h) \gamma_i + 4\gamma^{\sharp p} (\mathcal{L}_{\gamma^{\sharp}} h)_{ip} - 4\gamma_i d_h^* \gamma. \end{aligned}$$

While the only explicit direct use of (6.1), then in the rewritten form (6.2), is made in the proof of Theorem 7.1, its role is fundamental. By (5.7) a Weyl structure on a surface is automatically naive Einstein. In [13], Calderbank showed that taking (6.1) as the definition of Einstein-Weyl yields a nice theory, specializing that in higher dimensions. By definition the Einstein AH equations restrict to Calderbank's Einstein-Weyl equations, and Calderbank's definition provided essential motivation for the general case. In dimensions n > 2 the conservation condition has a more general definition, but it follows from the differential Bianchi identity that a naive Einstein AH structure with self-conjugate curvature is Einstein, satisfying the analogue of (6.1) with n in place of 2; see [30, Lemma 4.2].

6.2. Lemma 6.3 follows from (6.1), Lemma 4.2, and the discussion at the end of Section 4.10.

Lemma 6.3. A Riemannian signature Einstein AH structure on a surface is closed if and only if it has parallel scalar curvature, in which case either it is proper and exact or it has vanishing weighted scalar curvature.

Lemma 6.4. For an Einstein AH structure $([\nabla], [h])$ on a surface, any one of the following statements implies the other two.

- (1) $([\nabla], [h])$ is projectively flat.
- (2) ($[\nabla]$, [h]) is conjugate projectively flat.
- (3) The weighted scalar curvature is parallel.

In particular if an Einstein ($[\nabla]$, [h]) either is proper or has vanishing scalar curvature then it is projectively flat and conjugate projectively flat.

Proof. The first claim is immediate from (5.14) and the Einstein equations. The last claim follows because a proper Einstein AH structure is exact so has parallel scalar curvature.

6.3. The example of affine hypersurfaces gives the primary motivation for the definition of the Einstein equations for AH structures.

Theorem 6.5. For a non-degenerate positively co-oriented hypersurface immersion into flat three-dimensional affine space the following are equivalent:

- (1) The image of the immersion is an affine hypersphere.
- (2) The AH structure induced on the hypersurface is Einstein.

Moreover, (1) and (2) imply

(3) The induced AH structure is projectively flat.

Proof. The equivalence of (1) and (2) is proved by the same argument as in the n > 2 case, which can be found as [30, Theorem 4.6]. The AH structure induced on an affine hypersurface is conjugate projectively flat because the conjugate AH structure is that induced via the conormal Gauß map from the flat projective structure on oriented projective space, so Lemma 6.4 implies that that the induced AH structure is projectively flat as well. Alternatively, the AH structure induced on a non-degenerate hypersurface in affine space is always closed, and when it is Einstein, this implies the scalar curvature is parallel, so by Lemma 6.4, that it is projectively flat.

6.4. Recall the definitions of F_h and S from Section 5.2. Note that S is a smooth section of the complexification | Det $T^*M | \otimes_{\mathbb{R}} \mathbb{C}$ of the line bundle of 1-densities and that the (0, 1) part of the aligned representative ∇ of an AH structure induces a holomorphic structure on | Det $T^*M | \otimes_{\mathbb{R}} \mathbb{C}$.

Lemma 6.6. A naive Einstein Riemannian signature AH structure ($[\nabla], [h]$) on an oriented surface M is Einstein if and only if S is a holomorphic section of | Det $T^*M | \otimes_{\mathbb{R}} \mathbb{C}$ with respect to the holomorphic structure $\nabla^{0,1}$.

Proof. The claim to be verified is simply $\nabla^{0,1}\mathbf{S} = \nabla^{0,1}(R - i\mathbf{F}) = 0$. First it is shown that $([\nabla], [h])$ is Einstein if and only if for any $h \in [h]$ the complex valued function $\mathbf{R}_h - i\mathbf{F}_h$ is $\hat{\partial}$ -holomorphic for the holomorphic structure $\hat{\partial} = \bar{\partial} + 2\gamma^{(0,1)}$ on the trivial line bundle $M \times \mathbb{C}$. It is claimed that the conservation condition (6.1) is equivalent to

$$0 = \hat{\partial}(\mathbf{R}_{h} - i\mathbf{F}_{h}) = \bar{\partial}(\mathbf{R}_{h} - i\mathbf{F}_{h}) + 2(\mathbf{R}_{h} - i\mathbf{F}_{h})\gamma^{(0,1)}$$

= $\bar{\partial}(\mathbf{R}_{h} - 4|\gamma|_{h}^{2} - i\mathbf{F}_{h}) + 2\mathbf{R}_{h}\gamma^{(0,1)} + 4(\iota(\gamma^{\sharp})\mathfrak{L}_{\gamma^{\sharp}}h)^{(0,1)}.$ (6.3)

Rewriting the first equality of (6.2) yields

$$|\det h|^{-1/2}(\nabla_i R + 2\nabla^p F_{ip}) = d\mathbf{R}_{h\,i} + 2\gamma_i \mathbf{R}_h - (\star d\mathbf{F}_h)_i - 2\mathbf{F}_h(\star\gamma)_i.$$
(6.4)

The (0, 1) part of this last expression is

$$\bar{\partial}(\mathbf{R}_h - \mathbf{i}\mathbf{F}_h) + 2(\mathbf{R}_h - \mathbf{i}\mathbf{F}_h)\gamma^{(0,1)} = \hat{\partial}(\mathbf{R}_h - \mathbf{i}\mathbf{F}_h), \tag{6.5}$$

from which the first claim and the first equality of (6.3) are evident. Taking the (0, 1) part of

$$d_{i}|\gamma|_{h}^{2} = 2\gamma^{\sharp p}D_{i}\gamma_{p} = 2\gamma^{\sharp p}D_{(i}\gamma_{p)} + 2\gamma^{\sharp p}D_{[i}\gamma_{p]}$$

$$= \gamma^{\sharp p}(\mathfrak{L}_{\gamma^{\sharp}}h)_{ip} - \frac{1}{2}\gamma^{\sharp p}F_{h}\omega_{ip} = \gamma^{\sharp p}(\mathfrak{L}_{\gamma^{\sharp}}h)_{ip} + \frac{1}{2}F_{h}(\star\gamma)_{i},$$

yields $2\bar{\partial}|\gamma|_h^2 = iF_h\gamma^{(0,1)} + 2(\iota(\gamma^{\sharp})\mathfrak{L}_{\gamma^{\sharp}}h)^{(0,1)}$. Substituting this into what comes before yields the second equality of (6.3). By definition of γ_i , for any $h \in [h]$ there holds

$$\nabla^{0,1}(R - iF) = \bar{\partial}(R_h - iF_h) |\det h|^{1/2} + (R_h - iF_h)\nabla^{0,1} |\det h|^{1/2}$$

= $\left(\bar{\partial}(R_h - iF_h) + 2(R_h - iF_h)\gamma^{(0,1)}\right) |\det h|^{1/2},$

from which the claim follows.

By Lemma 6.6, a Riemannian AH structure on an oriented surface is Einstein if and only if its Ricci curvature has type (1, 1) and its complex weighted scalar curvature is holomorphic.

Lemma 6.7. Let $([\nabla], [h])$ be a Riemannian signature Einstein AH structure on an oriented surface M. For each Gauduchon metric $h \in [h]$ the square norm $|S_h|_h^2 = R_h^2 + F_h^2$ of the complex scalar curvature is h-harmonic. If M is moreover compact then $R_h^2 + F_h^2$ is constant. If $H^1(M; \mathbb{R}) = \{0\}$ there is a $v \in C^{\infty}(M)$ such that $e^{2iv}S_h$ is holomorphic on M.

Proof. Denote by the same notations the lifts to the universal cover \tilde{M} of M of $([\nabla], [h])$ and all the associated tensors, functions, etc. Let γ be the Faraday primitive of h. By assumption $\star \gamma$ is closed, so on \tilde{M} there is a smooth function v such that $\star \gamma = -dv$, or, what is equivalent, $\gamma = \star dv$. By (6.5) of Lemma 6.6 the function $b = e^{2iv}(R_h - iF_h)$ is holomorphic on \tilde{M} , for by construction $\bar{\partial}v = -i\gamma^{(0,1)}$. Hence $|b|_h^2 = R_h^2 + F_h^2$ is harmonic on \tilde{M} , and so the corresponding $R_h^2 + F_h^2$ on M must be harmonic as well. If M is moreover compact, then $R_h^2 + F_h^2$ is constant by the maximum principle. If $H^1(M; \mathbb{R}) = \{0\}$, then v could be taken initially to be defined on M, and so $e^{2iv}S_h$ is holomorphic on M.

7. Classification of Einstein AH structures by scalar curvature and genus

7.1. Theorem 7.1 is the key technical result for the description of Einstein AH structures on compact orientable surfaces. It generalizes the result for Einstein-Weyl structures proved in [13, Theorem 3.7]. The Killing property of the Gauduchon metric dual of the associated Faraday primitive generalizes to Einstein AH structures a property of the Gauduchon gauge for Einstein-Weyl structures first shown in [68, Theorem 2.2].

Theorem 7.1. A Riemannian signature AH structure $([\nabla], [h])$ on a compact orientable surface is Einstein if and only if for a Gauduchon metric $h \in [h]$ with Levi-Civita connection D and Faraday primitive γ_i there are satisfied the equations

$$D_p \mathcal{L}_{ij}{}^p = 0, \qquad |\mathcal{L}|_h^2 \gamma_i = 0, \qquad (\mathfrak{L}_{\gamma^{\sharp}} h)_{ij} = 2D_{(i} \gamma_{j)} = 0, \qquad (7.1)$$

$$D_i(\mathcal{R}_h - \frac{1}{4}|\mathcal{L}|_h^2 - 4|\gamma|_h^2) = D_i(\mathcal{R}_h - 4|\gamma|_h^2) = 0.$$
(7.2)

Moreover, each of \mathbf{R}_h , $|\mathcal{L}|_h^2$, and \mathbf{F}_h is constant along the flow of γ^{\sharp} .

J

Conversely, if on a manifold of dimension at least 2 (not necessarily compact) there are a Riemannian metric h (with Levi-Civita connection D), an h-Killing field X^i , and a completely symmetric, completely h-trace free tensor $B_{ijk} = B_{(ijk)}$, such that $\gamma_i := X^i h_{pi}$ and $\mathcal{L}_{ij}^{\ k} := h^{kp} B_{ijp}$ solve (7.1)-(7.2), then $\nabla := D - \frac{1}{2} \mathcal{L}_{ij}^{\ k} - 2\gamma_{(i}\delta_{j)}^{\ k} + h_{ij}\gamma^k$ is the aligned representative of an Einstein AH structure ([∇], [h]) for which h is a distinguished metric.

Proof. Let $h \in [h]$ and in this proof raise and lower indices with h_{ij} and h^{ij} . Recall that $\mathcal{L}_{\gamma^{\sharp}}h_{ij} = 2D_{(i\gamma_j)}$. The Ricci identity gives

$$2D_i D^p \gamma_p = 2D^p D_i \gamma_p - \mathcal{R}_h \gamma_i = D^p (\mathfrak{L}_{\gamma^{\sharp}} h)_{ip} - D^p F_{ip} - \mathcal{R}_h \gamma_i.$$
(7.3)
Let $|| \cdot ||_h$ denote the L^2 norm on tensors with respect to the *h*-volume measure $d \operatorname{vol}_h$. Contracting (7.3) with γ^{\sharp} and integrating the result gives

$$2||\mathfrak{L}_{\gamma^{\sharp}}h||_{h}^{2} - 8||d_{h}^{*}\gamma||_{h}^{2} - ||\mathbf{F}_{h}||_{h}^{2} + 4\int_{M}\mathfrak{R}_{h}|\gamma|_{h}^{2}d\operatorname{vol}_{h} = 0.$$
(7.4)

Contracting the second line of (6.2) with $\gamma^{\sharp i}$, integrating by parts, and substituting in (5.8) yields

$$||\mathbf{F}_{h}||_{h}^{2} - 2\int_{M} \mathbf{R}_{h} d_{h}^{*} \gamma \, d \operatorname{vol}_{h} = 4\int_{M} |\gamma|_{h}^{2} \mathbf{R}_{h} \, d \operatorname{vol}_{h}$$

$$= 4\int_{M} \mathcal{R}_{h} |\gamma|_{h}^{2} \, d \operatorname{vol}_{h} - \int_{M} |\gamma|_{h}^{2} |\mathcal{L}|_{h}^{2} \, d \operatorname{vol}_{h} \,.$$
(7.5)

Substituting (7.5) into (7.4) and taking $h \in [h]$ to be Gauduchon yields

$$2||\mathcal{L}_{\gamma^{\sharp}}h||_{h}^{2} + \int_{M} |\gamma|_{h}^{2}|\mathcal{L}|_{h}^{2} d\operatorname{vol}_{h} = 8||d_{h}^{*}\gamma||_{h}^{2} + 2\int_{M} \mathsf{R}_{h}d_{h}^{*}\gamma d\operatorname{vol}_{h} = 0.$$
(7.6)

Equation (7.6) implies the first two equalities of (7.1). By Lemma 3.2 that $|\gamma|_h^2|\mathcal{L}|_h^2 = 0$ is equivalent to $\gamma_p \mathcal{L}_{ij}{}^p = 0$. By (5.7) this implies $D_p \mathcal{L}_{ij}{}^p = 0$. Wherever γ_i is not zero, the one-form $|\mathcal{L}|_h^2 \gamma_i$ is *h*-orthogonal to the linearly independence one-forms γ_i and $J_i{}^p \gamma_p$, so vanishes identically. This completes the proof of (7.1). The first equality in (7.2) is true by (5.8). Because $D_{(i}\gamma_j) = 0$ and $d^*\gamma = 0$ there holds $\Box \gamma_i = d^*d\gamma_i = \mathcal{R}_h\gamma_i$. Substituting the preceeding observations into the last line of (6.2) gives (7.2). By (7.1) and (7.2), $\gamma^{\sharp i} D_i R_h = 8\gamma^{\sharp p}\gamma^{\sharp q} D_{(p}\gamma_{q)} = 0$, showing $dR_h(\gamma^{\sharp}) = 0$. Since γ^{\sharp} is *h*-Killing, there holds $\gamma^{\sharp p} D_p \mathcal{R}_h = 0$, and with $dR_h(\gamma^{\sharp}) = 0$ and (5.8) this shows $\gamma^{\sharp i} D_i |\mathcal{L}|_h^2 = 0$. As $\mathcal{L}_{\gamma^{\sharp}}\omega = d\iota(\gamma^{\sharp})\omega = d \star \gamma = 0$ there holds $2\mathcal{L}_{\gamma^{\sharp}}F = dF_h(\gamma^{\sharp})\omega$. Since $D_{(i}\gamma_j) = 0$ there holds $d|\gamma|_h^2 = 2\gamma^{\sharp p} D_i\gamma_p = \gamma^{\sharp p} d\gamma_{ip} = \gamma^{\sharp p} F_{pi}$. Hence $\mathcal{L}_{\gamma^{\sharp}}F = d(\iota(\gamma^{\sharp})F) = d(d|\gamma|_h^2) = 0$, showing $2\mathcal{L}_{\gamma^{\sharp}}F = dF_h(\gamma^{\sharp})\omega = 0$.

If given (h, X, B) as in the statement of the theorem, then it is straightforward to check that $([\nabla], [h])$ is an AH structure with cubic torsion $\mathcal{L}_{ij}{}^k$, aligned representative ∇ , and Gauduchon metric h. The curvatures of ∇ and D are related as in (5.8), and there hold (5.7) and (6.2). Together (7.1) and (5.7) show $E_{ij} = 0$, and so show the naive Einstein equations. Finally, substituting $D_i |\gamma|_h^2 = \gamma^{\sharp p} d\gamma_{ip}$ and $D^p d\gamma_{pi} = -\mathcal{R}_h \gamma_i$ into (6.2) yields $\nabla_i R + 2\nabla^p F_{ip} = 0$.

Remark 7.2. In Section 9.4 it is explained how to construct naive Einstein AH structures which are not Einstein, and which illustrate that not all of the conclusions of Theorem 7.1 hold for such structures.

Corollary 7.3. If $([\nabla], [h])$ is a Riemannian signature Einstein AH structure on a compact oriented surface M of genus g and $h \in [h]$ is a Gauduchon metric with associated Faraday primitive γ_i and cubic form $B_{ijk} := \mathcal{L}_{ij}{}^p h_{pk}$, then:

(1) With respect to the complex structure determined by [h] and the given orientation, $B^{(3,0)}$ and $\gamma^{\sharp(1,0)}$ are holomorphic. Moreover,

$$2\Delta_h |\gamma|_h^2 = |d\gamma|_h^2 - 2\mathbf{R}_h |\gamma|_h^2, \qquad \text{everywhere,}$$
(7.7)

$$\Delta_h \log |\gamma|_h^2 = -\mathbf{R}_h, \qquad \qquad \text{wherever } |\gamma|_h^2 > 0, \qquad (7.8)$$

$$\Delta_{h}|B|_{h}^{2} = 4|d|B||_{h}^{2} + 3\Re_{h}|B|_{h}^{2}$$

= 4|d|B||_{h}^{2} + 3R_{h}|B|_{h}^{2} + \frac{3}{4}|B|_{h}^{4}, \qquad \text{wherever } |B|_{h}^{2} > 0, \qquad (7.9)

$$\Delta_h \log |B|_h^2 = 3\Re_h = 3\Re_h + \frac{3}{4}|B|_h^2 \qquad \text{wherever } |B|_h^2 > 0.$$
(7.10)

- (2) If $([\nabla], [h])$ is not Weyl then it is exact, while if $([\nabla], [h])$ is not exact, then it is Weyl.
- (3) If $g \ge 2$ then ($[\nabla]$, [h]) is exact. In this case the metric ${}^*h_{ij} := |B|_h^{2/3}h_{ij}$ defined on the open submanifold $M^* := \{|B|^2 \ne 0\}$ is flat.
- (4) If g = 0 then ($[\nabla]$, [h]) is Weyl.

Proof. By the first and last equalities of (7.1) and Lemmas 3.3 and 3.5, $\gamma^{\sharp(1,0)}$ and $B^{(3,0)}$ are holomorphic. Since by (7.1) of Theorem 7.1, $d\gamma_{ij} = 2D_i\gamma_j$, the second equation of (3.12) of Lemma 3.6 applied to γ^{\sharp} (so with k = -1) reduces to (7.7), while by the second equation of (3.13) there holds $2\Delta_h \log |\gamma| = -\Re_h = -\Re_h - \frac{1}{2}|\mathcal{L}|_h^2$ wherever γ is non-zero. Multiplying through by $|\gamma|^2$ and using that, by Theorem 7.1, $|\gamma|^2|\mathcal{L}|^2 = 0$, this yields (7.8). Similarly, equations (7.9) and (7.10) follow from (3.12) and (5.8).

Since each of $\gamma^{\sharp (1,0)}$ and $B^{(3,0)}$ is holomorphic, the zeroes of each are isolated if it is not identically zero, and hence the same is true for γ^{\sharp} and *B*. By (7.1), there holds $|\gamma|_{h}^{2}|B|_{h}^{2} = 0$. Because when γ^{\sharp} or *B* is not zero its zeroes are isolated, this implies that if γ^{\sharp} is somewhere not zero then *B* is identically zero, and if *B* is not somewhere zero then *X* is identically zero. Hence if ($[\nabla], [h]$) is not Weyl it is exact and if ($[\nabla], [h]$) is not exact it is Weyl.

That $g \ge 2$ and g = 0 imply that $([\nabla], [h])$ is respectively exact and Weyl follows from the holomorphicity of $B^{(3,0)}$ and $\gamma^{\sharp(1,0)}$ and Riemann Roch. By Riemann Roch, when g > 1, a holomorphic cubic differential has at most 6(g - 1) zeroes, so M^* is the complement of at most 6(g - 1) points. By (7.10) of Corollary 7.3 there holds $|B|_h^{2/3} \mathcal{R}_{*h} = \mathcal{R}_h - \Delta_h \log |B|_h^{2/3} = 0$, so *h is flat on M^*

Lemma 7.4. If $([\nabla], [h])$ is a Riemannian signature Einstein AH structure on a compact orientable surface M, and $h \in [h]$ is any Gauduchon metric, then $R_h^2 + F_h^2$ is equal to the constant $(\max_M R_h)^2$.

Proof. If ($[\nabla]$, [h]) is exact then R_h is constant by (6.1), and $R_h^2 + F_h^2 = R_h^2 = (\max_M R_h)^2$. If ($[\nabla]$, [h]) is not exact, then by Corollary 7.3, M is a sphere or a

torus. In this case, by Lemma 6.7, $\beta = R_h^2 + F_h^2$ is constant. Since $|\gamma|_h^2$ is not identically zero, it assumes somewhere a positive maximum; at such a point there holds $0 = 2d_i |\gamma|_h^2 = 4\gamma^{\sharp p} D_i \gamma_p = 2\gamma^{\sharp p} F_{pi} = F_h(\star \gamma)_i$, so at such a point F_h vanishes, and β is equal to the value of R_h^2 at this point. Since by (7.2) of Theorem 7.1, $R_h - 4|\gamma|_h^2$ is constant, the functions R_h and $|\gamma|_h^2$ assume their maximum values at the same points, so $\beta = (\max_M R_h)^2$.

7.2. This section records a geometric interpretation of the integral curves of the Gauduchon metric dual of the Faraday primitive of the Gauduchon class. Recall that the **magnetic flow** determined on T^*M by a pair (g, μ) comprising a metric g and a closed two-form μ on M is the Hamiltonian flow of the function $G(s) = \frac{1}{2}g_{\pi(s)}^{ij}s_is_j$ on T^*M with respect to the symplectic form $\Omega_M - \pi^*(\mu)$, where $\pi : T^*M \to M$ is the canonical projection and Ω_M is the canonical symplectic form. If μ is exact the magnetic flow is said to be **exact**.

The **magnetic geodesics** are the images in M of the integral curves of this flow, which are the solutions of the equation $D_{\dot{\sigma}}\dot{\sigma} = A(\dot{\sigma})$ where A is the tensor defined by $A_i {}^p g_{pj} = \mu_{ij}$, D is the Levi-Civita connection of g, and $\sigma(t)$ is a curve in M. Along a magnetic geodesic $\sigma(t)$, the energy $|\dot{\sigma}|_g^2$ is constant. If M is an oriented surface then $\mu_{ij} = f \omega_{ij}$, where ω_{ij} is the Kähler form associated to g, and a routine computation shows that the geodesic curvature $\kappa_g(\sigma)$ of a magnetic geodesic σ having energy $|\dot{\sigma}|_g^2 = e^{2c}$ is $e^{-c} f(\sigma)$.

Theorem 7.5. Let $([\nabla], [h])$ be a non-exact Einstein AH structure on a compact orientable surface M. Let $h \in [h]$ be a Gauduchon metric with Faraday primitive γ_i . The integral curves of the vector field γ^{\sharp} are magnetic geodesics for the exact magnetic flow on T^*M determined by $(h, \frac{1}{2}d\gamma)$. The function $|\gamma|_h^2$ is constant along a non-trivial integral curve of γ^{\sharp} , and the geodesic curvature of such an integral curve is the restriction to the curve of $-\frac{1}{4}|\gamma|_h^{-1}F_h$, which is constant along the integral curve. The integral curves of γ^{\sharp} passing through the points where $|\gamma|_h^2$ attains its maximum value are D-geodesics; in particular, at least one integral curve of γ^{\sharp} is a D-geodesic. The non-trivial integral curves of $J(\gamma^{\sharp}) = (\star \gamma)^{\sharp}$ are projective geodesics of D.

Proof. By Corollary 7.3, $([\nabla], [h])$ is Weyl and *M* is a torus or a sphere. From (7.1) there follows

$$-4D_{\gamma^{\sharp}}\gamma^{\sharp} = \mathcal{F}_h J(\gamma^{\sharp}) = \mathcal{F}_h(\star\gamma)^{\sharp} = 2(d|\gamma|_h^2)^{\sharp}.$$
(7.11)

The tensor A_i^{j} such that $A_i^{p}h_{pj} = -\frac{1}{2}F_{ij} = -\frac{1}{4}F_h\omega_{ij}$ is $-\frac{1}{4}F_hJ_i^{j}$, so it follows from the first equality of (7.11) that the integral curves of γ^{\sharp} are magnetic geodesics. By the remarks preceeding the statement of the lemma, the geodesic curvature $\kappa_h(\sigma)$ of an integral curve $\dot{\sigma}(t) = \gamma^{\sharp}_{\sigma(t)}$ is $-\frac{1}{4}F_h|\dot{\sigma}|_h^{-1}$, which is constant along σ by Theorem 7.1. Because ([∇], [h]) is not exact, $|\gamma|_h^2$ assumes a

positive maximum at some point of M. Since, by (7.11), $2d|\gamma|_h^2 = F_h \star \gamma$, at such a point it must be that $F_h = 0$. Thus F_h has a zero which is not a zero of γ , and the integral curve of γ^{\sharp} passing through this zero of F_h has geodesic curvature 0, so is a D-geodesic. Because γ^{\sharp} is conformal Killing there holds $\mathcal{L}_{\gamma^{\sharp}}J = 0$, from which follows $[\gamma^{\sharp}, J\gamma^{\sharp}] = 0$. Using this and (7.11) yields $D_{J\gamma^{\sharp}}J\gamma^{\sharp} = J(D_{J\gamma^{\sharp}}\gamma^{\sharp}) = J(D_{\gamma^{\sharp}}J\gamma^{\sharp}) = -D_{\gamma^{\sharp}}\gamma^{\sharp} = \frac{1}{4}F_h J\gamma^{\sharp}$, which shows that the non-trivial integral curves of $J\gamma^{\sharp}$ are projective geodesics of D.

This suggests viewing these particular exact magnetic flows as *Einstein* and raises the question of whether there is a good notion of Einstein magnetic flows in higher dimensions.

Theorem 7.5 was suggested by the explicit models of Einstein-Weyl structures described in Section 10, from which more precise information can be extracted. From the discussion following equation (10.20) at the end of Section 10.11 it follows that (with the setting and notations as in the statement of Theorem 7.5) in the case of the sphere the vector field γ^{\sharp} has two zeroes, and its integral curves, which by Theorem 7.5 are magnetic geodesics, are simple closed circles separating these zeroes. Among these simple closed magnetic geodesics, there is a unique one of maximal energy on which $|\gamma|_h^2$ attains its maximum value and F_h vanishes, and which is moreover a *D*-geodesic, while for each positive energy below this maximum there are precisely two simple closed magnetic geodesics occurring as integral curves of γ^{\sharp} .

7.3. If a Riemannian signature AH structure $([\nabla], [h])$ on a compact, oriented surface *M* is exact and Weyl, then the aligned representative ∇ is the Levi-Civita connection of any Gauduchon metric $h \in [h]$, and (5.8) shows that $R_h = \mathcal{R}_h$, so that $([\nabla], [h])$ is moreover Einstein just when *h* has constant scalar curvature, that is *h* is a constant curvature metric. Thus in this case $([\nabla], [h])$ is naturally identified with a positive homothety class of constant curvature metrics; since [h] contains a unique such class, $([\nabla], [h])$ can be identified with [h], and such an $([\nabla], [h])$ will be said simply to be *a conformal structure*. In this case the weighted scalar curvature is parallel, and positive, zero, or negative according to whether the genus *g* is 0, 1, or at least 2. Alternatively, such an AH structure will be said to be *generated* by its representative constant curvature metric.

7.4. Since, for a $\binom{p}{q+1}$ tensor Q, rescaling h homothetically by $r \in \mathbb{R}^+$ causes $|Q|_h^2 |\det h|^{1/2}$ to rescale by r^{p-q} , on a compact surface the L^2 -norm $||Q||_h^2 = \int_M |Q|_h^2 d \operatorname{vol}_h$ of a $\binom{p}{p+1}$ tensor Q depends only on the conformal class of h, and not on the choice of h itself, so in this case it makes sense to write $||Q||^2 = ||Q||_h^2$, dropping the subscript indicating dependence on h. Let $([\nabla], [h])$ be a Riemannian Einstein AH structure $([\nabla], [h])$ on a compact, oriented surface M. Let $h \in [h]$ be a Gauduchon metric with associated Faraday primitive γ_i . Then $||\mathcal{L}||_h^2 = \int_M |\mathcal{L}|_H^2$ and $||\gamma||_h^2$ are unchanged if h is homothetically rescaled. Although the L^2 norm of

a vector field is changed by rescaling *h*, the L^2 norm $||\gamma^{\sharp}||_h^2 = ||\gamma||_h^2$ is not because γ^{\sharp} also rescales when *h* is rescaled, in a way that compensates for the change in norm. Similarly for the cubic form $B_{ijk} := \mathcal{L}_{ij} {}^p h_{pk}$ there holds $||B||_h^2 = ||\mathcal{L}||^2$. Whether there will be written $||\gamma||^2$, $||\gamma^{\sharp}||_h^2$, $||B||_h^2$, or $||\mathcal{L}||^2$ will depend on the emphasis desired. From (7.2) it follows that there is a constant κ such that $R_h = 4|\gamma|_h^2 + \kappa$. This κ does depend on *h* in that rescaling *h* by $r \in \mathbb{R}^+$ rescales κ by r^{-1} . Integrating (5.8) against *d* vol_h using the Gauß-Bonnet theorem yields

$$8\pi(1-g) = \int_{M} \mathcal{R}_{h} d \operatorname{vol}_{h} = \frac{1}{4} ||\mathcal{L}||^{2} + \int_{M} \mathcal{R}_{h} d \operatorname{vol}_{h}$$

= $\frac{1}{4} ||\mathcal{L}||^{2} + 4 ||\gamma||^{2} + \kappa \operatorname{vol}_{h}(M),$ (7.12)

in which g is the genus of M. The number $\nu := \kappa \operatorname{vol}_h(M)$ does not depend on the choice of Gauduchon metric. For reasons explained in Section 8, it is called the **vortex parameter** of the Riemannian Einstein AH structure on the compact, oriented surface M. Lemma 7.6 follows from (7.12).

Lemma 7.6. For a Riemannian Einstein AH structure $([\nabla], [h])$ on a compact, orientable surface of genus g, the vortex parameter v satisfies $v \leq 8\pi(1-g)$, with equality if and only if $([\nabla], [h])$ is the AH structure generated by a constant curvature metric.

If $([\nabla], [h])$ is exact Einstein, $\kappa = R_h$ so $\nu = R_h d \operatorname{vol}_h(M) = \int_M R_h d \operatorname{vol}_h = \int_M R$ is the total weighted scalar curvature. In particular, this holds if the genus is greater than 1, or the genus equals 1 and $([\nabla], [h])$ is not Weyl.

7.5. Combining Theorem 7.1, Corollary 7.3, Lemma 7.6, and making some arguments to relate topological conditions with the sign of the weighted scalar curvature yields Theorem 7.7, which is the principal structural result about Riemannian Einstein AH structures on compact surfaces.

Recall from the introduction the definition of a (strictly) convex flat real projective structure.

Theorem 7.7. For a Riemannian signature Einstein AH structure $([\nabla], [h])$ on a compact oriented surface M of genus g there holds one of the following mutually exclusive possibilities.

- (1) ($[\nabla]$, [h]) is exact and Weyl, so identified with the unique positive homothety class of constant curvature metrics contained in the underlying conformal structure [h]. There holds $v = 8\pi(1 g)$.
- (2) *R* is negative and parallel, ([∇], [h]) is exact and not Weyl, and g ≥ 2. ([∇], [h]) is projectively flat and conjugate projectively flat, and both [∇] and [∇] are strictly convex. A distinguished metric h ∈ [h] has scalar curvature of the form

$$\mathcal{R}_{h} = \frac{1}{4} \left(|\mathcal{L}|_{h}^{2} - \frac{||\mathcal{L}||^{2}}{\operatorname{vol}_{h}(M)} \right) + \frac{8\pi (1-g)}{\operatorname{vol}_{h}(M)},$$
(7.13)

and \Re_h is everywhere non-positive and somewhere negative. The cubic differential $B^{(3,0)}$, where $B_{ijk} := \mathcal{L}_{ij} {}^p h_{pk}$, is holomorphic. On the open submanifold $M^* := \{|B|^2 \neq 0\}$, the metric ${}^*h_{ij} := |B|_h^{2/3} h_{ij}$ is flat. There holds $\nu < 8\pi(1-g)$.

- (3) *R* is negative and parallel, ([∇], [h]) is exact and not Weyl, and M is a torus. A distinguished metric h is flat and B_{ijk} := L_{ij} ^ph_{pk} is parallel with respect to the Levi-Civita connection of h, so, in particular, has constant non-zero norm. ([∇], [h]) is projectively flat and conjugate projectively flat, and both [∇] and [∇] are convex but not strictly convex. There holds v = -1/4||B||_h² < 0.
- (4) R = 0, $([\nabla], [h])$ is Weyl and closed but not exact, and M is a torus. A Gauduchon metric $h \in [h]$ is flat, the Faraday primitive γ_i of h is parallel with respect to the Levi-Civita connection of h, and $\gamma^{(1,0)}$ is holomorphic. The aligned representative of $([\nabla], [h])$ and is affinely flat, and its (1, 0) part is a holomorphic affine connection. There holds $v = -4||\gamma||_h^2 < 0$.
- (5) *R* is somewhere positive and somewhere negative, ([∇], [h]) is Weyl and not closed, and *M* is a torus. For a Gauduchon metric h ∈ [h] with Faraday primitive γ_i the scalar curvature is R_h = 4(|γ|²_h ||γ||²), and there holds v = -4||γ||² < 0.
- (6) *R* is somewhere positive, $([\nabla], [h])$ is Weyl and not closed, and *M* is a sphere. A Gauduchon metric $h \in [h]$ with Faraday primitive γ_i has scalar curvature of the form

$$\mathcal{R}_{h} = \mathcal{R}_{h} = 4\left(|\gamma|_{h}^{2} - \frac{||\gamma||^{2}}{\mathrm{vol}_{h}(M)}\right) + \frac{8\pi}{\mathrm{vol}_{h}(M)}.$$
(7.14)

On the open submanifold $M^* := \{|\gamma|^2 \neq 0\}$, the metric ${}^*\!h_{ij} := |\gamma|_h^{-2}h_{ij}$ is flat. There holds $\nu = 8\pi - 4||\gamma||_h^2 < 8\pi$.

Proof. In this proof $h \in [h]$ is always a Gauduchon metric with Faraday primitive γ_i , and $B_{ijk} = \mathcal{L}_{ij} {}^p h_{kp}$. By (7.2), $\kappa := R_h - 4|\gamma|_h^2$ is constant. The theorem follows by assembling the following claims.

- [a] If ($[\nabla]$, [h]) is not exact then g < 2, by Corollary 7.3, and R is not everywhere negative, for, by (7.8), at a point at which $|\gamma|$ attains its maximum value there holds $R_h = -2\Delta_h \log |\gamma| \ge 0$.
- [b] If ([∇], [*h*]) is closed, then, by (6.1), *R* is parallel, and it follows from Lemma 6.4 that ([∇], [*h*]) is projectively flat and conjugate projectively flat. If, more-over, *R* = 0, then by (5.3) the aligned representatives of ([∇], [*h*]) and its conjugate are flat.
- [c] If $([\nabla], [h])$ is not exact and *R* is non-positive, then, by the maximum principle applied to (7.7), γ is closed, and so by (7.1), γ is parallel. In particular $([\nabla], [h])$ is closed and $|\gamma|_h^2$ is a non-zero constant. By (7.8) this forces $R \equiv 0$. Since *R*, E_{ij} and F_{ij} all vanish, the aligned representative $\nabla \in [\nabla]$ is affinely flat. By Corollary 7.3, since $([\nabla], [h])$ is not exact, it is Weyl, and so $\mathcal{R}_h = R_h$

by (5.8), which by Gauß-Bonnet forces g = 1. Hence M is a torus and a Gauduchon metric is flat with parallel Faraday primitive. By (3.11) there holds $-2\bar{\partial}\gamma^{(1,0)} = F^{(1,1)} + d_h^*\gamma \bar{h}^{(1,1)} = 0$, so the one-form $\gamma^{(1,0)}$ is holomorphic, and hence the complex affine connection $\nabla^{1,0} = D^{1,0} - 2\gamma^{(1,0)}$ is holomorphic.

- [d] If R is non-positive and somewhere negative, then $([\nabla], [h])$ must be exact by [c], and so R is parallel by (6.1). Since R is somewhere negative and parallel, it is strictly negative. Since \mathbb{R}_h is constant, $\mathcal{R}_h = \mathbb{R}_h + \frac{1}{4}|B|_h^2$ and $|B|_h^2$ assume their respective maximum values on M at the same points. At a point at which $|B|^2$ takes its maximum value, (7.9) yields $0 \ge \Delta_h |B|_h^2 = 3\mathcal{R}_h |B|_h^2$, so at such a point $\mathcal{R}_h \le 0$. Since such a point is also a maximum of \mathcal{R}_h , this shows $\mathcal{R}_h \le 0$. Solving (7.12) for κ shows that \mathcal{R}_h has the form (7.13). If \mathcal{R}_h is identically zero, integrating (7.13) shows that g = 1; hence if g > 1 then \mathcal{R}_h must be somewhere negative. Conversely, if g = 1 then since $\mathcal{R}_h \le 0$, integrating (7.13) shows that $\mathcal{R}_h = 4\mathbb{R}_h + |B|_h^2$, so $|B|_h^2$ is constant, equal to $-4\mathbb{R}_h$.
- [e] If g = 1 and $([\nabla], [h])$ is exact then $R \le 0$ by (7.12). Since R is parallel, either R is identically 0, in which case $([\nabla], [h])$ is Weyl, or R is negative, in which case R_h is a negative constant. Since R_h is constant, $\mathcal{R}_h = R_h + \frac{1}{4}|B|_h^2$ and $|B|_h^2$ assume their maximum values at the same points. By the same argument as in [d] there holds $\mathcal{R}_h \le 0$, and by Gauß-Bonnet this means \mathcal{R}_h is identically zero. Since R_h is constant, this implies $|B|_h^2$ is constant. Since $B^{(3,0)}$ is holomorphic, it follows from (3.13) of Lemma 3.6 that $0 = 2|d|B||^2 = |DB|_h^2$, so that B is parallel with respect to h.
- [g] If $g \ge 2$, ([∇], [h]) is exact by Corollary 7.3 and hence R is parallel. By (7.12), R is negative.
- [g] If g = 0, ($[\nabla]$, [h]) is Weyl by Corollary 7.3. Solving (7.12) for κ and substituting into $\Re_h = \Re_h = 4|\gamma|_h^2 + \kappa$ yields (7.14). Since by (7.12), $\int_M R = 8\pi$, R must be positive somewhere.
- [h] If *R* is somewhere negative and somewhere positive then it is not parallel, and so by (6.1) ([∇], [*h*]) is not closed. Hence, by Corollary 7.3, ([∇], [*h*]) is Weyl. In this case, since R_h is somewhere negative, v is negative.
- [i] If g = 1 and $([\nabla], [h])$ is not exact, then it is Weyl. By (7.12), $0 = \int_M R$, so either *R* is identically 0, or *R* is somewhere positive and somewhere negative. In the latter case, $([\nabla], [h])$ is not closed by [h], and so from (7.12) it follows that $v = -4||\gamma||_h^2 < 0$.
- [j] If *R* is non-negative and somewhere positive then by (5.8) the same is true of *R_h*, and so *g* = 0 by Gauß-Bonnet and thus ([∇], [*h*]) is Weyl by Corollary 7.3. If ([∇], [*h*]) is moreover closed then the Einstein condition implies *R* is parallel, and so everywhere positive, and hence ([∇], [[*h*]) is exact by Lemma 6.3. Thus in this case ([∇], [*h*]) is exact Weyl and a distinguished metric has constant curvature.
- [k] That the flat projective structures in ({2}) and ({3}) are properly convex follows from Theorem 11.4, which is deferred to Section 11.8 because its proof

uses a point of view more conveniently introduced later. That the projective structures of ({2}) are strictly convex while those of ({3}) are not follows from Theorem 1.1 of Y. Benoist ([4]), that a discrete group which divides some properly convex domain in the projective sphere is Gromov hyperbolic if and only if the domain is strictly convex, coupled with the observation that the fundamental group of a surface of genus g > 1 is Gromov hyperbolic, while that of the torus is not.

In Section 9 it will be shown that all the possibilities identified in Theorem 7.7 actually occur.

7.6. Item [d] of the proof of Theorem 7.7 shows that for an Einstein AH structure on a surface of genus g > 1 the norm squared of the cubic torsion with respect to a Gauduchon metric $h \in [h]$ satisfies the equivalent pointwise bounds

$$|B|_{h}^{2} = -4R_{h} + 4\mathcal{R}_{h} \le -4R_{h}, |B|_{h}^{2} - ||B||_{h}^{2} \operatorname{vol}_{h}(M)^{-1} \le 2\pi (g-1) \operatorname{vol}_{h}(M)^{-1}.$$
(7.15)

The pull-back of such an Einstein AH structure to the universal cover of M can be identified with that induced on an affine hypersphere asymptotic to the cone over the developing map image of the universal cover (see Section 11.8), and then the non-positivity of the scalar curvature of a Gauduchon metric is the conclusion of the main theorem (Theorem 5.1) of Calabi's [9]. Here this non-positivity has been given a direct, autonomous proof. Based on these considerations it seems reasonable to expect the following.

Theorem 7.8. On an oriented surface M, let $([\nabla], [h])$ be an exact Einstein AH structure with negative weighted scalar curvature for which a distinguished metric is complete. Then the curvature of a distinguished metric is non-positive.

Proof. Since the curvature of a distinguished metric $h \in [h]$ is $\mathbb{R}_h + \frac{1}{4}|B|_h^2$ (the notations are as usual), it is equivalent to show that the function $u := |B|_h^2$ is not greater than the constant $-4\mathbb{R}_h$. There holds (7.9), and so u satisfies a differential inequality of the form $\Delta_h u \ge Bu^2 - Au$ where $A = -3\mathbb{R}_h$ and B = 3/4. A theorem of Cheng and Yau, from [20] and [24], a statement and proof of which can be found as [30, Theorem 6.6], shows that for a complete metric g with Ricci curvature bounded from below, a differential inequality of the form $\Delta_g v \ge bv^{1+\sigma} - av$, with positive b and holding where the non-positive smooth function v is positive, implies an upper bound $v \le |a/b|^{\sigma}$. Applying this with $\sigma = 1$, v = u, a = A and b = B yields the desired bound.

There is no similar lower bound for \mathcal{R}_h in the case {6} of Theorem 7.7. Suppose the Einstein AH structure ($[\nabla], [h]$) is not exact, so is Weyl, and M must be a sphere or a torus. If the constant $\kappa = R_h - 4|\gamma|_h^2$ is negative then the constant function $\log(4^{-1}|\kappa|)$ solves

$$\Delta_h \phi + \kappa + 4e^{\phi} = 0. \tag{7.16}$$

By (7.8) the function $\psi = \log |\gamma|_h^2$ solves (7.16) on the complement M^* of the zero set of γ , which is discrete because $h^{ip}\gamma_p$ is the real part of a holomorphic vector field. Since ψ tends to $-\infty$ at the zeroes of γ it is tempting to conclude that ψ is bounded from above by $\log(4^{-1}|\kappa|)$, but the standard comparison argument fails because in the operator $\Delta_h \phi + \kappa + 4e^{\phi}$ the zeroth order term is increasing in ϕ , while the routine application of the maximum principle needs it to be non-increasing. In Section 10.11 it is shown explicitly how to construct an Einstein-Weyl structure on the two sphere \mathbb{S}^2 for which κ has an arbitrarily negative value.

8. Relation with the Abelian vortex equations

In this section it is shown that an exact Einstein AH structure determines a solution of the Abelian vortex equations. On the other hand, an Einstein-Weyl structure gives rise to a solution of equations superficially similar to the vortex equations, but differing from them by a change of sign in one term.

Let *M* be a compact manifold and let (g, J, ω) be a Kähler structure on *M*. Let \mathcal{E} be a smooth complex line bundle over *M*. The **Abelian vortex equations** with parameter τ are the following equations for a triple (∇, k, s) , in which *k* is a Hermitian metric on \mathcal{E} , ∇ is a Hermitian connection on (\mathcal{E}, k) , and *s* is a smooth section of \mathcal{E} :

$$\Omega^{(0,2)} = 0, \qquad \bar{\partial}_{\nabla} s = 0, \qquad i\Lambda(\Omega) + \frac{1}{2}|s|_{k}^{2} = \frac{1}{2}\tau. \tag{8.1}$$

Here Ω_{ij} is the curvature of ∇ , viewed as a real-valued two-form on M; $\bar{\partial}_{\nabla}$ is the (0, 1) part of ∇ ; and Λ is the dual Lefschetz operator given on (1, 1) forms by $\Lambda(A) = -\omega^{\alpha\bar{\beta}}A_{\alpha\bar{\beta}} = -iA_{\sigma}^{\sigma}$. The first two equations say that $\bar{\partial}_{\nabla}$ is a holomorphic structure on \mathcal{E} with respect to which *s* is a holomorphic section, while the third equation is something like an Einstein equation. A solution of (8.1) is **non-trivial** if *s* is not identically zero. The trivial solutions correspond to holomorphic structures on \mathcal{E} ; a precise statement is [5, Theorem 4.7].

The basic theorem about the Abelian vortex equations on a surface is the following.

Theorem 8.1 ([5,33,61]). Let M be a compact surface equipped with a Kähler metric (g, J). Let \mathcal{E} be a smooth complex line bundle with a fixed Hermitian metric k. Let D be an effective divisor of degree equal to deg (\mathcal{E}) . There exists a non-trivial solution (s, ∇) of the vortex equations (8.1), unique up to gauge equivalence, if and only if $4\pi \deg(\mathcal{E}) < \tau \operatorname{vol}_g(M)$. Moreover the holomorphic line bundle and section canonically associated to D are $(\mathcal{E}, \overline{\partial}_{\nabla})$ and s.

The space of effective divisors on M of a given degree r is the symmetric product $S^{r}(M)$ of M and Theorem 8.1 shows that $S^{\deg(\mathcal{E})}(M)$ is in bijection with the moduli space of gauge equivalence classes of vortex solutions on \mathcal{E} . It is shown in [33] by symplectic reduction, that this moduli space carries a Kähler structure.

Note that *a priori* the Kähler metric g_{ij} and the Hermitian metric k on \mathcal{E} are unrelated. However, in what follows the interest will be on solutions for which \mathcal{E} qua holomorphic bundle is identified with a power \mathcal{K}^p of the canonical bundle, and k and ∇ are the Hermitian metric and Hermitian connection induced by the underlying Kähler metric and connection on M. Equivalently, the corresponding effective divisor is a *p*-canonical divisor, that is it is in the *p*-fold product of the canonical divisor class. This motivates the following definition. A solution (s, ∇) of the Abelian vortex equations on (M, g, J) is *p*-canonical if the divisor of *s* is *p*-canonical; equivalently *s* is a section of \mathcal{K}^p holomorphic with respect to the complex structure induced by *J*, and ∇ is the Hermitian connection induced on \mathcal{K}^p by the Levi-Civita connection of *g*.

Let *M* be a compact orientable surface of genus *g* and let ($[\nabla]$, [h]) be an exact Riemannian signature Einstein AH structure. Let $h \in [h]$ be a Gauduchon metric and $B_{ijk} = \mathcal{L}_{ij}{}^p h_{pk}$. Let the Kähler structure on *M* be that determined by *h* and the given orientation. Let k = h be the Hermitian metric induced on \mathcal{K}^3 by *h*. The Levi-Civita connection *D* of *h* induces a Hermitian connection, also denoted by *D*, on \mathcal{K}^3 , for which the induced holomorphic structure is the canonical one. It is claimed that for $\tau = -3R_h$, the section $s = (3/2)^{1/2}B^{(3,0)}$ of \mathcal{K}^3 solves the vortex equations (8.1). Note that $|s|_h^2 = (3/2)|B^{(3,0)}|^2 = (3/4)|\mathcal{L}|_h^2$. The second equation of (8.1) is valid by construction. The curvature of *D* on \mathcal{K}^3 is $\Omega_{ij} = (3i\mathcal{R}_h/2)\omega_{ij}$, so that $\Omega^{(2,0)} = 0$ and $\Lambda(\Omega) = (3i/2)\mathcal{R}_h$. Because $([\nabla], [h])$ is exact, R_h is constant, and it follows from (5.8) that $3\mathcal{R}_h = 3(R_h + \frac{1}{4}|\mathcal{L}|_h^2) = -\tau + |s|_h^2$. Hence

$$i\Lambda(\Omega) + (1/2)|s|_k^2 - \frac{1}{2}\tau = -(3/2)\left(\Re_h - \frac{1}{4}|B|_h^2 - \Re_h\right) = 0,$$

which shows the claim. Note that the resulting solution to the vortex equations is trivial exactly when $([\nabla], [h])$ is simply a conformal structure. In this case the vortex parameter ν is $-\tau/3$, which explains the terminology for ν .

By Corollary 7.3 if a Riemannian signature Einstein AH structure on a compact surface is not exact then it is Weyl. In this case, let $h \in [h]$ be a Gauduchon metric and $X^i = h^{ip}\gamma_p$, where γ_i is the Faraday primitive of h. Let the Kähler structure on M be that determined by h and the given orientation. Let k = h be the Hermitian metric induced on \mathcal{K}^{-1} by h. The Levi-Civita connection D of hinduces a Hermitian connection, also denoted by D, on \mathcal{K}^{-1} , for which the induced holomorphic structure is the canonical one. By Theorem 7.1 there is a constant κ such that $\mathcal{R}_h = \mathcal{R}_h = \kappa + 4|X|_h^2$. Let $\tau = \kappa$ and consider the section $s = 2^{3/2}X^{(1,0)}$ of \mathcal{K}^{-1} . The curvature of D on \mathcal{K}^{-1} is $\Omega_{ij} = -(i\mathcal{R}_h/2)\omega_{ij}$, so that $\Omega^{(2,0)} = 0$ and $\Lambda(\Omega) = -(i/2)\mathcal{R}_h$. Hence

$$i\Lambda(\Omega) - (1/2)|s|_k^2 - \frac{1}{2}\tau = (1/2)\left(\mathcal{R}_h - 4|X|_k^2 - \kappa\right) = 0.$$
 (8.2)

The equation (8.2) differs from (8.1) by the change of sign on the $|s|_k^2$ term. This affects the interpretation of the equations. The vortex equations are the adaptation to compact surfaces of the Landau-Ginzburg equations modeling the macroscopic

behavior of superconducting materials. In the context of superconductors the wrong sign on the $|s|_k^2$ term corresponds to a physically unreasonable negative sign on the quartic term in the free energy. While the equations (8.2) by themselves make sense, and, as follows from Section 10, have solutions, they are not the usual vortex equations, and a more satisfactory derivation of them is needed before any significance can be attached to their similarity with the vortex equations.

Nonetheless, the preceeding can be given the following uniform, albeit unmotivated description. Let the setup be as in the two preceeding paragraphs. Let q be either -1 or 3 and let σ be X or B. Let κ be the constant such that $R_h = \kappa + 4|\gamma|_h^2$. Let $\tau = -q\kappa$ and $s = 2^{(2-q)/2}|q|^{1/2}\sigma^{(k,0)}$. Then s is a section of \mathcal{K}^q solving the following modified vortex equations with respect to the holomorphic structure and Hermitian metric and connection on \mathcal{K}^q induced by h:

$$\Omega^{(0,2)} = 0, \qquad \bar{\partial}_{\nabla} s = 0, \qquad 2i\Lambda(\Omega) + \operatorname{sgn}(q)|s|_k^2 = \tau.$$
(8.3)

Note that distinct exact Einstein AH structures need not determine gauge inequivalent solutions of (8.1). If $B^{(3,0)}$ is replaced by $e^{i\theta}B^{(3,0)}$ for a constant θ , the resulting vortex solutions are gauge equivalent. In Section 9.7 it is shown that the real part of $e^{i\theta}B^{(3,0)}$ is the cubic torsion of an Einstein AH structure with the same underlying conformal structure and Gauduchon metric. The solutions to the Abelian vortex equations with deg(D) = $3\chi(M)$ which arise in this way from exact Einstein AH structures are exactly the 3-canonical solutions. This essentially means that on a compact, orientable surface of genus g > 1 the quotient of the moduli space of Einstein AH structures by the action of \mathbb{C}^{\times} is the space of gauge equivalence classes of 3-canonical Abelian vortices; see Section 9 for related discussion.

The existence of p-canonical Abelian vortices is not immediate from Theorem 8.1; it is demonstrated in Corollary 9.5. In particular, the existence of Einstein AH structures as in (2) of Theorem 7.7 is not immediate from the existence theorem for Abelian vortices; the complication is the additional requirement of compatibility of the underlying Kähler structure (the putative Gauduchon metric) and the holomorphic and metric structure on the line bundle.

However, note that the constraint on the vortex parameter given by Lemma 7.6 also follows from Theorem 8.1 and the reinterpretation of the Einstein AH equations in terms of the vortex equations.

9. Einstein AH structures on compact orientable surfaces of genus at least two

In this section and the next the classification of Riemannian Einstein AH structures on compact, orientable surfaces is completed by showing that all the possibilities identified in Theorem 7.7 are realized. Throughout M is a compact, orientable surface of genus g.

Theorem 7.7 shows that a Riemannian Einstein AH structure on M must determine data of one of the following forms

- (1) A constant curvature metric. (Any g).
- (2) A conformal structure and a non-trivial holomorphic vector field. $(g \in \{0, 1\})$.
- (3) A conformal structure and a non-trivial holomorphic cubic differential. $(g \ge 1)$.

In order to complete the classification it is necessary to show how to construct from data of type (2) or (3) a Riemannian Einstein AH structure, and to analyze when two Riemannian Einstein AH structures are equivalent modulo $\text{Diff}^+(M)$. The second step is basically straightforward because of the uniformization theorem, and so the content of this section and the next is the analysis of the first step. For whatever *g* the given data of a holomorphic section and a conformal structure determine an elliptic PDE for the conformal factor expressing a putative Gauduchon metric in terms of a constant scalar curvature background metric.

For $g \ge 2$ the existence of a unique solution for the resulting PDE is a straightforward application of standard elliptic PDE techniques. This is explained in the present section. For $g \in \{0, 1\}$ the uniqueness fails because the conformal automorphism group is large, but the holomorphic vector field induces an S^1 symmetry which can be used to reduce the PDE to an ODE which is easily solved. This is described in Section 10.

9.1. On a smooth surface M, associate to a triple (h, F, B) comprising a Riemannian metric h, a smooth function $F \in C^{\infty}(M)$, and the real part B of a smooth k-differential, the differential operator defined for $\phi \in C^{\infty}(M)$ by

$$\mathcal{A}(h, F, B)(\phi) := \Delta_h \phi - \mathcal{R}_h + F e^{\phi} + 2^{1-k} e^{(1-k)\phi} |B|_h^2.$$
(9.1)

It is convenient to include the constant factor 2^{1-k} . More generally, there can be several *B*'s, for different *k*'s, *e.g.* for a holomorphic one-form $X^{(1,0)}$, and a holomorphic cubic differential $B^{(3,0)}$, it is convenient to write $\mathcal{A}(h, F, X, B)(\phi) =$ $\Delta_h \phi - \mathcal{R}_h + F e^{\phi} + 4e^{2\phi}|X|_h^2 + \frac{1}{4}e^{-2\phi}|B|_h^2$. The metric *h* will be called the **background** metric of the equation. Note that in the analysis of \mathcal{A} the holomorphicity or not of $B^{(k,0)}$ plays *no* role; it is important only for the properties of the objects constructed from the solutions. For $\mu, \lambda \in C^{\infty}(M)$ there holds the scaling rule

$$e^{\mu}\mathcal{A}(e^{\mu}h, e^{\lambda}F, e^{(1-k)\lambda/2}B)(\phi - \mu - \lambda) = \mathcal{A}(h, F, X, B)(\phi) - \Delta_h\lambda.$$
(9.2)

9.2. Let $([\nabla], [h])$ be an AH structure on a compact orientable surface M of genus g. Let $\nabla \in [\nabla]$ be its aligned representative and let $\tilde{h}_{ij} = e^{-\phi}h_{ij} \in [h]$ and $h \in [h]$ have Levi-Civita connections related by $\tilde{D} = D + 2\sigma_{(i}\delta_{j)}^{k} - h_{ij}\sigma^{k}$, in which $2\sigma_{i} := -d\phi_{i}$ and $\sigma^{i} := h^{ip}\sigma_{p}$. Recall the notational conventions established in Section 2.5. Let γ_{i} be the Faraday primitive of h and let $B_{ijk} := \mathcal{L}_{ij} {}^{p}h_{pk}$. Using

(2.1), (5.8),
$$d_{h}^{*}\gamma = e^{-\phi}d_{\tilde{h}}^{*}\gamma$$
, and $\Delta_{h}\phi = e^{-\phi}\Delta_{\tilde{h}}\phi$ there results

$$\mathcal{A}(\tilde{h}, \mathbf{R}_{h}, B)(\phi) + d_{\tilde{h}}^{*}\gamma = \Delta_{\tilde{h}}\phi - \mathcal{R}_{\tilde{h}} + e^{\phi}\mathbf{R}_{h} + \frac{1}{4}e^{-2\phi}|B|_{h}^{2} + d_{\tilde{h}}^{*}\gamma$$

$$= e^{\phi}\left(\Delta_{h}\phi + \mathbf{R}_{h} + \frac{1}{4}|\mathcal{L}|_{h}^{2} + d_{h}^{*}\gamma\right) - \mathcal{R}_{\tilde{h}} = 0.$$
(9.3)

Equation (9.3) can be used to construct AH structures. The metric \tilde{h} will be treated as the background metric and h (equivalently ϕ) as the unknown. If \tilde{h}_{ij} , γ_i , R_h , and B_{ijk} are given, then solving (9.3) for ϕ yields an AH structure with cubic torsion $\mathcal{L}_{ij}{}^{k} = h^{kp}B_{ijp}$ and scalar curvature $R = |\det h|^{1/2}R_h$, for which $h = e^{\phi}\tilde{h}$ is a representative metric with Faraday primitive γ_i . It is convenient to seek Gauduchon h, in which case it must be that $d^*_{\tilde{h}}\gamma = e^{\phi}d^*_{h}\gamma = 0$, and the equation to be solved reduces to $\mathcal{A}(\tilde{h}, R_h, B)(\phi) = 0$. If the AH structure obtained by solving (9.3) is to be Einstein then it must be that $B^{(3,0)}$ is holomorphic, and there must be a constant κ and a holomorphic vector field $X^{(1,0)}$ such that $X^p h_{ip} = \gamma_i$ is the Faraday primitive of h and $R_h = 4|X|^2_h + \kappa = 4e^{\phi}|X|^2_{\tilde{k}} + \kappa$. In (9.3) this yields

$$\mathcal{A}(\tilde{h},\kappa,X,B)(\phi) = \Delta_{\tilde{h}}\phi - \mathcal{R}_{\tilde{h}} + \kappa e^{\phi} + 4e^{2\phi}|X|_{\tilde{h}}^2 + \frac{1}{4}e^{-2\phi}|B|_{\tilde{h}}^2 = 0, \quad (9.4)$$

as the equation to be solved. Of course, by Corollary 7.3, if there is to be a solution only one of X and B can be non-zero.

It follows from (9.2) that ϕ solves (9.4) if and only if for any $\mu \in C^{\infty}(M)$ and $r \in \mathbb{R}^+$ the function $\psi = \phi - \log r - \mu$ solves $\mathcal{A}(e^{\mu}h, r\kappa, rX, r^{-1}B)(\psi) = 0$. The resulting metrics $\bar{h}_{ij} = e^{\psi}e^{\mu}\tilde{h}_{ij} = r^{-1}e^{\phi}\tilde{h}_{ij}$ and $h_{ij} = e^{\phi}\tilde{h}_{ij}$ are positively homothetic, while the resulting tensors $\gamma_i = X^p h_{pi} = rX^p \tilde{h}_{pi}$ and $\mathcal{L}_{ij}{}^k = h^{kp}B_{ijp} = \bar{h}^{kp}r^{-1}B_{ijp}$ are the same, so that the AH structures resulting from these solutions are the same. Thus such rescaling is trivial from the point of view of constructing AH structures, and it is natural to restrict the allowable ϕ by imposing some normalization which eliminates this freedom.

The scaling in *h* given by μ is eliminated by fixing a convenient background metric \tilde{h} . The freedom in *r* is most naturally eliminated by imposing some condition on the resulting metric $h = e^{\phi}\tilde{h}$, *e.g.* fixing the minimum of \mathcal{R}_h . There are several possibilities. An obvious one is to require that R_h take a specific value; this amounts to fixing κ . Another is to demand that $\operatorname{vol}_h(M) = \int_M e^{\phi} d \operatorname{vol}_{\tilde{h}}$ have some prespecified value; as will be made precise below, this puts some conditions on κ necessary for the existence of solutions. The curvature normalization is probably more natural from the geometric point of view, while the volume normalization is probably more natural from the point of view of partial differential equations. A function $\phi \in C^{\infty}(M)$ is **volume normalized** with respect to \tilde{h} if $\int_M e^{\phi} d \operatorname{vol}_{\tilde{h}} = \operatorname{vol}_{\tilde{h}}(M)$, in which case $\operatorname{vol}_h(M) = \operatorname{vol}_{\tilde{h}}(M)$. From Lemma 7.6 it follows that for there to exist a volume normalized ϕ solving (9.4) it is necessary that $\kappa \leq 4\pi \chi(M)/\operatorname{vol}_{\tilde{h}}(M)$. Moreover, for equality to hold it is necessary that both X and B be identically zero, in which case (9.4) becomes simply $\Re_{e^{\phi}\tilde{h}} = e^{-\phi} \left(\Re_{\tilde{h}} - \Delta_{\tilde{h}}\phi \right) = 4\pi \chi(M) / \operatorname{vol}_{\tilde{h}}(M)$. This is simply the equation that $e^{\phi}\tilde{h}$ have constant curvature; in this case there is by the uniformization theorem a unique normalized ϕ solving the equation. Such a solution to (9.4) will be referred to as **uniformizing**.

In summary, if ϕ is a C^{∞} solution of (9.4) then $h_{ij} := e^{\phi} \tilde{h}_{ij}$ is the Gauduchon metric with associated Faraday primitive $\gamma_i = X^p h_{ip}$ of an Einstein AH structure having cubic torsion $\mathcal{L}_{ij}{}^k = h^{kp} B_{ijp}$, scalar curvature $R_h = 4|X|_h^2 + \kappa$, and vortex parameter $\nu = \kappa \operatorname{vol}_h(M)$. The natural choices for the background metric are the metric $\tilde{h} \in [h]$ having constant scalar curvature in $\{0, \pm 2\}$, and the metrics ${}^*h = |X|_{\tilde{h}}^{-2}\tilde{h}$ and ${}^*h = |B|_{\tilde{h}}^{2/3}\tilde{h}$ (which do not depend on the choice of $\tilde{h} \in [h]$), which are flat by Lemma 3.6. The metrics *h are defined only on an open subset of M, so if they are to used there have to be imposed boundary conditions on ϕ .

9.3. The existence of a unique solution to $\mathcal{A}(h, k, B)(\phi) = 0$ when $\chi(M) \leq 0$, *B* is somewhere non-zero, and $k \leq 4\pi \chi(M) / \operatorname{vol}_h(M)$ follows from Lemmas 9.1 and Lemma 9.4. These lemmas are, however, stated more generally, so as to yield also Corollary 9.5, which shows how to associate to a Riemann surface equipped with a non-trivial holomorphic k-differential a canonical smooth metric in the given conformal structure, by solving the Abelian vortex equations. For a cubic holomorphic differential, the existence and uniqueness statements of Lemmas 9.1 and 9.4 for a metric of constant scalar curvature -2 on a compact oriented surface *M* of genus greater than 1 and k = 2, are Loftin's [51, Theorem 4.0.2], and the proofs given here are very similar to Loftin's, which are themselves based on standard arguments as in [44] or, particularly, [77]. The generality of allowing nonconstant *k* is convenient for the example of Section 9.4.

Lemma 9.1. If M is a compact, orientable surface with $\chi(M) \leq 0$, h is a Riemannian metric, and, for an integer $p \geq 1$, B is the real part of a non-trivial smooth p-differential, then the equation $\mathcal{A}(h, k, B)(\phi) = 0$ has at most one solution in $C^{\infty}(M)$ for each $k \in C^{\infty}(M)$ satisfying $k \leq 0$.

Proof. Suppose $\phi, \psi \in C^{\infty}(M)$ solve $\mathcal{A}(h, k, B)(\phi) = 0$ for some $k \in C^{\infty}(M)$. Then

$$\Delta_h(\phi - \psi) = -k(e^{\phi} - e^{\psi}) - 2^{(1-p)/2}(e^{(1-p)\phi} - e^{(1-p)\psi})|B|_h^2.$$
(9.5)

If k is strictly negative the claim follows from the maximum principle applied to (9.5). If $k \le 0$, it follows from (9.5) that

$$\Delta_{h}(\phi - \psi)^{2} = 2(\phi - \psi)\Delta_{h}(\phi - \psi) + 2|d(\phi - \psi)|_{h}^{2}$$

= $-2k(\phi - \psi)(e^{\phi} - e^{\psi}) - 2^{2-p}|B|_{h}^{2}(\phi - \psi)(e^{(1-p)\phi} - e^{(1-p)\psi})$ (9.6)
 $\geq 2|d(\phi - \psi)|_{h}^{2} \geq 0,$

so that $(\phi - \psi)^2$ is subharmonic and therefore constant, since *M* is compact. Write $\psi = \phi + c$ for a constant *c*, and substitute this into (9.5) to obtain $e^{-p\phi}|B|_h^2(1 - \psi)^2$

 $e^{(1-p)c}$) = $2^{p-1}k(e^c - 1)$. If not both p = 1 and $k \equiv 0$, then, since the signs of $e^c - 1$ and $1 - e^{-2c}$ are the same if $c \neq 0$, and, by hypothesis, $k \leq 0$ and B is not identically zero, this can be only if c = 0. If both p = 1 and $k \equiv 0$, there is no solution, for if $\mathcal{A}(h, k, B)(\phi) = 0$, then integration yields $0 \geq 4\pi \chi(M) = \int_M \mathcal{R}_h d \operatorname{vol}_h = ||B||_h^2$, contradicting the assumption that B is not identically zero.

Lemma 9.2. Let *M* be a smooth torus. Let *h* be a flat Riemannian metric, and let *B* be the real part of a non-trivial holomorphic *p*-differential for $p \ge 1$. Then *B* is parallel and $|B|_h^2$ is constant. For each constant $\kappa < 0$ the unique solution ϕ to the equation $\mathcal{A}(h, \kappa, B)(\phi) = 0$ is the constant function $\frac{1}{p} \log (2^{1-p}|\kappa|^{-1}|B|_h^2)$.

Proof. By Corollary 3.7, *B* is *h*-parallel, and hence $|B|_h^2$ is a non-zero constant, so the given ϕ solves $\mathcal{A}(h, \kappa, B)(\phi) = 0$. This is the unique solution by Lemma 9.1.

Lemma 9.3 is needed in the proof of Lemma 9.4.

Lemma 9.3. Let $0 and <math>0 < a, b \in \mathbb{R}$. The unique positive root r_1 of $f(r) = r^p - ar^{p-1} - b$ satisfies $a \le r_1 \le a + b^{1/p}$.

Proof. For p = 1 the positive root of g is a + b, while for p = 2 it is $(a + \sqrt{a^2 + 4b})/2 \le a + b^{1/2}$, so suppose p > 2. Since f is negative at r = 0, has a negative minimum at a(1 - 1/p), is monotone decreasing on (0, a(1 - 1/p)) and monotone increasing on $(a(1 - 1/p), \infty)$, and satisfies $\lim_{r \to \infty} f(r) = \infty$, it has a unique positive real root r_1 which is greater than a(1 - 1/p), and, since $a + b^{1/p} > a > a(1 - 1/p)$, it suffices to observe f(a) = -b < 0 and

$$f(a+b^{1/p}) = (a+b^{1/p})^{p-1}b^{1/p} - b = \sum_{s=0}^{p-2} \binom{p-1}{s} b^{(s+1)/p}a^{p-1-s} > 0. \quad \Box$$

Lemma 9.4. Let M be a compact, orientable surface with $\chi(M) < 0$. Let h be a Riemannian metric, let B be the real part of a smooth p-differential not everywhere zero, and let $k \in C^{\infty}(M)$ be everywhere negative. Let $q = \min_{M} \Re_{h}$, $Q = \max_{M} \Re_{h}$, $P = \max_{M} |B|_{h}^{2}$, $\kappa = \min_{M} k$ and $K = \max_{M} k$. If the curvature of h is negative, then there is a unique solution ϕ to the equation $\mathcal{A}(h, \kappa, B)(\phi) = 0$, and it satisfies

$$\kappa^{-1}\left(\max_{M} \mathcal{R}_{h}\right) \leq e^{\phi} \leq K^{-1}\left(\min_{M} \mathcal{R}_{h}\right) + 2^{(1-p)/p} |K|^{-1/p} \left(\max_{M} |B|_{h}^{2/p}\right).$$
(9.7)

Proof. The uniqueness follows from Lemma 9.1. Following the proof in [51, Theorem 4.0.2] the existence is demonstrated by applying of [64, Theorem V.1.1], which shows that if a semi-linear elliptic equation $\Delta_h u + F(x, u)$ with $F \in C^{\infty}(M \times \mathbb{R})$ on a compact manifold M admits a C^2 supersolution u^+ and a C^2 subsolution u^- such that $u^- \leq u^+$ then it admits a C^{∞} solution u such that $u^- \leq u \leq u^+$ (this is proved by a modification of a standard iteration argument closely related versions

of which can be found in Section 2 of the appendix to the fourth chapter of [26] and [44, Section 9]). Suppose Q < 0. Since both Q and κ are negative, Q/κ is a positive constant, and $\mathcal{A}(h, k, B)(\log(Q/\kappa)) \ge -Q + kQ/\kappa \ge 0$, so $\log(Q/\kappa)$ is a subsolution. On the other hand, the polynomial $f(r) = Kr^3 - qr^2 + \frac{1}{4}P$ has the unique real root r_1 , which is easily seen to be positive and no smaller than q/K, which in turn is no smaller than Q/κ . As $\mathcal{A}(h, \kappa, B)(\log r_1) \le r_1^{-1}f(r_1) \le 0$, $\log r_1$ is a supersolution. There follows $\log(Q/\kappa) \le \phi \le \log r_1$. The bound (9.7) follows from Lemma 9.3 with a = q/K and $b = 2^{1-p}|K|^{-1}P$.

There follows the existence of p-canonical Abelian vortices, as claimed in Section 8.

Corollary 9.5. Let (M, [h], J) be a compact Riemann surface of genus g > 1, let $0 , let <math>B^{(p,0)}$ be a non-trivial holomorphic p-differential, and let κ be a negative constant. There is a unique representative $h \in [h]$ such that $s = 2^{1-p/2}p^{1/2}B^{(p,0)}$ solves the Abelian vortex equations (8.1) for $\tau = -p\kappa$ and the Hermitian structure induced on \mathcal{K}^p by h.

Proof. Let $\tilde{h} \in [h]$ and write $h = e^{\phi} \tilde{h}$. Arguing as in Section 8 the putative h must satisfy

$$0 = \frac{2}{p} \left(i\Lambda(\Omega) + \frac{1}{2} |s|_{h}^{2} - \frac{1}{2}\tau \right) = -\mathcal{R}_{h} + 2^{2-p} |B^{(p,0)}|_{h}^{2} + \kappa$$
$$= e^{-\phi} \left(\Delta_{\tilde{h}} \phi - \mathcal{R}_{\tilde{h}} + \kappa e^{\phi} + 2^{1-p} e^{(1-p)\phi} |B|_{\tilde{h}}^{2} \right) = e^{-\phi} \mathcal{A}(\tilde{h}, \kappa, B)(\phi).$$

For \tilde{h} such that $\mathcal{R}_{\tilde{h}} = -2$, the existence of a unique solution follows from Lemmas 9.1 and 9.4.

For a compact orientable surface, Corollary 9.5 associates to each $(J, B) \in Q^k(M)$ a distinguished metric representing the conformal structure. If (J, B) is constructed from a flat metric *h with conical singularities as in Section 3.6, then this associates to such a metric a smooth conformal metric in a canonical manner. This observation might be of use in the study of such metrics.

Lemma 9.6. Let $1 \le p \in \mathbb{Z}$. Let M be a compact, orientable surface with $\chi(M) < 0$, let \tilde{h} be a Riemannian metric with negative scalar curvature satisfying $\min_M \Re_{\tilde{h}} = -2$, let $B^{(p,0)}$ be holomorphic, and let κ be a negative constant. If ϕ solves $\mathcal{A}(\tilde{h}, \kappa, B)(\phi) = 0$, and $h_{ij} = e^{\phi} \tilde{h}_{ij}$ then

$$-2\kappa^{-1} + 2^{(1-p)/p} |\kappa|^{-1/p} (\max_{M} |B|_{\tilde{h}}^{2/p}) \ge e^{\phi} \ge 2^{(1-p)/p} |\kappa|^{-1/p} |B|_{\tilde{h}}^{2/p}, \qquad (9.8)$$

$$(-2\kappa^{-1} + 2^{(1-p)/p} |\kappa|^{-1/p} (\max_{M} |B|_{\tilde{h}}^{2/p})) \operatorname{vol}_{\tilde{h}}(M) \geq \operatorname{vol}_{h}(M) \geq 2^{(1-p)/p} |\kappa|^{-1/p} \int_{M} |B|_{\tilde{h}}^{2/p} d \operatorname{vol}_{\tilde{h}} = 2^{(1-p)/p} |\kappa|^{-1/p} \operatorname{vol}_{*h}(M).$$

$$(9.9)$$

Here ${}^{*}h \in [h]$ *is the singular flat metric* ${}^{*}h = |B|_{\tilde{h}}^{2/p} \tilde{h}$.

Proof. In the p = 3 case, by (7.15) applied to the resulting exact Einstein AH structure with Gauduchon metric $h = e^{\phi}\tilde{h}$, there holds $-4\kappa = -4R_h \ge |B|_h^2 = e^{-3\phi}|B|_{\tilde{h}}^2$, which is the second inequality of (9.8). Alternatively, the following proof, applicable for all $p \ge 1$, can be viewed as giving another proof of (7.15). By (3.12) of Lemma 3.6, $\psi = \log\left(2^{(1-p)/p}|\kappa|^{-1/p}|B|_{\tilde{h}}^{2/p}\right)$ solves $\mathcal{A}(\tilde{h}, \kappa, B)(\psi) = 0$ on the complement of the zero set of B. (Essentially this observation is key in both [61] and the proof of [54, Proposition 1]). Since ψ goes to $-\infty$ on the zero set of B, there is $\epsilon > 0$ so that on the boundary ∂M^{ϵ} of the complement M^{ϵ} of an ϵ neighborhood of the zero set of B there holds $\psi \le \min_M \phi$. Since the zeroth order part of $\mathcal{A}(\tilde{h}, \kappa, B)(u)$ is non-increasing in u, it follows that

$$\Delta_{\tilde{h}}(\psi - \phi) = -\kappa (e^{\psi} - e^{\phi}) - 2^{1-p} (e^{(1-p)\psi} - e^{(1-p)\phi}) |B|_{\tilde{h}}^2$$

is non-negative on the domain $U = \{x \in M^{\epsilon} : \psi(x) > \phi(x)\}$. Since by the choice of ϵ the closure of U is contained properly in M^{ϵ} , the maximum principle implies U is empty, showing that $\psi \le \phi$ on M^{ϵ} . Letting $\epsilon \to 0$ yields the second inequality of (9.8). The first inequality of (9.8) follows from (9.7). Integrating (9.8) yields the volume bound (9.9).

9.4. Example: naive Einstein AH structures which are not Einstein

On a compact surface M of genus g > 1, let \tilde{h} be a Riemannian metric with scalar curvature -2, and let $B^{(3,0)}$ be a cubic holomorphic differential. Let k be a smooth function on M which satisfies k < 0. By Lemmas 9.1 and 9.4 the equation $\mathcal{A}(\tilde{h}, k, B)(\phi) = 0$ admits a unique smooth solution ϕ . Let $h_{ij} = e^{\phi}\tilde{h}_{ij}$, $\mathcal{L}_{ij}{}^k = h^{kp}B_{ijp}$, and $\nabla = D - \frac{1}{2}\mathcal{L}_{ij}{}^k$, in which D is the Levi-Civita connection of h_{ij} . Then ∇ is the aligned representative of the AH structure which it generates with [h], which is exact, and h is a distinguished representative of [h]. There holds $D_p\mathcal{L}_{ij}{}^p = h^{pq}D_pB_{ijq} = 0$ because B is the real part of a holomorphic differential, and so by (5.7), ($[\nabla], [h]$) is naive Einstein. On the other hand, by (5.8), there holds $\mathbb{R}_h = \mathcal{R}_h - \frac{1}{4}|\mathcal{L}|_h^2 = \mathcal{R}_h - \frac{1}{4}|B|_h^2$, while by (2.1) and the construction of ϕ there holds $\mathcal{R}_h = e^{-\phi}(-2 - \Delta_{\tilde{h}}\phi) = k + \frac{1}{4}|B|_h^2$, so that $\mathbb{R}_h = k$. Thus if k is not constant, then \mathbb{R}_h is not constant, so by (7.1), ($[\nabla], [h]$) is not Einstein.

9.5. Let *M* be a compact, oriented surface. Recall the spaces defined in Section 3.5. Let E(M) be the space of Riemannian signature Einstein AH structures on *M*. The underlying conformal structure of each element of E(M) determines a complex structure inducing the given orientation, and so there is an evidently sujective map $E(M) \rightarrow J(M)$. The group Diff⁺(*M*) acts on E(M) by pull-back. Define the **deformation space** $\mathcal{E}(M) = E(M) / \text{Diff}_0(M)$ of Einstein AH structures on *M* and the **moduli space** $\mathcal{M}(M) = E(M) / \text{Diff}^+(M)$ of Einstein AH structures on M. The oriented mapping class group Map⁺(*M*) acts on $\mathcal{E}(M)$ with quotient $\mathcal{M}(M)$. The canonical map $E(M) \rightarrow J(M)$ descends to a map $\mathcal{E}(M) \rightarrow \mathcal{T}(M)$, and similarly at the level of moduli spaces.

9.6. Suppose the compact, oriented surface *M* has genus g > 1. For each $a \in \mathbb{R}^+$ define a map $\Psi^a : E(M) \to Q^3(M)$ by $\Psi^a([\nabla], [h]) = (J, B)$ in which *J* is the complex structure determined by [h] and the given orientation, and *B* is defined by $B_{ijk} = \mathcal{L}_{ij} {}^p h_{pk}$ for the unique distinguished metric $h \in [h]$ such that $R_h = -2a^{-1}$. Alternatively, $||B||_h^2$ does not depend on the choice of $h \in [h]$ and the choice of h is determined by requiring that $2 \operatorname{vol}_h(M) = a(4^{-1}||B||_h^2 - 4\pi \chi(M))$.

Evidently Ψ^a is Diff⁺(*M*) equivariant, so descends to a Map⁺(*M*)-equivariant map Ψ^a : $\mathcal{E}(M) \to \mathcal{Q}^3(M)$, which covers the identity on $\mathcal{T}(M)$. It is convenient to write $\Psi^a([\nabla], [h]) = [\Psi^a([\nabla], [h])]$ for the image of an equivalence class $[[\nabla], [h]] \in \mathcal{E}(M)$ (the notation indicating the equivalence class of $([\nabla], [h])$ was omitted on the left-hand side).

Running through the definitions shows that, for $r \in \mathbb{R}^+$, $\Psi^{ra} = (J, rB)$ if $\Psi^a = (J, B)$, which in part accounts for using a^{-1} in the definition of Ψ^a . Since $(\Psi^r)^{-1}(J, B) = (\Psi^1)^{-1}((J, r^{-1}B))$ it in general suffices to work with the fixed map $\Psi := \Psi^1$ and its inverse. The duality on E(M) given by conjugacy of AH structures corresponds under Ψ to replacing the holomorphic cubic differential B by -B, in the sense that $\Psi([\bar{\nabla}], [h]) = (J, -B)$ if $\Psi([\nabla], [h]) = (J, B)$.

Theorem 9.7. For a compact, oriented surface M of genus g > 1, the Diff⁺(M)equivariant map $\Psi : E(M) \to Q^3(M)$ is a bijection, and so the Map⁺(M)equivariant map $\Psi : \mathcal{E}(M) \to Q^3(M)$ is a bijection as well.

Proof. For $(J, B) \in Q^3(M)$ let [h] be the conformal structure determined by J. For the unique representative $\tilde{h} \in [h]$ having constant scalar curvature -2, there is by Lemma 9.4 a unique solution ϕ to $\mathcal{A}(\tilde{h}, -2, B)(\phi) = 0$. The metric $h = e^{\phi}\tilde{h} \in [h]$ is the distinguished representative of an exact Einstein AH structure $([\nabla], [h])$ having aligned representative $\nabla = D - \frac{1}{2}h^{kp}B_{ijp}$ and $R_h = -2$, so that $\Psi([\nabla], [h]) = (J, B)$. This shows that Ψ is onto.

Now suppose that $\Psi([\nabla], [h]) = (J, B) = \Psi([\tilde{\nabla}], [g])$. Since [g] and [h]induce the same complex structure J, they are equal. Let $\nabla \in [\nabla]$ and $\tilde{\nabla} \in [\tilde{\nabla}]$ be the aligned representatives, and let $h \in [h]$ and $\bar{h} \in [h]$ be the distinguished representatives of $([\nabla], [h])$ and $([\tilde{\nabla}], [h])$ such that the corresponding curvatures R_h and $\tilde{R}_{\bar{h}}$ equal -2. Let $\tilde{h} \in [h]$ be the unique representative with constant scalar curvature $\mathcal{R}_{\bar{h}} = -2$ and write $h_{ij} = e^{\phi}\tilde{h}_{ij}$ and $\bar{h}_{ij} = e^{\bar{\phi}}\tilde{h}_{ij}$. Then ϕ and $\bar{\phi}$ solve $\mathcal{A}(\tilde{h}, -2, B)(\psi) = 0$, and so, by Lemma 9.1, $\bar{\phi} = \phi$. This shows that $\bar{h}_{ij} = h_{ij}$. It follows that $\tilde{\nabla} = \nabla$, and so $[\tilde{\nabla}] = [\nabla]$, and the injectivity of Ψ has been shown. \Box

The content of the surjectivity statement in Theorem 9.7 is not essentially different from that of C. P. Wang's [75, Theorem 3.4], in which it is shown how to construct an affine hypersphere from a conformal metric and a cubic holomorphic differential. As is explained in Section 11.8, $\mathcal{E}(M)$ can be identified with the deformation space $\mathcal{P}(M)$ of convex flat real projective structures on M. The key point is that an exact Einstein AH structure is determined by its underlying flat projective structure. The resulting identification of $\Omega^3(M)$ with $\mathcal{P}(M)$ is due independently to F. Labourie and J. Loftin. **9.7.** The results of this section can be viewed as preliminary steps in the direction of understanding the action of $GL^+(2, \mathbb{R})$ on $\mathcal{E}(M)$ described in Section 3.6. In the remainder of the section M is a compact, oriented surface of genus at least two and $B \in \Gamma(S_0^3(T^*M))$ is the real part of a cubic holomorphic differential supposed not identically zero. Sense can be made of all the results of this section for $1 \le p \in \mathbb{Z}$ in place of 3 if Gauduchon metrics are interpreted as the representatives given by Corollary 9.5 and the numbers 3 and 2 are replaced by p and (p - 1), where appropriate.

Specializing the action of $GL^+(2, \mathbb{R})$ on $Q^3(M)$ to $\mathbb{C}^{\times} = C^+O(2)$, write $z \cdot (J, B^{(3,0)}) = (z \cdot J, z \cdot B^{(3,0)})$. Then, for $z = re^{i\theta} = e^t e^{i\theta} \in \mathbb{C}^{\times}, z \cdot J = J$ and $2\text{Re}(z \cdot B^{(3,0)}) = 2\text{Re}(z^3B^{(3,0)}) = e^{3t}(\cos(3\theta)B + \sin(3\theta)\mathbb{J}(B))$. Let $([\nabla], [h]) = \Psi^{-1}(J, B)$ and $([\tilde{\nabla}], [h]) = \Psi^{-1}(J, B(t, \theta))$ where $B(t, \theta) = e^{3t}(\cos(\theta)B + \sin(\theta)\mathbb{J}(B))$ for some fixed t > 0 and θ . Retaining the factor of 3 in the scale factor, while dropping it in the rotational part simplifies formulas appearing later. Both $([\nabla], [h])$ and $([\tilde{\nabla}], [h])$ are exact Einstein AH structures. Let $h \in [h]$ and $\tilde{h} \in [h]$ be the respective distinguished metrics such that $R_h = -2 = R_{\tilde{h}}$, and write $\tilde{h}_{ij} = e^{\phi}h_{ij}$. By construction the cubic torsions are $\mathcal{L}_{ij}{}^k = h^{kp}B_{ijp}$ and

$$\begin{aligned} \tilde{\mathcal{L}}_{ij}^{\ k} &= \tilde{h}^{kp} (e^t e^{\mathrm{i}\theta} \cdot B)_{ijp} = e^{3t - \phi} h^{kp} (\cos(\theta) B_{ijp} + \sin(\theta) \mathbb{J}(B)_{ij}^{\ p}) \\ &= e^{3t - \phi} (\cos(\theta) \mathcal{L}_{ij}^{\ k} + \sin(\theta) J_i^{\ p} \mathcal{L}_{pj}^{\ k}), \end{aligned}$$

(recall (4.6) of Lemma 4.3). By construction ϕ , and so also the Gauduchon representative of $\Psi^{-1}([\tilde{\nabla}], [h])$, does not depend on θ , so in analyzing the dependence on t it may as well be assumed that $\theta = 0$. The dependence on t was partly analyzed in [54, Proposition 1], and that result, as well as similar ones for quadratic differentials in the context of Teichüller space proved in [77, Section 5], provided motivation for Lemma 9.8 and Theorem 9.9.

Since, by Theorem 7.1, $\mathcal{R}_h = -2 + 4^{-1} |B|_h^2$, by construction ϕ solves

$$\mathcal{O}(\phi) := \mathcal{A}(h, -2, e^{3t}B)(\phi) = \Delta_h \phi + 2 - 2e^{\phi} + 4^{-1}(e^{2(3t-\phi)} - 1)|B|_h^2 = 0.$$
(9.10)

Lemma 9.8. Let $(J, B) \in Q^3(M)$ for a compact smooth orientable surface M of genus g > 1. Let h and $\tilde{h} = e^{\phi}h$ be the Gauduchon metrics of $\Psi^{-1}(J, B)$ and $\Psi^{-1}(J, B(t, \theta))$ such that $R_h = -2 = R_{\tilde{h}}$. Then ϕ is the unique solution of (9.10), and satisfies

$$\max\{0, 2t + (2/3)\log|B|_h - \log 2\} \le \phi \le 2t, \qquad \text{if } t \ge 0. \tag{9.11}$$

(In particular ϕ is identically 0 if t = 0).

Proof. By (7.15), $|B|_{h}^{2} \leq 8$, and for t > 0 there results

$$\mathcal{O}(2t) = 2 - 2e^{2t} + 4^{-1}(e^{2t} - 1)|B|_h^2 = 4^{-1}(e^{2t} - 1)(|B|_h^2 - 8) \le 0, \quad (9.12)$$

showing that the constant function 2t is a supersolution of \mathcal{O} . Since $\mathcal{O}(0) \ge 0$ also, there exists a solution ψ of $\mathcal{O}(\psi) = 0$ such that $0 \le \psi \le 2t$. If φ is a second

solution then

$$\Delta_h(\varphi - \psi)^2 = 2|d(\varphi - \psi)|^2 + 4(\varphi - \psi)(e^{\varphi} - e^{\psi}) - 2^{-1}e^{6t}(\varphi - \psi)(e^{-2\varphi} - e^{-2\psi}) \ge 0.$$
(9.13)

By the maximum principle there is a constant c such that $\varphi - \psi = c$, and in (9.13) this yields $8c(e^c - 1)e^{3\varphi} = ce^{6t}(e^{-2c} - 1)|B|_h^2$. Since B is not identically zero this can be only if c = 0. Thus ϕ is the unique solution of $\mathcal{O}(\phi) = 0$, and $0 \le \phi \le 2t$. By (7.10), the function $\psi = \log(2^{-1}e^{2t}|B|_h^{2/3})$ satisfies $\mathcal{O}(\psi) = 0$ on the complement M^* of the zero set of B. Since ψ tends to $-\infty$ at the zeroes of B and ϕ is bounded on M, the closure of the set $U = \{x \in M^* : \psi(x) > \phi(x)\}$ is contained properly in M^* . On U there holds

$$\Delta_h(\psi - \phi) = 2(e^{\psi} - e^{\phi}) - 4^{-1}e^{6t}(e^{-2\psi} - e^{-\phi})|B|_h^2 \ge 0,$$

which by the maximum principle contradicts that U be non-empty. This proves (9.11).

Theorem 9.9. Let M be a smooth compact orientable surface of genus g > 1. Fix $(J, B) \in Q^3(M)$ such that B is not identically zero. For $t \ge 0$, let th be the Gauduchon metric of $\Psi^{-1}(J, e^{3t}B)$ such that $\mathbb{R}_{th} = -2$. Write $h = {}^{1}h$ and ${}^{th} = e^{\phi_t}h$. Then ϕ_t is pointwise non-decreasing and Lipschitz as a function of $t \in (0, \infty)$, with Lipschitz constant 2.

Proof. Applying Lemma 9.8 with $t = t_2 - t_1$ and $\phi = \phi_{t_2} - \phi_{t_1}$, so that ${}^{t_2}h = e^{\phi t_1}h$, yields

$$\max\{0, 2(t_2 - t_1) + (2/3) \log |e^{3t_1}B|_{e^{\phi_{t_1}}h} - \log 2\} \le \phi_{t_2} - \phi_{t_1} \le 2(t_2 - t_1).$$
(9.14)

which, after simplifying and rearranging terms is

$$\max\{\phi_{t_1}, 2t_2 + \frac{2}{3}\log|B|_h - \log 2\} \le \phi_{t_2} \le \phi_{t_1} + 2(t_2 - t_1).$$
(9.15)

The first inequality of (9.15) shows $\phi_{t_2} \ge \phi_{t_1}$, so ϕ_t is non-decreasing for $t \in (0, \infty)$. From the second inequality of (9.15) it follows that $0 \le \phi_{t_2} - \phi_{t_1} \le 2|t_2 - t_1|$ for $t_1, t_2 \in (0, \infty)$.

Corollary 9.10. Let M be a smooth compact orientable surface of genus g > 1. Fix $(J, B) \in Q^3(M)$ such that B is not identically zero. For $t \ge 0$ let th be the Gauduchon metric of $\Psi^{-1}(J, e^{3t}B)$ such that $R_{th} = -2$ and let ^t $g = (e^{-2t})^{th}$ (which is also a Gauduchon metric for $\Psi^{-1}(J, e^{3t}B)$). Then the limits $\lim_{t\to\infty} \operatorname{vol}_{tg}(M)$ and $\lim_{t\to\infty} ||B||_{t_p}^2$ exist and satisfy

$$0 < \frac{1}{2} \operatorname{vol}_{h}(M) = \frac{1}{2} \int_{M} |B|_{h}^{2/3} d \operatorname{vol}_{h} \le \lim_{t \to \infty} \operatorname{vol}_{g}(M)$$

$$= \frac{1}{8} \lim_{t \to \infty} ||B||_{t_{g}}^{2} \le \operatorname{vol}_{h}(M).$$
(9.16)

Here $^*h = |B|_h^{2/3}h$. The curvature \Re_g satisfies $0 \ge \Re_{t_g} \ge 4^{-1}e^{2t}(|B|_h^2 - 8)$.

Proof. Write $h = {}^{1}h$ and ${}^{t}h = e^{\phi_{t}}h$. From the second inequality of (9.14) of the proof of Theorem 9.9 it is immediate that $\operatorname{vol}_{t_{g}}(M)$ is non-increasing for t > 0, and by (9.11) of Lemma 9.8, $\operatorname{vol}_{h}(M) \ge e^{-2t} \operatorname{vol}_{t_{h}}(M) = \operatorname{vol}_{t_{g}}(M) \ge \frac{1}{2} \int_{M} |B|_{h}^{2/3} d \operatorname{vol}_{h}$. This shows the existence of $\lim_{t\to\infty} \operatorname{vol}_{t_{g}}(M)$ and the bounds of (9.16). From (7.12) it follows that $\operatorname{vol}_{t_{g}}(M) - 8^{-1} ||B||_{r_{g}}^{2} = 4\pi (g-1)e^{-2t}$, from which the the equality of the limits in (9.16) is apparent. The final estimate on the curvature follows from $e^{-2t} \mathcal{R}_{t_{g}} = \mathcal{R}_{t_{h}} = -2 + 4^{-1}e^{6t}|B|_{r_{h}}^{2} = -2 + 4^{-1}e^{6t}e^{-3\phi_{t}}|B|_{h}^{2}$ coupled with (9.11).

Lemma 9.11 does not appear to be particularly useful, but it illustrates how the bounds of Lemma 9.6 can be used, and confirms a naive expectation.

Lemma 9.11. Let M be a compact orientable smooth surface. Fix a complex structure J and equip the space of holomorphic cubic differentials with the sup norm and $C^{\infty}(M)$ with the sup norm. Then $\Psi^{-1}((J, \cdot))$ is continuous with respect to these sup norms.

Proof. Let \tilde{h} be the metric of curvature -2 representing the conformal structure determined by J. Let $B^{(3,0)}$ and $C^{(3,0)}$ be holomorphic cubic differentials and suppose $|B - C|_{\tilde{h}} \leq \epsilon$ on M. Let $\phi = \Psi^{-1}((J, B))$ and $\psi = \Psi^{-1}((J, C))$. By construction $4\Delta_{\tilde{h}}(\psi - \phi) = 8(e^{\psi} - e^{\phi}) + e^{-2\phi}|B|_{\tilde{h}}^2 - e^{-2\psi}|C|_{\tilde{h}}^2$. Suppose $(\psi - \phi)^2$ assumes a positive maximum at $p \in M$. Then $|\psi - \phi|$ also assumes a positive maximum at p. Without loss of generality it may be supposed that $\psi(p) > \phi(p)$, so that $\max_M |\psi - \phi| = \psi(p) - \phi(p)$. At p there holds

$$0 \ge 2(\psi - \phi)^{-1} \Delta_{\tilde{h}}(\psi - \phi)^2 \ge 8(e^{\psi} - e^{\phi}) + (e^{-2\phi}|B|_{\tilde{h}}^2 - e^{-2\psi}|C|_{\tilde{h}}^2)$$

= $8(e^{\psi} - e^{\phi}) + (e^{-2\phi} - e^{-2\psi})|B|_{\tilde{h}}^2 + e^{-2\psi}(|B|_{\tilde{h}}^2 - |C|_{\tilde{h}}^2).$ (9.17)

The inequality (9.17) forces that the value at p of $|C|_{\tilde{h}}^2$ is greater than the value at p of $|B|_{\tilde{h}}^2$. Using $|B|_{\tilde{h}}^2 \leq 8$, it follows that at p there holds

$$e^{-2\psi}\epsilon(\epsilon+16) \ge e^{-2\psi}(\epsilon+16)|C-B|_{\tilde{h}} \ge e^{-2\psi}(|C|_{\tilde{h}}-|B|_{\tilde{h}})(2|B|_{\tilde{h}}+\epsilon) \ge e^{-2\psi}(|C|_{\tilde{h}}^2-|B|_{\tilde{h}}^2) \ge 8(e^{\psi}-e^{\phi})+(e^{-2\phi}-e^{-2\psi})|B|_{\tilde{h}}^2.$$
(9.18)

By the second inequality of (9.8) of Lemma 9.6, $e^{-2\psi}|C|_{\tilde{h}}^{4/3} \leq 4$. In (9.18) this yields that at p,

$$4\epsilon(\epsilon + 16) \ge 8|C|_{\tilde{h}}^{4/3}(e^{\psi} - e^{\phi}) \ge 8|C|_{\tilde{h}}^{4/3}(\psi(p) - \phi(p))$$

= $8|C|_{\tilde{h}}^{4/3} \max_{M} |\psi - \phi|.$ (9.19)

Since, by (9.17), $|C|_{\tilde{h}}^2$ is positive at p, this shows the claimed continuity.

10. Einstein-Weyl structures on the sphere and torus

In this section the deformation spaces of Einstein-Weyl structures on the two sphere \mathbb{S}^2 and torus \mathbb{T}^2 and some geometric properties of their members are described. In [13] and [15] Calderbank found explicit descriptions of Einstein-Weyl structures on \mathbb{S}^2 and \mathbb{T}^2 . While he did not explicitly address the description of the deformation spaces, all the necessary information is at least implicit in what he writes. He finds a local normal form for solutions of the two-dimensional Einstein-Weyl equations, and shows that on \mathbb{S}^2 or \mathbb{T}^2 the solutions are defined globally. Essentially he writes the underlying conformal structure in isothermal coordinates and uses the Killing field provided by the Gauduchon gauge to reduce the Einstein-Weyl equations to an ODE; the reduction and solution of the resulting ODE given in [13], though the solutions that are found are given a bit more explicitly, in terms of elementary trigonometric functions (rather than elliptic functions). The scalar curvatures and Faraday curvatures are computed explicitly, and their values are related to the parameterization of the deformation space.

10.1. An [h]-conformal Killing field X is **inessential** if there is some $h \in [h]$ for which X is Killing, *i.e.* $\mathcal{L}_X h = 0$, and is **essential** otherwise. By Theorem 7.1 the Gauduchon metric dual of the Faraday primitive of the Gauduchon class of an Einstein AH structure on a compact orientable surface is inessential.

By Lemma 3.6, every conformal Killing vector field on \mathbb{T}^2 is parallel (and so Killing) for a flat representative of the conformal structure, and so, if non-trivial, is inessential and nowhere vanishing. A flat metric on \mathbb{T}^2 may be represented as that induced by the Euclidean metric on the quotient of \mathbb{R}^2 by a rank two lattice Γ . A non-trivial conformal Killing field is parallel in this flat metric and so can be written as a linear combination of the constant vector fields generating Γ . Such a vector field is **rational** if it is a real multiple of a linear combination of the generators of the lattice with integer coefficients, and **irrational** otherwise. A conformal Killing field on \mathbb{T}^2 is rational if and only if its orbits are simple closed curves, while when irrational it has a single, dense orbit.

Lemma 10.1. A vector field X Killing for a non-flat Riemannian metric h on \mathbb{T}^2 is rational.

Proof. Since \mathbb{T}^2 is compact, the group *G* of isometries of *h* is a compact Lie group containing the one-parameter subgroup generated by *X*. Since the infinitesimal generator of any one-parameter subgroup of *G* is *h*-Killing, it is parallel in a flat metric conformal to *h*, so has no fixed points. If dim $G \ge 2$ then in a neighborhood of each point of \mathbb{T}^2 its action generates linearly independent *h*-Killing fields, and it follows that *h* is flat. Since *h* is not flat, *G* is a union of circles. Since the one-parameter subgroup generated by *X* is connected, it must be a circle. Its non-trivial orbits must be simple closed curves, so *X* is rational.

Since the interest here is in the deformation space of Einstein-Weyl structures and any two Kähler structures on \mathbb{S}^2 are equivalent, in considering \mathbb{S}^2 it will suffice to regard it as the Riemann sphere $\mathbb{P}^1(\mathbb{C})$ with its standard Kähler structure. A holomorphic vector field on \mathbb{S}^2 has either one multiplicity two zero, or two multiplicity one zeroes.

Lemma 10.2. An inessential conformal Killing field on \mathbb{S}^2 has two zeroes.

Proof. By [19, Lemma 0.1], a non-trivial inessential conformal Killing vector field X on \mathbb{S}^2 generates an isometric S^1 action on \mathbb{S}^2 fixing some zero p of X. Since the orbit of a point q distinct from p but close to p is a loop comprising points equidistant from p, the index of X at p is 1, and so by the Hopf index theorem X must have a second zero.

10.2. On \mathbb{S}^2 not every conformal Killing field arises as the Gauduchon dual of the Faraday primitive of an Einstein-Weyl structure because not every conformal Killing field is inessential. While on a torus every conformal Killing field is inessential, and every such vector field arises from a closed but non-exact Einstein-Weyl structure, if such a vector field arises from a non-closed Einstein-Weyl structure then by Lemma 10.1 it must be rational. One consequence of what follows is to show that these necessary conditions on *X* are sufficient for it to arise in this way from an Einstein-Weyl structure. The remainder of the section is dedicated to analyzing when the solutions obtained in this way are equivalent modulo Diff₀(*M*), and to describing explicitly the geometry of the resulting solutions.

10.3. Suppose (M, J, [h]) is a Kähler structure on a compact surface of genus g equal to zero or one, $\tilde{h} \in [h]$ has constant scalar curvature 2(1-g), and $X \in \Gamma(TM)$ is an inessential conformal Killing field. Given some background metric $g \in [h]$ it is desired to find $\phi \in C^{\infty}(M)$ such that $h = e^{\phi}g$ is a Gauduchon metric of an Einstein-Weyl structure ([∇], [h]) with aligned representative $\nabla = D - 2\gamma_{(i}\delta_{j)}^{k} +$ $h_{ij}h^{kp}\gamma_p$, in which D is the Levi-Civita connection of h and $\gamma_i = X^p h_{ip}$. Since X^i is to be h-Killing, by Lemmas 10.1 and 10.2 it must generate a circle action on M. The initial idea is to use this circle action to reduce $\mathcal{A}(\tilde{h}, \kappa, X)(\phi) = 0$ to an ODE. In the case of the torus this works, but in the case of the sphere, it is not obvious how to make the reduction, and the substitute is to work with the flat metric ${}^{*}h = |X|_{h}^{-2}h$ on the complement M^* of the zero set of X (when g = 1, the metric *h depends on the choice of \tilde{h}). So there will be sought $\phi \in C^{\infty}(M^*)$ such that the metric $h = e^{\phi * h}$ extends C^{∞} smoothly to M and is otherwise as above. The difference between the torus and sphere cases is in the boundary conditions necessary for ϕ to extend smoothly to all of M. For the torus $M^* = M$, while for the sphere, M^* is the complement of two points. Note that $|X|_{*h}^2 = 1$ and $e^{\phi} = |X|_h^2$. By Theorem 7.1, there must be a constant κ such that $R_h - 4|X|_h^2 = R_h - 4|\gamma|_h^2 = \kappa$. Rewriting this last equation in terms of *h using (5.8) shows that the desired Einstein-Weyl

structure can be found if ϕ solves

$$\Delta_{h}\phi + \kappa e^{\phi} + 4e^{2\phi} = 0, \tag{10.1}$$

where in the spherical case (10.1) is supplemented by the condition that ϕ extends smoothly to M.

10.4. The Killing property of X implies $e^{\phi} d\phi(X) = \mathcal{L}_X(|X|_h^2) = (\mathcal{L}_X h)(X, X) = 0$, so that $d\phi(X) = 0$. Since $\mathcal{L}_X J = 0$ there holds [X, JX] = 0. Let r and s be parameters for the flows of X and JX, respectively; then $X = \partial_r$, $JX = \partial_s$. Since X and JX are complete on M, their flows on M^* exist for all time, so r and s are global coordinates on the universal cover of M^* , which is the Euclidean plane. Note that each of r and s is determined only up to a translation. Since $d\phi(X) = 0$, ϕ is a function of s alone, and since in these coordinates ${}^*h = dr^2 + ds^2$, (10.1) becomes the ODE

$$\ddot{\phi} + \kappa e^{\phi} + 4e^{2\phi} = 0, \tag{10.2}$$

in which a dot indicates differentiation with respect to s. If ϕ solves (10.2) then there is a some constant c such that that

$$c = (\dot{\phi})^2 + 2\kappa e^{\phi} + 4e^{2\phi} = (\dot{\phi})^2 + (2e^{\phi} + \kappa/2)^2 - \kappa^2/4.$$
(10.3)

Then the function $u = e^{-\phi}$ must solve

$$(\dot{u})^2 = cu^2 - 2\kappa u - 4 = \operatorname{sgn}(c) \left(\sqrt{|c|}u - \frac{\operatorname{sgn}(c)\kappa}{\sqrt{|c|}}\right)^2 - \left(4 + \frac{\kappa^2}{c}\right). \quad (10.4)$$

(For the second equality of (10.4), assume $c \neq 0$). Conversely, if u is a C^{∞} positive solution of (10.4) then $\phi = -\log u$ solves $\dot{\phi}(\ddot{\phi} + \kappa e^{\phi} + 4e^{2\phi}) = 0$. There is a unique C^{∞} smooth solution ϕ of (10.2) with prescribed 1-germ at any given point. From (10.2) it follows that $4c + \kappa^2 > 0$ with equality if and only if ϕ is constant, equal to $\log(-\kappa/4)$, in which case it must be $\kappa < 0$. If $\kappa \ge 0$ then $\ddot{\phi} < 0$, so any zero of $\dot{\phi}$ is isolated and, moreover, ϕ has no local minimum, so M^* is not compact. If $\kappa < 0$ and ϕ is not constant, then $\kappa^2 > -4c$. If $\dot{\phi}(s_0) = 0 = \ddot{\phi}(s_0)$ then by (10.2) and (10.2) there holds $\kappa e^{\phi(s_0)} = c$; in particular c < 0. Substituting this into (10.3) gives $c(4c + \kappa^2) = 0$, which is a contradiction since neither c nor $\kappa^2 + 4c$ is zero. Hence if ϕ is not constant, then, whatever is κ , the zeroes of $\dot{\phi}$ are isolated, so that, by continuity, a solution of (10.4) determines a solution of (10.2). This shows that to solve (10.2) it suffices to find C^{∞} positive solutions of (10.4). Differentiating (10.4) shows that where \dot{u} is not zero, u solves

$$\ddot{u} = cu - \kappa. \tag{10.5}$$

Since a positive C^{∞} solution of (10.4) can be written as $u = e^{-\phi}$ for some ϕ solving (10.2), it follows from the isolation of the zeroes of $\dot{\phi}$ that the zeroes of \dot{u}

are isolated, and so u is a positive C^{∞} solution of (10.4) then it solves (10.5) subject to (10.4), viewed as a constraint.

The constant c in (10.4) can be interpreted as follows. If ϕ solves (10.2) then $\mathcal{R}_h = \kappa + 4e^{\phi}$. Since e^{ϕ} must extend smoothly to M it assumes a maximum value on M, and \mathcal{R}_h assumes its maximum value on M at this same point. Since e^{ϕ} must vanish off of M^* , it must assume its maximum in the interior of M^* , and so ϕ must assume a maximum in M^* as well. From (10.3) it follows that at such a point there holds

$$4c + \kappa^2 = \left(\max_M \mathcal{R}_h\right)^2. \tag{10.6}$$

10.5. The Einstein-Weyl structure $([\nabla], [h])$ resulting from a solution ϕ of (10.1) given X and κ will be said to be *determined by* (X, κ, ϕ) . Note that solutions of (10.1) need not be unique, and it can occur that the same Einstein-Weyl structure is determined by various triples (X, κ, ϕ) . This possibility will now be illustrated. Let $([\nabla], [h])$ be determined by (X, κ, ϕ) . If instead of X there is considered $\bar{X} = e^{-\lambda}X$, then $*\bar{h} = e^{2\lambda}h$, and $\bar{\phi} = \phi - \lambda$ is a solution of (10.1) with $\bar{\kappa} = e^{-\lambda}\kappa$ in place of κ and $*\bar{h}$ in place of *h. The resulting Gauduchon metric $\bar{h} = e^{\bar{\phi}}*\bar{h} = e^{\lambda}h$ is positively homothetic to h; since the resulting one-form $\bar{X}^p \bar{h}_{ip}$ is equal to $X^p h_{ip}$, the solution determined by $(\bar{X}, \bar{\kappa}, \bar{\phi})$ is the same as that determined by (X, κ, ϕ) .

The parameters \bar{r} and \bar{s} corresponding to \bar{X} and $J\bar{X}$ are related to r and s by $\bar{r} = e^{\lambda}r$ and $\bar{s} = e^{\lambda}s$. The function $\bar{u}(\bar{s})$ defined by $\bar{u}(\bar{s}) := e^{-\bar{\phi}}(\bar{s})$ is $\bar{u}(\bar{s}) = e^{\lambda}u(\bar{s}) = e^{\lambda}u(e^{\lambda}s)$ and solves $(\frac{\partial\bar{u}}{\partial\bar{s}})^2 = \bar{c}\bar{u}^2 - 2e^{-\lambda}\kappa\bar{u} - 4$ with $\bar{c} = e^{-2\lambda}c$. If u solves (10.4) then $(X, \kappa, -\log u)$ determines an Einstein-Weyl structure. The preceeding shows that the same Einstein-Weyl structure is determined by the triple $(\bar{X}, \bar{\kappa}, -\log \bar{u})$ resulting from the solution \bar{u} of (10.4) with $\bar{\kappa}$ and \bar{c} in place of κ and c. Thus solving (10.4) for distinct values of c related by a positive constant will not result in inequivalent Einstein-Weyl structures. It follows that when $c \neq 0$ the value of c can without loss of generality be normalized to be any given non-zero number having the same sign as has c. By (10.6) the geometric meaning of such a normalization is to fix the maximum value of the scalar curvature of h.

10.6. The Einstein-Weyl structures determined by (X, κ, ϕ) and (X, κ, ϕ) , in which $\bar{\phi}(s) = \phi(s - a)$ need not be the same, but they are always equivalent by an orientation-preserving diffeomorphism isotopic to the identity. Let $\tau_t : M \to M$ be the flow of -JX, the fixed points of which are the zeroes of X. The restriction to M^* of τ_t is given in (r, s) coordinates by $\tau_t(r, s) = (r, s - t)$. Evidently τ_t is an isometry of *h preserving X and satisfying $\bar{\phi} = \phi \circ \tau_a$. It follows that the resulting Gauduchon metrics $\bar{h} = e^{\bar{\phi}*h}$ and $h = e^{\phi*h}$ are related by $\bar{h} = \tau_a^*(h)$, and so the resulting Einstein-Weyl structures determine the same point in the deformation space $\mathcal{E}(M)$. The consequence of this observation relevant in the sequel is that the parameter s can always be modified by a translation, as is convenient.

10.7. Consider now the case of the torus. There are discussed first the constant solutions of (10.2), and then the nonconstant ones. By Theorem 7.7, ν must be

negative, so $\kappa < 0$ in (10.1). The only relevant solutions of (10.4) are those for which *u* is non-constant, positive, and bounded from above. For each (X, κ) , there is a constant solution to (10.1), namely $\phi = \log(-\kappa/4)$ (equivalently, the constant function $-4/\kappa$ solves (10.4) with $c = -\kappa^2/4$). Hence $h = -(\kappa/4)^*h$ is simply a flat representative of the given conformal structure [*h*]. The resulting γ_i is $-(\kappa/4)X^{p*}h_{ip}$, and the parameter ν of the resulting Einstein-Weyl structure is $\nu = \kappa \operatorname{vol}_h(M) = -\kappa^2 \operatorname{vol}_{*h}(M)/4$. Since, by the discussion in Section 10.5, $(e^{-\lambda}X, e^{-\lambda}\kappa, \log(-\kappa/4) - \lambda)$ determines the same Einstein-Weyl structure as does $(X, \kappa, \log(-\kappa/4))$, there can be imposed a normalization fixing X, and a convenient one is to require that $\operatorname{vol}_{*h}(M) = 4\pi^2$, so that $\nu = -\pi^2\kappa^2$. The resulting Gauduchon metric having volume $4\pi^2$ is simply *h, and the dual to γ_i in this metric is $Y^i = *h^{ip}\gamma_p = -(\kappa/4)X^i$.

Replacing X by $X^{\theta} = \cos \theta X + \sin \theta J X$ for any $\theta \in [0, 2\pi)$ gives rise to an Einstein-Weyl structure with the same v. Since the group of conformal automorphisms of a torus is conjugate to the elliptic modular group, which acts discretely and properly discontinuously on the upper half space, which is the Teichmüller space of the smooth torus, the Einstein-Weyl structures determined by $(X, \kappa, \log(-\kappa/4))$ and $(X^{\theta}, \kappa, \log(-\kappa/4))$ are equivalent modulo an element of $\text{Diff}_0(M)$ if and only if $\theta = 0$. It follows that distinct elements of the deformation space $\mathcal{E}(M)$ are obtained as (κ, θ) varies over $(-\infty, 0) \times [0, 2\pi)$, that is as $-\kappa e^{i\theta}$ varies over \mathbb{C}^{\times} . All the possibilities can be determined in terms of a given fixed X as follows. Let ${}^{*}h \in [h]$ be the flat representative of volume $4\pi^2$. Then to each $-\kappa e^{i\theta} \in \mathbb{C}^{\times}$ there corresponds an Einstein-Weyl structure such that the *h dual of the Faraday primitive is $Y^i = -(\kappa/4)X^{\theta}$, and these Einstein-Weyl structures determine distinct elements of the deformation space $\mathcal{E}(M)$ (they might not be distinct in the moduli space $\mathcal{M}(M)$; for instance for the square and hexagonal tori the modular group fixes some points). These Einstein-Weyl structures are characterized by having weighted scalar curvature equal to zero. The preceeding shows that the trivial complex line bundle over the upper half space parameterizes the Einstein-Weyl structures on a torus having zero weighted scalar curvature considered up to deformation.

An intrinsic formulation of the preceeding goes as follows. As explained in the proof of Theorem 7.7 an Einstein-Weyl structure on the torus which is closed but not exact determines a holomorphic affine connection. Namely, in this case a Gauduchon metric $h \in [h]$ with Levi-Civita connection D and Faraday primitive γ_i is flat, and the (1, 0) part $\nabla^{1,0} = D^{1,0} - 2\gamma^{(1,0)}$ is holomorphic because $\gamma^{(1,0)}$ is holomorphic. Moreover, every holomorphic affine connection on M has the form $D^{1,0} - 2\gamma^{(1,0)}$ for some holomorphic one-form $\gamma^{(1,0)}$ and the Levi-Civita connection D of a flat metric $h \in [h]$, so that the space of holomorphic affine connections on torus with a fixed complex structure is parameterized by $H^0(M, \mathcal{K}^1) \simeq \mathbb{C}$, the origin corresponding to $D^{1,0}$, where D is the Levi-Civita connection of any flat metric representing the conformal structure determined by the given complex structure. This description of the holomorphic affine connections on a two torus seems to have been first explicitly observed by A. Vitter in [73]. The preceeding can be summarized as saying that on the smooth torus the space of Einstein-Weyl structures which are closed but not exact considered up to equivalence modulo $\text{Diff}_0(M)$ is in bijection with the complement of the zero section in the bundle over the upper half space (the Teichmüller space of M) the fiber of which over a given conformal structure on M comprises the one complex dimensional vector space of holomorphic one forms.

10.8. Now there will be sought non-constant, positive, bounded solutions to (10.4) on the torus. By (10.6), $-\kappa^2 \le 4c$. The equation (10.5) can have positive bounded solutions only if c < 0, in which case the general solution is $u(s) = c^{-1}(\kappa - \sqrt{\kappa^2 + 4c} \cos(\sqrt{|c|s} - \alpha))$, in which α is arbitrary. By the discussion in Section 10.6, it can with no loss of generality be supposed that $\alpha = 0$. Thus $h = -c (\sqrt{\kappa^2 + 4c} \cos(\sqrt{|c|s}) - \kappa)^{-1} (dr^2 + ds^2)$, which has period $2\pi/\sqrt{|c|}$ in s.

X and JX are linearly independent commuting vector fields on \mathbb{T}^2 preserving *h, the composition of their flows defines an isometric action of \mathbb{R}^2 on \mathbb{T}^2 for which the stabilizers of any two points are conjugate, and the stabilizer of a given point is a discrete subgroup of \mathbb{R}^2 , so a lattice. The lifts to the universal cover of the flows of X and JX are by translations parallel to the generators of the lattice. A fundamental domain for the action of $\pi_1(\mathbb{T}^2)$ is a half-closed parallelogram. If h is to define a metric on \mathbb{T}^2 , it must be that the value of h is the same at every preimage of any $p \in \mathbb{T}^2$. This forces the periodicity of h in s to be commensurate with that of the fundamental domain, so that one closed side of the fundamental domain lies on the s axis, and the length of this side is a multiple of $2\pi/\sqrt{|c|}$ by a positive integer m. Since $\operatorname{vol}_{*h}(\mathbb{T}^2)$ is given by integrating $dr \wedge ds$ over the fundamental domain, the length of the side lying on the r-axis must be $\ell = \sqrt{|c|} \operatorname{vol}_{*h}(\mathbb{T}^2)/2\pi m$. Using that for b > 1 a primitive of $(\cos x + b)^{-1}$ is $\frac{2}{\sqrt{b^2-1}} \arctan\left(\sqrt{\frac{b-1}{b+1}} \tan \frac{x}{2}\right)$ yields

$$\operatorname{vol}_{h}(\mathbb{T}^{2}) = \int_{\mathbb{T}^{2}} u^{-1} \, dvol_{\tilde{h}} = \frac{2m\ell\sqrt{|c|}}{\sqrt{\kappa^{2}+4c}} \int_{0}^{\pi} \left(\cos t - \frac{\kappa}{\sqrt{\kappa^{2}+4c}}\right)^{-1} \, dt$$
$$= \pi m\ell = \sqrt{|c|} \operatorname{vol}_{*h}(\mathbb{T}^{2})/2.$$

By the discussion in Section 10.5 fixing either of c or $vol_{*h}(\mathbb{T}^2)$ determines the other, and such a choice can be made as is convenient without changing the equivalence class of the Einstein-Weyl structure which results; make the normalization c = -4, so that the metric h is

$$h = 4\left(\sqrt{\kappa^2 - 16\cos(2s) - \kappa}\right)^{-1} (dr^2 + ds^2), \tag{10.7}$$

in which $\kappa < -4$. By construction $\nu = \kappa \operatorname{vol}_h(\mathbb{T}^2) = \kappa \operatorname{vol}_{*h}(\mathbb{T}^2)$. Computing the scalar curvature of *h* using (2.1) gives

$$-\sqrt{\kappa^2 - 16} = \min_{\mathbb{T}^2} \mathcal{R}_h \le \mathcal{R}_h = \sqrt{\kappa^2 - 16} \left(\frac{\kappa \cos(2s) - \sqrt{\kappa^2 - 16}}{\sqrt{\kappa^2 - 16} \cos(2s) - \kappa} \right)$$

$$\le \max_{\mathbb{T}^2} \mathcal{R}_h = \sqrt{\kappa^2 - 16}.$$
(10.8)

For $\kappa = -4$ the metric *h* corresponds to the solutions described in Section 10.7 having $R \equiv 0$, while for $\kappa < -4$ these yield solutions have weighted scalar curvature which is somewhere positive and somewhere negative, as in {5} of Theorem 7.7. The metrics ${}^{\nu}g$ and ${}^{\nu}h$ homothetic to *h* and defined by the requirements that $\operatorname{vol}_{{}^{\nu}g}(\mathbb{T}^2) = 4\pi^2$ and $\max_{\mathbb{T}^2} \mathcal{R}_{{}^{\nu}h} = 2$, are ${}^{\nu}g = 4\pi^2 \operatorname{vol}_h(\mathbb{T}^2)^{-1}h$ and

$${}^{\nu}h = 2^{-1}\sqrt{\kappa^2 - 16} h = 2\left(\cos(2s) - \frac{\kappa}{\sqrt{\kappa^2 - 16}}\right)^{-1} \left(dr^2 + ds^2\right)$$

As $\nu \to -\infty$ (equivalently $\kappa \to -\infty$) the metrics ${}^{\nu}h$ tend pointwise to the degenerate metric ${}^{\infty}h = 2(1 + \cos 2s)^{-1}(dr^2 + ds^2)$, which is the hyperbolic metric of constant scalar curvature -2 on its domain of non-degeneracy, which is the strip $\mathbb{R} \times (-\pi/2, \pi/2)$. Since ${}^{*}h = dr^2 + ds^2$ and $|X|_{*h}^2 = 1$,

$$\gamma = 4(\sqrt{\kappa^2 - 16}\cos(2s) - \kappa)^{-1}dr, \quad |\gamma|_h^2 = 4(\sqrt{\kappa^2 - 16}\cos(2s) - \kappa)^{-1},$$

$$F = \frac{8}{\sqrt{\kappa^2 - 16}}\frac{\sin(2s)}{(\cos(2s) - \frac{\kappa}{\sqrt{\kappa^2 - 16}})^2}dr \wedge ds, \quad F_h = \frac{4\sin(2s)}{\cos(2s) - \frac{\kappa}{\sqrt{\kappa^2 - 16}}}.$$
 (10.9)

Note that $\Re_h - 4|\gamma|_h^2 = \kappa$, as must be the case by construction, and that $\Re_h^2 + F_h^2 = \kappa^2 - 16$, as required by Lemma 6.7. For b > 1 a primitive for $(\cos t + b)^{-2}$ is

$$\frac{1}{b^2 - 1} \left(\frac{2b}{\sqrt{b^2 - 1}} \arctan\left(\sqrt{\frac{b - 1}{b + 1}} \tan\left(\frac{t}{2}\right)\right) - \frac{\sin t}{\cos t + b} \right),$$

so that

$$\int_0^{\pi} \left(\cos t - \frac{\kappa}{\sqrt{\kappa^2 - 16}} \right)^{-2} dt = -\pi \kappa (\kappa^2 - 16)/64,$$

from which follows $4||\gamma||_h^2 = -\kappa \operatorname{vol}_{*h}(M) = -\nu$, as required by {5} of Theorem 7.7.

This completes the analysis of the torus case. The Einstein-Weyl structures on the torus which are not closed are determined up to the action of orientationpreserving diffeomorphisms by the choice of a conformal structure (a point in the hyperbolic plane) and a rational number (determining an inessential conformal Killing field). However, it is not clear how to describe the deformation/moduli space of solutions in a conceptual, geometric way (as was done in Section 10.7 for the closed but not exact Einstein-Weyl structures).

10.9. Now consider the case g = 0. In considering (10.4) on the sphere, the only difference with the analysis in the case of the torus is that the boundary conditions are different. The requirement that $|X|_h^2 = e^{\phi}$ forces that ϕ tends to $-\infty$ at the zeroes of X, or, what is the same, that u tends to $+\infty$ at the zeroes of X. Thus the only solutions of (10.4) that are relevant are those for which u is positive and tends to $+\infty$ at the boundary of \mathbb{S}^{2*} . Where $\dot{u} \neq 0$ there holds $\ddot{u} = cu - \kappa$; if c < 0 then the solutions are bounded, so it must be that $c \ge 0$.

10.10. For c = 0 the general solution of (10.4) is $u(s) = -\frac{\kappa}{2}s^2 + as + b$ for some constants *a* and *b*. In this case either *u* is equal to a positive constant, or κ is negative, for otherwise *u* would be somewhere negative. Evaluating (10.4) at s = 0 shows $a^2 = -2b\kappa - 4$. Since the coordinate *s* is determined only up to translation it can be supposed that a = 0 by making a translation in *s*. In this case $b = -2/\kappa$, so $u = -\frac{\kappa}{2}s^2 - \frac{2}{\kappa}$. Since \mathbb{S}^{2*} is the complement of two points, it is topologically a cylinder, and so its universal cover is the plane with the global coordinates *r* and *s*; in this case the metric $h = u^{-1}(dr^2 + ds^2)$ can descend to a metric on the cylinder obtained by quotienting by translations by 2π in the *r* direction. This metric gives rise to an Einstein-Weyl structure on the cylinder which is not exact and which has a distinguished complete metric representative $h \in [h]$ of infinite volume. Because this metric gives the cylinder infinite volume, there is no way it extends smoothly to \mathbb{S}^2 . Thus these solutions of (10.4) do not yield Einstein-Weyl structures on the sphere.

10.11. Now suppose that c > 0. The general solution of (10.4) which tends to $+\infty$ as $s \to \pm\infty$ is $u(s) = c^{-1}(\sqrt{\kappa^2 + 4c}\cosh(\sqrt{cs} + \alpha) + \kappa)$, in which α is arbitrary. By the discussion in Section 10.6, it can with no loss of generality be supposed that $\alpha = 0$. Following the discussion in Section 10.5, make the normalization c = 4. It will be convenient to write also $\mu = 2\kappa/\sqrt{\kappa^2 + 16}$, which ranges over (-2, 2) as $\kappa = 4\mu/\sqrt{4 - \mu^2}$ ranges over \mathbb{R} , and to introduce the coordinates $x = e^s \cos r$ and $y = e^s \sin r$, and to write $\rho = \sqrt{x^2 + y^2} = e^s$. Then

$$h = \frac{4(dr^2 + ds^2)}{\sqrt{\kappa^2 + 16}\cosh(2s) + \kappa} = \frac{8}{\sqrt{\kappa^2 + 16}} \frac{dx^2 + dy^2}{1 + \mu\rho^2 + \rho^4}$$

= $\frac{2}{\sqrt{\kappa^2 + 16}} \frac{1 + 2\rho^2 + \rho^4}{1 + \mu\rho^2 + \rho^4} \tilde{h}.$ (10.10)

(Recall $\tilde{h} = 4(1 + \rho^2)^{-2}(dx^2 + dy^2)$ is the metric of scalar curvature 2 and volume 4π). In this form it is evident that *h* extends smoothly through the origin of the *x* and *y* coordinates, which corresponds to the fixed point as $s \to -\infty$. Replacing *s*

by -s in the preceeding gives a coordinate system in which it is evident that h is smooth when $s \to \infty$. Thus the metric h extends to a smooth metric on all of the two-sphere which by construction generates an Einstein-Weyl structure of the sort in {6} of Theorem 7.7.

It will now be shown that for distinct values of κ the Einstein-Weyl structures so obtained are inequivalent. It is convenient to introduce the function

$$\tau(z) = 2z \arctan(\sqrt{z^2 + 1} - z) = z(\pi/2 - \arctan(z)), \quad (10.11)$$

which is a C^{∞} orientation preserving diffeomorphism of \mathbb{R} onto $(-\infty, 1)$. The parameter which does not depend on the scaling of *h* is $\nu = \kappa \operatorname{vol}_h(\mathbb{S}^2)$. The Kähler form associated to *h* is

$$\omega = 8(\kappa^2 + 16)^{-1/2} \rho (1 + \mu \rho^2 + \rho^4)^{-1} dr \wedge d\rho$$

= $dr \wedge d \left(\arctan\left(\kappa/4 + \sqrt{\kappa^2/16 + 1} \rho^2 \right) \right),$ (10.12)

and so

$$\nu = \kappa \operatorname{vol}_h(\mathbb{S}^2) = \kappa \int_{\mathbb{S}^2} \omega = 2\pi \kappa \left(\pi/2 - \arctan(\kappa/4) \right) = 8\pi \tau(\kappa/4).$$
(10.13)

Letting κ run over \mathbb{R} , this shows that all values of $\nu < 8\pi$ are realized by some Einstein-Weyl structure on the sphere.

Henceforth, κ will be viewed as a function of ν via $\kappa = 4\tau^{-1}(\nu/8\pi)$. However, because of the implicit nature of this definition, it will be convenient to continue writing κ and μ in formulas. Using (2.1) the scalar curvature \mathcal{R}_h of h can be computed by finding the Euclidean Laplacian of the conformal factor in (10.10). The result is

$$\mathcal{R}_{h} = \frac{\sqrt{\kappa^{2} + 16}}{2} \left(\frac{\mu + 4\rho^{2} + \mu\rho^{4}}{1 + \mu\rho^{2} + \rho^{4}} \right) = \kappa + \frac{32}{\sqrt{\kappa^{2} + 16}} \left(\frac{\rho^{2}}{1 + \mu\rho^{2} + \rho^{4}} \right). \quad (10.14)$$

For fixed μ , \mathcal{R}_h takes its maximum value on the equatorial circle $\rho = 1$, where its value is $\sqrt{\kappa^2 + 16}$, while \mathcal{R}_h attains no minimum in the plane, tending to κ as ρ tends to either 0 or ∞ . In particular,

$$\kappa = \min_{\mathbb{S}^2} \mathfrak{R}_h \le \mathfrak{R}_h \le \max_{\mathbb{S}^2} \mathfrak{R}_h = \sqrt{\kappa^2 + 16}.$$
 (10.15)

Observe that if ν is positive or non-negative, then the scalar curvature \mathcal{R}_h has the same property. Let ${}^{\nu}h$ and ${}^{\nu}g$ be the Gauduchon metrics of the Einstein-Weyl structure ([∇], [h]) corresponding to ν distinguished respectively by the requirements that $\max_{\mathbb{S}^2} \mathcal{R}_{\nu h} = 2$ and $\operatorname{vol}_{\nu g}(\mathbb{S}^2) = 4\pi$. Then ${}^{\nu}g = 4\pi \operatorname{vol}_h(\mathbb{S}^2)^{-1}h$ while

$${}^{\nu}h = (\sqrt{\kappa^2 + 16}/2)h = 4\left(1 + \mu\rho^2 + \rho^4\right)^{-1} \left(dx^2 + dy^2\right) = \frac{1 + 2\rho^2 + \rho^4}{1 + \mu\rho^2 + \rho^4}\tilde{h}.$$
 (10.16)

Hence

$$-2 \leq \mu = \min_{\mathbb{S}^2} \Re_{\nu_h} \leq \Re_{\nu_h} = \mu + (4 - \mu^2) \left(\frac{\rho^2}{1 + \mu \rho^2 + \rho^4} \right) \leq \max_{\mathbb{S}^2} \Re_{\nu_h} = 2,$$

$$4\pi \leq \operatorname{vol}_{\nu_h}(\mathbb{S}^2) = \frac{4\pi \sqrt{\kappa^2 + 16}}{\kappa} \tau(\kappa/4).$$
 (10.17)

As $\nu \to 8\pi$ (so $\kappa \to \infty$ and $\mu \to 2$) the metric ${}^{\nu}h$ tends pointwise to the spherical metric \tilde{h} . As $\nu \to -\infty$ (so $\kappa \to -\infty$ and $\mu \to -2$), the volume of ${}^{\nu}h$ goes to $+\infty$. On either of the disks complementary in \mathbb{S}^2 to the equatorial circle $\rho = 1$, the metric ${}^{\nu}h$ converges pointwise to the hyperbolic metric $4(1-\rho^2)^{-2}(dx^2+dy^2)$ of constant scalar curvature -2. The family ${}^{\nu}h$ interpolates between the spherical metric and the hyperbolic metric. The positive curvature concentrates on the equatorial circle as μ nears -2, while the negative curvature concentrates on the complementary disks. Precisely, for $\mu < 0$ the curvature is positive for $(-2 + \sqrt{4 - \mu^2})/\mu < \rho^2 < -(2 + \sqrt{4 - \mu^2})/\mu$.

By construction the vector field X^i is $x \partial_y - y \partial_x = \partial_r$. Explicit expressions for $\gamma_i = X^p h_{pi}$ and its norm are

$$\gamma = \frac{8}{\sqrt{\kappa^2 + 16}(1 + \mu\rho^2 + \rho^4)} (xdy - ydx),$$

$$|\gamma|_h^2 = \frac{8\rho^2}{\sqrt{\kappa^2 + 16}(1 + \mu\rho^2 + \rho^4)},$$
(10.18)

the second of which in any case follows from $\Re_h - \kappa = 4|\gamma|_h^2$ in conjunction with (10.14). Consequently,

$$F = -d\gamma = \frac{16}{\sqrt{\kappa^2 + 16}} \frac{\rho^4 - 1}{(1 + \mu\rho^2 + \rho^4)^2} dx \wedge dy, \ F_h^2 = \frac{16(1 - \rho^4)^2}{(1 + \mu\rho^2 + \rho^4)^2}.$$
 (10.19)

Equation (10.19) shows that the Faraday curvature is non-zero except along the equatorial circle, and is largest at the poles. There can be verified the equivalent

$$\Re_h^2 + F_h^2 = \kappa^2 + 16,$$
 $\Re_{\nu_g}^2 + F_{\nu_g}^2 = 16\mu^{-2}\tau(\kappa/4)^2,$ (10.20)

confirming in this case the claim of Lemma 7.4. By Theorem 7.5 the equatorial circle, along which vanishes *F*, must be a geodesic; its *h*-length and ^{*v*}*h*-length are respectively $\pi(\sqrt{\kappa^2 + 16} - \kappa)^{1/2}$ and $2\sqrt{2}\pi(\kappa^2 + 16)^{1/4}(\sqrt{\kappa^2 + 16} + \kappa)^{-1/2}$.

A map associating to an Einstein-Weyl structure on the sphere the real part of a holomorphic vector field is determined by a manner of choosing a normalized Gauduchon metric. For example, the metric *h* is normalized by specifying the minimum value assumed by its scalar curvature, which is the parameter κ . Similarly, ^v*h* is determined by the requirement that $\max_{\mathbb{S}^2} \Re_{vh} = 2$. The associated vector field $Y^i = {}^{\nu}h^{ij}\gamma_j$ is $\frac{16}{\sqrt{\kappa^2+16}}(x\partial_y - y\partial_x)$; observe that this vector field is associated to two distinct Einstein-Weyl structures, those corresponding to $\pm\kappa$. The normalization like that used for surfaces of higher genus selects the metric ${}^{\nu}g$ having volume 4π . The associated vector field $Z^i = {}^{\nu}g^{ij}\gamma_j$ is $2\kappa^{-1}\tau(\kappa/4)(x\partial_y - y\partial_x)$. Since $\kappa \to 2\kappa^{-1}\tau(\kappa/4)$ is an orientation-reversing diffeomorphism of \mathbb{R} onto $(0, \pi/2)$, in this case there is a unique vector field associated to each Einstein-Weyl structure.

It is noted in passing that the claims about magnetic geodesics made immediately following the statement of Theorem 7.5 follow straightforwardly from the explicit form of Y^i and (10.18) and (10.19).

In the coordinate system (x, y) let z = x + iy and view z as the inhomogeneous coordinate on the standard chart in $\mathbb{P}^1(\mathbb{C})$. To an element $\Phi = \begin{pmatrix} a & b \\ c & d \end{pmatrix} \in \mathfrak{sl}(2, \mathbb{C})$ associate the holomorphic vector field X^{Φ} on $\mathbb{P}^1(\mathbb{C})$ defined by $X_p^{\Phi} = \frac{d}{dt}_{|t=0} \exp(t\Phi)p$ which in the inhomogeneous coordinate z has the form $(b + (a - d)z - cz^2)\partial_z$. An element $\Psi \in \mathrm{PSL}(2, \mathbb{C})$ is elliptic if and only if $(\mathrm{tr} \Psi)^2/\det \Psi \in [0, 4)$, in which case Ψ is conjugate in $\mathrm{PSL}(2, \mathbb{C})$ to a unique element of the form $\exp \Psi_{\theta}$ where $\Psi_{\theta} := \begin{pmatrix} \mathrm{i}\theta & 0 \\ 0 & -\mathrm{i}\theta \end{pmatrix}$ with $\theta \in (0, \pi/2)$. The real part of $X^{\Psi_{\theta}} = 2\mathrm{i}\theta z\partial_z$ is $\theta(x\partial_y - y\partial_x)$. It follows that the vector field Z described in the preceeding paragraph is the real part of $X^{\Psi_{2\tau(\kappa/4)/\kappa}}$. Since $\kappa \to 2\tau(\kappa/4)/\kappa$ is a diffeomorphism of \mathbb{R} onto $(0, \pi/2)$, this shows that each Einstein-Weyl structure on the sphere determines a unique conjugacy class of elliptic elements in $\mathrm{PSL}(2, \mathbb{C})$. Precisely, the vector field Z is the real part of the holomorphic vector field generated by the elliptic transformation given in the inhomogeneous coordinate by $z \to e^{4\mathrm{i}\tau(\kappa/4)/\kappa}z = \frac{\kappa + 4\mathrm{i}}{\sqrt{\kappa^2 + 16}}z$. The parameter θ is expressed in terms of the scale invariant parameter ν are expressed in terms of θ by

$$\kappa = 4 \cot 2\theta, \qquad \qquad \nu = 16\pi\theta \cot 2\theta. \qquad (10.21)$$

Note that no Einstein-Weyl structure on the two sphere gives rise to the conjugacy class of elliptic elements corresponding to $\theta = \pi/2$. What distinguished these elliptic elements is that the associated homography leaves invariant a circle. There is an Einstein-Weyl structure corresponding to $\theta = \pi/2$, namely that generated by the hyperbolic metric and corresponding to the degeneration as $\kappa \to -\infty$. This picture was already described in the last paragraph of [15, Section 5]. Here the equivalence problem is resolved explicitly. In particular it is shown that two Einstein-Weyl structures on the sphere are equivalent if and only if their vortex parameters ν are the same, and that the corresponding extended elliptic homography can be explicitly described by a rotation by an angle explicitly expressible in terms of ν , namely via (10.21).

An Einstein-Weyl structure ($[\nabla]$, [h]) on the sphere having vortex parameter ν is equivalent to one represented by the Gauduchon metric ${}^{\nu}g$ of volume 4π where

the explicit expressions for ${}^{\nu}g$, $\mathcal{R}_{{}^{\nu}g}$, and γ in terms of $\theta \in [0, \pi/2)$ defined from ν by (10.21) are

$${}^{\nu}g = \frac{2\sin 2\theta}{\theta(1+2\rho^2\cos 2\theta+\rho^4)} \left(dx^2+dy^2\right),$$

$$\gamma = \frac{2\sin 2\theta}{1+2\rho^2\cos 2\theta+\rho^4} \left(xdy-ydx\right).$$

$$\Re_{\nu g} = \frac{\theta}{\sin 2\theta} \left(\frac{4\cos 2\theta+8\rho^2+4\rho^4\cos 2\theta}{1+2\rho^2\cos 2\theta+\rho^4}\right).$$
(10.22)

By the uniformization theorem, $T(\mathbb{S}^2)$ is a point. Since Map⁺(\mathbb{S}^2) is trivial the moduli space $\mathcal{M}(\mathbb{S}^2)$ equals the deformation space $\mathcal{E}(\mathbb{S}^2)$. Since under the usual definition the identity element of $PSL(2, \mathbb{C})$ is not considered to be elliptic, it will be convenient to call **extended elliptic** a homography Ψ that is either elliptic or the identity. For such Ψ there holds $(tr \Psi)^2/\det \Psi \in [0, 4]$. Note, however, that $(tr \Psi)^2/\det \Psi \in [0, 4]$ need not imply that Ψ is extended elliptic because while the function $(tr \Psi)^2/det \Psi$ distinguishes the conjugacy classes of non-identity elements, it does not distinguish a parabolic transformation from the identity. Let $\mathcal{C}_{ell}(PSL(2,\mathbb{C}))$ denote the space of conjugacy classes of extended elliptic elements in PSL(2, \mathbb{C}). Suppose given an Einstein-Weyl structure ([∇], [h]) on the two sphere which is not that generated by the uniformizing conformal structure. After a diffeomorphism it may be supposed that [h] is the standard conformal structure. Associate to (∇, h) the element $\Psi \in \mathfrak{sl}(2, \mathbb{C})$ such that the (1, 0) part of the vector field $X^i = h^{ip} \gamma_n$, where $h \in [h]$ is the Gauduchon metric with volume 4π and γ is the corresponding Faraday primitive, is equal to X^{Ψ} . The element exp Ψ is elliptic and conjugate to $\exp \Psi_{\theta}$ for $\theta = \nu/16\tau^{-1}(\nu/8\pi)$. Applying the same construction to the pull-back of $([\nabla], [h])$ by an element of PSL(2, $\mathbb{C})$ yields an element exp Ψ' , conjugate to exp $\Psi_{\theta'}$ for some θ' . However, since ($[\nabla], [h]$) and its pull-back determine the same parameter v there holds $\theta' = v/16\tau^{-1}(v/8\pi) = \theta$, and so Ψ' is conjugate to Ψ . Hence the map associating to $([\nabla], [h])$ the infinitesimal generator Ψ descends to an injection $\mathcal{M}(\mathbb{S}^2) \to \mathcal{C}_{ell}(PSL(2,\mathbb{C}))$; the image omits the conjugacy class of the simple inversions corresponding to $\theta = \pi/2$. The map sending $[\Psi] \in \mathcal{C}_{ell}(PSL(2,\mathbb{C}))$ to the unique $\theta \in [0,\pi/2]$ such that $4\cos^2\theta = (\mathrm{tr}\,\Psi)^2/\det\Psi$ identifies the space $\mathcal{C}_{ell}(\mathrm{PSL}(2,\mathbb{C}))$ with the interval $[0, \pi/2]$, and the subspace $[0, \pi/2)$ corresponds to $\mathcal{M}(\mathbb{S}^2)$. While the topologies on the spaces $\mathcal{M}(\mathbb{S}^2)$ and $\mathcal{C}_{ell}(PSL(2,\mathbb{C}))$ have not been discussed, it makes sense to regard the interval $[0, \pi/2]$ as the compactification of $\mathcal{M}(\mathbb{S}^2)$. Theorem 10.3 summarizes the preceeding discussion.

Theorem 10.3. The following spaces are in pairwise bijection

- (1) The moduli space $\mathcal{M}(\mathbb{S}^2)$ of Einstein-Weyl structures on \mathbb{S}^2 .
- (2) The space of conjugacy classes of extended elliptic elements of $PSL(2, \mathbb{C})$ which leave invariant no circle.
- (3) The half-open interval $[0, \pi/2)$.

Precisely, to $\theta \in [0, \pi/2)$ correspond the conjugacy class of the homography $z \to e^{2i\theta}z$ and the equivalence class $\{[\nabla], [h]\}_{\theta}$ of Einstein-Weyl structures having vortex parameter $v = 16\pi\theta \cot 2\theta \in (-\infty, 8\pi]$. To $\theta = 0$ corresponds the Einstein-Weyl structure generated by the standard round metric. The equivalence class $\{[\nabla], [h]\}_{\theta}$ is represented by an Einstein-Weyl structure $([\nabla], [h])_{\theta}$ for which [h] is the standard conformal structure on $\mathbb{P}^1(\mathbb{C})$, and the Faraday primitive γ of the Gauduchon class and the Gauduchon metric ${}^{\nu}g \in [g]$ of volume 4π are as in (10.22). For every $\theta \in (0, \pi/2)$ the zero set of the Faraday curvature of the associated Einstein-Weyl structure is the equatorial circle. For $([\nabla], [h])_{\theta}$ the Gauduchon metric ${}^{\nu}h \in [h]$ such that $\max_{\mathbb{S}^2} \Re_{\nu h} = 2$ is ${}^{\nu}h = \frac{\theta}{2\sin 2\theta} {}^{\nu}g$. As $\theta \to 0$, ${}^{\nu}h$ tends pointwise to the Fubini-Study metric \tilde{h} , while as $\theta \to \pi/2$, the restriction to either connected component of the complement of the zero set of the Faraday curvature of the metric ${}^{\nu}h$ tends pointwise to the hyperbolic metric of constant scalar curvature -2.

10.12. Differentiating the family (10.10) of metrics with respect to the parameter $t \in (-\pi/2, 0)$ defined by $\kappa = -2 \cot 2t$ and comparing the result with (10.14) shows that the one-parameter family of metrics h(t) constitutes a solution to the Ricci flow, $\frac{d}{dt}h = -\Re_h h$. The one-parameter family s(t) = -ih(it) of metrics on the sphere obtained from h(t) after conjugation by a rotation in the parameter space is an ancient solution of the Ricci flow having positive curvature discovered in [29], where these metrices were called *sausage metrics*; they are known to mathematicians as the King-Rosenau metrics. Similarly, the family (10.7) of metrics on the torus is a solution of the Ricci flow with respect to the parameter $t \in (0, \infty)$ defined by $\kappa = -4 \coth 4t$. It would be interesting to explain conceptually why solutions of the Ricci flow arise naturally from Einstein-Weyl structures.

11. Convexity and Hessian metrics

In this section it is shown that the cone over an exact Einstein AH structure with negative scalar curvature carries particularly nice Riemannian and Lorentzian Hessian metrics, and there is proved Theorem 11.2. These constructions are used in Section 11.8, where it is explained that such an Einstein AH structure is determined by its underlying flat projective structure, and there is proved Theorem 1.1.

11.1. Following [23], a pair $(\widehat{\nabla}, g)$ comprising a flat affine connection $\widehat{\nabla}$ and a pseudo-Riemannian metric g such that around each point there are local affine coordinates x^i (meaning that the dx^i are $\widehat{\nabla}$ -parallel) and a **potential function** F such that $g_{ij} = \widehat{\nabla}_i dF_j$, is called a **Kähler affine metric**. A Kähler affine metric will be said to be a **Hessian metric** if the potential function is globally defined, as will be the case in the examples constructed in what follows. (This terminological distinction between *Hessian metric* and *Kähler affine metric* is not standard).

Let $(\widehat{\nabla}, g)$ be a Kähler affine metric. Let $\mu = dx^1 \wedge \cdots \wedge dx^{n+1}$ be the volume form determined by the choice of local affine coordinates. Then det_{μ} g :=

 $(\det g)/\mu^2$ is a function. It is not well-defined, but its logarithmic differential $d \log \det_{\mu} g$ is, because changing the choice of affine coordinates only changes μ by a constant factor. The **Ricci tensor** of a Kähler affine metric is defined to be $-\widehat{\nabla}_i d \log(\det_{\mu} g)_j$. Note that this Ricci tensor is not in general the Ricci tensor of either $\widehat{\nabla}$ or g. Rather it is defined in analogy with the Ricci form of a Kähler metric. A Kähler affine metric is said to be Einstein if its Ricci tensor is a constant multiple of g. Ricci flat Kähler affine metrics should be seen as real analogues of Calabi-Yau manifolds. Here they will be called **Monge-Ampère metrics**, as in [48].

11.2. Let M be a surface. Let $\rho : \hat{M} \to M$ be the principal \mathbb{R}^{\times} bundle over M such that the third power of \hat{M} is equal to the complement Det $T^*M \setminus \{0\}$ in Det T^*M of the zero section, viewed as a principal bundle. Let \mathcal{V} be the line bundle associated to \hat{M} the sections of which are identified with homogeneity 1 functions on the total space of \hat{M} . A section of \mathcal{V}^3 is naturally viewed as a section of Det TM. A section of \mathcal{V}^k will be said to have **weight** k. Let R_r denote dilation in the fibers of \hat{M} by $r \in \mathbb{R}^{\times}$, and let \mathbb{E} be the vector field generating the flow R_{e^t} . If $u \in \Gamma(\mathcal{V}^k)$ then the associated equivariant function $\tilde{u} \in C^{\infty}(\hat{M})$ has homogeneity k; in particular $d\tilde{u}(\mathbb{E}) = k\tilde{u}$. On the total space of \hat{M} there is a tautological 2-form μ defined for $X, Y \in T_{\theta}\mathcal{V}$ by $\mu_{\theta}(X, Y) = \theta^3(T\rho(\theta)(X), T\rho(\theta)(Y))$, in which θ^3 is viewed as a 2-form on $T_{\rho(\theta)}M$. It is straightforward to check that $\Psi := d\mu$ is a volume form.

The existence part of Theorem 11.1 is due to T.Y. Thomas; see [67]. That M have dimension 2 in Theorem 11.1 is unimportant, but greater generality is not needed here. It is convenient to use uppercase Latin letters as abstract indices on \hat{M} .

Theorem 11.1. Let M be a smooth surface equipped with a projective structure $[\nabla]$. There is a unique torsion-free affine connection $\widehat{\nabla}$ on \widehat{M} satisfying

- (1) $\widehat{\nabla}_I \mathbb{E}^J = \delta_I^{J}$.
- (2) $\widehat{\nabla}\Psi = 0.$
- (3) $\widehat{\nabla}$ is Ricci flat.
- (4) $R_r^*(\widehat{\nabla}) = \widehat{\nabla} \text{ for all } r \in \mathbb{R}^{\times}.$
- (5) The inverse image in \hat{M} of a projective geodesic of M is a totally geodesic surface in \hat{M} tangent to \mathbb{E} .
- (6) The curvature $\hat{R}_{IJK}^{\ \ L}$ of $\widehat{\nabla}$ satisfies $\widehat{\nabla}_Q \hat{R}_{IJK}^{\ \ Q} = 0$.

The connection $\widehat{\nabla}$ is the **Thomas connection** associated to $[\nabla]$. The assignment $[\nabla] \to \widehat{\nabla}$ is equivariant with respect to the action of Diff(M) on M in the sense that for $\phi \in \text{Diff}(M)$ there holds $L(\phi)^*(\widehat{\nabla}) = \widehat{\phi^*([\nabla])}$ in which $L(\phi)$ is the unique principal bundle automorphism of \widehat{M} covering ϕ and preserving the tautological two-form μ . Moreover, $L(\phi) \in \text{Aut}(\widehat{\nabla}, \Psi)$ if and only if $\phi \in \text{Aut}([\nabla])$.

Indication of proof. A principal connection on \hat{M} determines a principal connection on Det T^*M and vice-versa. A torsion-free affine connection induces a principal connection on Det T^*M and hence on \hat{M} . It is easily seen that each principal

connection β on \hat{M} is induced by a unique torsion-free ∇ representing $[\nabla]$. Fix a principal connection β on \hat{M} and let $X \to \hat{X}$ be the horizontal lift of $X \in \Gamma(TM)$ to $\hat{X} \in \Gamma(\hat{M})$ determined by β . Note that $[\hat{X}, \hat{Y}] = \widehat{[X, Y]} - \rho^*(\omega(X, Y))\mathbb{E}$, in which $\rho^*(\omega) = d\beta$. Let $\nabla \in [\nabla]$ be the representative determined by β . Define $\widehat{\nabla}$ by requiring that it be torsion-free, that it satisfy $\widehat{\nabla}_I \mathbb{E}^J = \delta_I^{-J}$, and that for any $X, Y \in \Gamma(TM)$ there hold $\widehat{\nabla}_{\hat{X}} \hat{Y} = \widehat{\nabla}_X Y + P(X, Y)\mathbb{E}$, in which $P_{ij} = -R_{(ij)} - \frac{1}{3}R_{[ij]} = -R_{(ij)} - \frac{1}{2}\omega_{ij}$. That $\widehat{\nabla}$ verifies all the stated conditions is straightforward. That it does not depend on the choice of β can be verified directly, but it is probably easier, and conceptually better, to deduce this as a consequence of the claimed uniqueness. Verifying the uniqueness is a bit more involved; in this regard note that it is straightforward to construct examples on $\mathbb{R}^3 \setminus \{0\} \to \mathbb{S}^2$ showing the necessity, for the uniqueness, of the condition $R_r^*(\widehat{\nabla}) = \widehat{\nabla}$.

The curvature \hat{R}_{IJK}^{L} of $\widehat{\nabla}$ is given by

$$\hat{R}_{IJK}{}^L = \rho^*(C)_{IJK} \mathbb{E}^L, \qquad (11.1)$$

in which $C_{ijk} = 2\nabla_{[i}P_{j]k}$ is the projective Cotton tensor of $[\nabla]$. In particular $\widehat{\nabla}$ is flat if and only if $[\nabla]$ is projectively flat. For later use note that if β is the principal connection on \widehat{M} induced by $\nabla \in [\nabla]$ then there holds

$$\widehat{\nabla}_I \beta_J = -\beta_I \beta_J - \rho^*(P)_{IJ}, \qquad (11.2)$$

in which P_{ij} is the modified Ricci tensor of ∇ .

11.3. For a section $u \in \Gamma(\mathcal{V})$ there holds $\tilde{\nabla}_i u = \nabla_i u + \gamma_i u$ if $\tilde{\nabla} = \nabla + 2\gamma_{(i}\delta_{j)}^k$. Using this it can be verified that the operator $\mathcal{B}(u)_{ij} := \nabla_i \nabla_j u - P_{ij} u$ is projectively invariant in the sense that it does not depend on the choice of $\nabla \in [\nabla]$. If \tilde{u} is the homogeneity 1 function on \hat{M} corresponding to u then the symmetric tensor $\hat{\nabla}_I d\tilde{u}_J$ has the properties that $\mathbb{E}^I \hat{\nabla}_I d\tilde{u}_J = 0$ and that for all $X, Y \in \Gamma(TM), \hat{X}^I \hat{Y}^J \hat{\nabla}_I d\tilde{u}_J$ is the homogeneity 1 function on \hat{M} corresponding to $\mathcal{B}(u)_{ij} X^i Y^j$. Similarly, if $v = u^2$ (equivalently $\tilde{v} = \tilde{u}^2$), then

$$\frac{1}{2}\widehat{\nabla}_{I}d\widetilde{v}_{J} = \begin{pmatrix} u\mathcal{B}(u)_{ij} + \nabla_{i}u\nabla_{j}u \ u\nabla_{i}u \\ u\nabla_{j}u \ u^{2} \end{pmatrix},$$
(11.3)

in which the right-hand side is a notationally abusive shorthand utilizing the splitting of $T\hat{M}$ determined by β and signifying, for example, that $\hat{X}^I \mathbb{E}^J \widehat{\nabla}_I d\tilde{v}_J$ is the equivariant function corresponding to $2u\nabla_X u$. Since det $\mathcal{B}(u)$ is naturally viewed as a section of $(\text{Det } T^*M)^2 \otimes \mathcal{V}^2 \simeq \mathcal{V}^{-4}$, the operator $\mathcal{M}(u) := u^4 \det \mathcal{B}(u)$ is function-valued. Calculating det $\widehat{\nabla} d\tilde{v}$ by applying elementary row and column operations to (11.3) yields

$$\det \widehat{\nabla} d\widetilde{v} = 8\rho^*(\mathcal{M}(u))\Psi^2. \tag{11.4}$$

It follows immediately that $\mathcal{M}(u)$ is constant if and only if det $\widehat{\nabla} d\widetilde{v}$ is $\widehat{\nabla}$ -parallel.
Applying (11.1) and the Ricci identity yields

$$2\widehat{\nabla}_{[I}\widehat{\nabla}_{J]}d\widetilde{v}_K = -\widehat{R}_{IJK} \,^Q d\widetilde{v}_Q = -2\widetilde{v}\rho^*(C)_{IJK},\tag{11.5}$$

so that, if $\widehat{\nabla}_I d\widetilde{v}_J$ is non-degenerate, then it forms with $\widehat{\nabla}$ a Hessian metric if $[\nabla]$ is projectively flat.

11.4. Let *M* be a surface and let $([\nabla], [h])$ be an exact Riemannian signature Einstein AH structure with negative weighted scalar curvature *R*. Let $u = (R/2)^{-1/3} \in \Gamma(\mathcal{V})$ and $v = u^2 = 2^{2/3}R^{-2/3}$. Because $\nabla_i u = 0$ for the aligned representative $\nabla \in [\nabla]$, the principal connection β_I induced by ∇ is $\tilde{u}^{-1}d\tilde{u}_I = \frac{1}{2}\tilde{v}^{-1}d\tilde{v}_I$. Let $F = -\frac{3}{2}\log\tilde{v} = -3\log|\tilde{u}|$, which is logarithmically homogeneous in the sense that $R_{e^t}^*(F) = F - 3t$. Define covariant symmetric two-tensors g_{IJ} and f_{IJ} on \hat{M} by $g_{IJ} := \frac{1}{2}\widehat{\nabla}_I d\tilde{v}_J$ and $f_{IJ} = \widehat{\nabla}_I dF_J$. It will be shown that with ∇ the (1, -2) signature metric g_{IJ} and the Riemannian metric f_{IJ} form with $\widehat{\nabla}$ Hessian metrics which are respectively Monge-Ampère and Einstein Kähler affine.

Let \tilde{R} be the homogeneity -3 function on \hat{M} corresponding to R, and note that $d\tilde{R}(\hat{X})$ is the homogeneous function corresponding to $\nabla_X R$, so equals 0. By definition and (11.2) there hold

$$g_{IJ} = \frac{1}{2} \widehat{\nabla}_{I} d\tilde{v}_{J} = \tilde{v} \left(\widehat{\nabla}_{I} \beta_{J} + 2\beta_{I} \beta_{J} \right) = \tilde{v} \left(\beta_{I} \beta_{J} - \rho^{*}(P)_{IJ} \right)$$

$$= 2^{2/3} \tilde{R}^{-2/3} \left(\beta_{I} \beta_{J} + \frac{1}{2} \rho^{*}(RH)_{IJ} \right), \qquad (11.6)$$

$$f_{IJ} = -3 \widehat{\nabla}_{I} \beta_{J} = -3 \tilde{v}^{-1} g_{IJ} + 6\beta_{I} \beta_{J} = 3 \rho^{*}(P)_{IJ} + 3\beta_{I} \beta_{J}.$$

As RH_{ij} is an unweighted tensor, its pull-back $\rho^*(RH)_{IJ}$ has sense. From (11.6) it is apparent that g_{IJ} has signature (1, -2) and f_{IJ} is Riemannian. By (5.14) $[\nabla]$ is projectively flat, and so each of g_{IJ} and f_{IJ} forms with $\widehat{\nabla}$ a Hessian metric. Because $\nabla_i u = 0$ there holds $\mathcal{B}(u) = -P_{ij}u = (R/2)H_{ij}u = u^{-2}H_{ij}$, and so $\mathcal{M}(u) = 1$. Hence by (11.4) and the definition of g_{IJ} there holds

$$\det g = \Psi^2. \tag{11.7}$$

In particular det g is $\widehat{\nabla}$ -parallel, so that $(\widehat{\nabla}, g_{IJ})$ is a Monge-Ampère Hessian metric of signature (1, -2). Since $\mathbb{E}^{P}g_{IP} = \widetilde{v}\beta_{I}$ there holds $g^{IP}\beta_{P} = \widetilde{v}^{-1}\mathbb{E}^{I}$, and so $f_{IP}g^{JP} = -3\widetilde{v}^{-1}\delta_{I}^{J} + 6\widetilde{v}^{-1}\beta_{I}\mathbb{E}^{J}$, from which it follows that

$$\det f = 27\tilde{v}^{-3} \det g = 27e^{2F}\Psi^2, \tag{11.8}$$

so that $(\widehat{\nabla}, f_{IJ})$ is a Riemannian signature Einstein Hessian metric.

11.5. Alternatively, $\widehat{\nabla}$ and g_{IJ} generate an exact flat AH structure ([$\widehat{\nabla}$], [g]) with $\widehat{\nabla}$ as the aligned representative and g_{IJ} as a distinguished metric. The cubic torsion $\widehat{\mathcal{L}}_{IJ}{}^{K}$ is $\widehat{\mathcal{L}}_{IJ}{}^{K} = g^{KQ} \nabla_{I} g_{JQ} = g^{KQ} \nabla_{Q} g_{IJ}$. It is convenient to write $\widehat{\mathcal{L}}_{IJK} =$

 $\hat{\mathcal{L}}_{IJ} \,^{Q}g_{KQ} = \nabla_{I}g_{JK} = \frac{1}{2}\nabla_{I}\nabla_{J}d\tilde{v}_{K}$. The Levi-Civita connection \hat{D} of g_{IJ} is $\hat{D} = \hat{\nabla} + \frac{1}{2}\hat{\mathcal{L}}_{IJ}^{K}$, and the Levi-Civita connection \tilde{D} of f_{IJ} is

$$\tilde{D} = \hat{D} - 2\beta_{(I}\delta_{J)}{}^{K} - \frac{1}{3}f_{IJ}\mathbb{E}^{k} + 2\beta_{I}\beta_{J}\mathbb{E}^{K}.$$
(11.9)

Since $\mathbb{E}^I \widehat{\nabla}_I g_{JK} = (\mathfrak{L}_{\mathbb{E}} g)_{JK} - 2g_{JK} = 0$, there holds $\widehat{D}_I \mathbb{E}^J = \delta_I^J$, and, since $\mathbb{E}^P f_{IP} = 3\beta_I$, there hold $\widetilde{D}_I \mathbb{E}^J = 0$ and $\widetilde{D}_I \beta_J = 0$.

Because $\widehat{\nabla}_I d\widetilde{v}_J$ is non-degenerate, and because of the form of g_{IJ} it is evident that the submanifolds $M_{c^2} = \{p \in \widehat{M} : \widetilde{v}(p) = c^2\}$ are immersed and space-like for c > 0. In particular, the induced metric $\widetilde{v}^*(g)_{ij}$ is $c^{-1}h_{ij}$. Because $\mathbb{E}^I \mathbb{E}^J g_{IJ} = -\widetilde{v}$, the vector field $N^I = \widetilde{u}^{-1} \mathbb{E}^I$ is a unit normal along M_{c^2} . Since $\widehat{D}_I N^J = \widetilde{u}^{-1} \delta_I {}^J - \beta_I N^J$, the hypersurface M_{c^2} is totally umbilic with constant mean curvature with respect to g_{IJ} . Similarly, because $\mathbb{E}^I \mathbb{E}^J f_{IJ} = 3$, the \widetilde{D} -parallel vector field $\frac{1}{\sqrt{3}} \mathbb{E}^I$ is an f-unit normal to the submanifolds M_{c^2} , which are therefore totally geodesic with respect to f_{IJ} . The preceeding is summarized in Theorem 11.2.

Theorem 11.2. On a smooth surface M, let $([\nabla], [h])$ be an exact Riemannian signature Einstein AH structure with negative weighted scalar curvature R. The metric g_{IJ} defined in (11.6) forms with the Thomas connection $\widehat{\nabla}$ on \widehat{M} a Lorentzian signature Monge-Ampère Hessian metric structure, such that the level surfaces of \widehat{R} are smoothly immersed and, with respect to g_{IJ} are space-like, umbilic, and have constant mean curvature. The metric f_{IJ} defined in (11.6) forms with $\widehat{\nabla}$ a Riemannian signature Einstein Hessian structure, such that the level surfaces of \widehat{R} are totally geodesic surfaces with respect to f_{IJ} .

11.6. Let $\hat{\mathcal{R}}_{IJK}^{L}$ be the curvature of \hat{D} . From the flatness of $\hat{\nabla}$ and

$$\widehat{\nabla}_{[I} \hat{\mathcal{L}}_{J]K}{}^{L} = -\frac{1}{2} \hat{R}_{IJK}{}^{L} - \frac{1}{2} \hat{R}_{IJ}{}^{L}{}_{K} - \hat{\mathcal{L}}_{Q[I}{}^{L} \hat{\mathcal{L}}_{J]K}{}^{Q} = -\hat{\mathcal{L}}_{Q[I}{}^{L} \hat{\mathcal{L}}_{J]K}{}^{Q},$$
(11.10)

there results $\hat{\mathcal{R}}_{IJK}{}^{L} = -\frac{1}{2}\hat{\mathcal{L}}_{Q[I}{}^{L}\hat{\mathcal{L}}_{J]K}{}^{Q}$, and so $\hat{\mathcal{R}}_{IJ} = \frac{1}{4}\hat{\mathcal{L}}_{IA}{}^{B}\hat{\mathcal{L}}_{JB}{}^{A}$ and $\hat{\mathcal{R}}_{g} = \frac{1}{4}|\hat{\mathcal{L}}|_{g}^{2}$. Since \hat{M} is 3-dimensional the metric g is conformally flat, and so its curvature is completely determined by its Ricci curvature $\hat{\mathcal{R}}_{IJ}$.

Since $R\mathcal{L}_{ijk}$ is an unweighted tensor, it makes sense to write $\rho^*(R\mathcal{L})_{IJK}$ for its pull-back to \hat{M} . Differentiating (11.6) yields

$$\hat{\mathcal{L}}_{IJK} = \widehat{\nabla}_{I}g_{JK} = \widetilde{v} \left(\beta_{I}\rho^{*}(RH)_{JK} + \beta_{(J}\rho^{*}(RH)_{K)I} + \frac{1}{2}\widehat{\nabla}_{I}\rho^{*}(RH)_{JK} \right).$$
(11.11)

Since *R* is parallel, there holds $\hat{X}^I \hat{Y}^J \hat{Z}^K \widehat{\nabla}_I \rho^*(RH)_{JK} = \rho^*(RH(\mathcal{L}(X, Y), Z)),$ and there hold

$$\mathbb{E}^{I}\widehat{\nabla}_{I}\rho^{*}(RH)_{JK} = (\mathfrak{L}_{\mathbb{E}}\rho^{*}(RH))_{JK} - 2\rho^{*}(RH)_{JK} = -2\rho^{*}(RH)_{JK},$$
$$\mathbb{E}^{J}\widehat{\nabla}_{I}\rho^{*}(RH)_{JK} = -\rho^{*}(RH)_{IK},$$

so that the right-hand side of (11.11) is equal to $\frac{1}{2}\tilde{v}\rho^*(R\mathcal{L})_{IJK}$, showing

$$\hat{\mathcal{L}}_{IJK} = \frac{1}{2}\tilde{v}\rho^*(R\mathcal{L})_{IJK}.$$
(11.12)

There is a unique tensor \widetilde{H}^{IJ} such that for any covectors μ_i and ν_i there holds $\widetilde{H}^{IJ}\rho^*(\mu)_I\rho^*(\nu)_J = \widetilde{H}^{ij}\mu_i\nu_j$ and $\widetilde{H}^{IJ}\beta_J = 0$. Since $\widetilde{R}^{-1}\widetilde{H}^{IQ}\rho^*(RH)_{QJ} = \delta_I^{J} - \beta_I \mathbb{E}^J$ there holds

$$g^{IJ} = \tilde{v}^{-1} \left(\mathbb{E}^I \mathbb{E}^J + 2\tilde{R}^{-1} \tilde{H}^{IJ} \right).$$
(11.13)

It follows that

$$|\hat{\mathcal{L}}|_g^2 = \frac{1}{2}\tilde{v}^{-1}\rho^*(R^{-1}|\mathcal{L}|_H^2).$$
(11.14)

Using (11.12), $2\mathcal{L}_{ia}{}^{b}\mathcal{L}_{jb}{}^{a} = |\mathcal{L}|_{H}^{2}H_{ij}$, (11.6), (11.13), and (11.14) it follows straightforwardly that

$$\begin{aligned} 4\hat{\mathcal{R}}_{IJ} &= \hat{\mathcal{L}}_{IA}{}^{B}\hat{\mathcal{L}}_{JB}{}^{A} = \frac{1}{4}\tilde{v}^{2}\rho^{*}(R\mathcal{L})_{IA}{}^{B}\rho^{*}(R\mathcal{L})_{JB}{}^{A} \\ &= \frac{1}{2}\rho^{*}(|\mathcal{L}|_{H}^{2}H)_{IJ}, \\ 4\hat{\mathcal{R}}_{g}g_{IJ} &= |\hat{\mathcal{L}}|_{g}^{2}g_{IJ} = \rho^{*}(|\mathcal{L}|_{H}^{2}H)_{IJ} + 2\rho^{*}(R^{-1}|\mathcal{L}|_{H}^{2})\beta_{I}\beta_{J}, \quad (11.15) \\ \hat{\mathcal{R}}_{IJ} - \frac{1}{2}\hat{\mathcal{R}}_{g}g_{IJ} &= -\frac{1}{4}\rho^{*}(R^{-1}|\mathcal{L}|_{H}^{2})\beta_{I}\beta_{J} \\ &= -2^{-8/3}\tilde{R}^{-1/3}|\hat{\mathcal{L}}|_{H}^{2}N_{I}^{\flat}N_{J}^{\flat} = \hat{T}_{IJ}, \end{aligned}$$

in which $N^I = \tilde{u}^{-1} \mathbb{E}^I$ and $N_I^{\flat} = N^Q g_{IQ} = \tilde{u}\beta_I$. The last equation of (11.15) can be interpreted as an instance of the 2 + 1 dimensional general relativistic Einstein equations with a stress energy tensor corresponding to a pressureless perfect fluid (a *dust*), if N^I is viewed as the velocity field of the fluid and $-2^{-8/3}\tilde{R}^{-1/3}|\tilde{\mathcal{L}}|_H^2$ is viewed as its mass-energy density. Note that $\hat{T}_{IJ}U^IU^J \ge 0$ for all vector fields U^I on \hat{M} .

11.7. For C > 0 let $\hat{M}_C = \{x \in \hat{M} : F(x) > -\log c\}$. Define $\Psi(t)$ by

$$\Psi(t) = \int_{e^{-t/3}}^{C^{1/3}} (C - r^3)^{1/3} dr, \qquad (11.16)$$

for $t > -\log C$, and set $\phi = \Psi(F)$. Let $\phi_{II} = \widehat{\nabla}_I d\phi_I$. From

$$\phi_{IJ} = \dot{\Psi}(F)f_{IJ} + \ddot{\Psi}(F)F_IF_J, \qquad f^{JP}\phi_{IP} = \dot{\Psi}(F)\delta_I^{\ P} - \ddot{\Psi}(F)F_I\mathbb{E}^J,$$

it follows that ϕ_{IJ} is positive definite on \hat{M}_C . Noting that $\dot{\Psi} + 3\ddot{\Psi} = (Ce^t - 1)^{-1}\dot{\Psi}$ and using (11.8), that

$$\det \phi_{II} = 27 \dot{\Psi}(F)^2 (\dot{\Psi}(F) + 3 \ddot{\Psi}(F)) e^{2F} = 1.$$

This shows that ϕ_{IJ} is a Riemannian signature Monge-Ampère metric on \hat{M}_C .

Theorem 11.3. On a smooth surface M, let $([\nabla], [h])$ be an exact Riemannian signature Einstein AH structure with negative weighted scalar curvature R. For each C > 0 the metric $\phi_{IJ} = \widehat{\nabla}_I d\phi_J$ where $\phi(F) = \Psi(F)$ for Ψ defined by (11.16) and F is the function defined in terms of R as in Section 11.4 is a Riemannian signature Monge-Ampère metric on $\widehat{M}_C = \{x \in \widehat{M} : F(x) > -\log c\}$.

Theorem 11.3 is essentially the same as Proposition 1 of the unpublished erratum [57] to [56], and the method of construction, solving the ODE for Ψ that results from requiring $\Psi(F)$ to be Monge-Ampère, simply follows an example in [56, Section 2].

11.8. Here the constructions of Section 11.3 are used to show the convexity of the projective structure underlying an Einstein AH structure on a surface of genus g > 1. Let the notation be as in that section.

Theorem 11.4. Let $([\nabla], [h])$ be an exact Riemannian signature Einstein AH structure with negative weighted scalar curvature R on a smooth surface M. If a distinguished metric $h \in [h]$ is complete then the Riemannian metric f_{IJ} on \hat{M} is complete and $[\nabla]$ is a convex flat real projective structure; in particular this is the case if M is compact.

Proof. Note that $[\nabla]$ is projectively flat by Lemma 6.4. Let $h \in [h]$ be a distinguished metric. If h_{ij} is complete then $g_{ij} = 3P_{ij} = -(3R_h/2)h_{ij}$ is complete, and it follows from (11.6) that on \hat{M} the metric f_{IJ} has the form $\rho^*(g)_{IJ} + dt_I dt_J$ where $t = F/\sqrt{3}$. It is clear from this form that f_{IJ} is complete if h_{ij} is. [65, Theorem 2.1] shows that if a simply connected manifold admits a complete Hessian metric then its affine developing map is a diffeomorphism onto a convex domain Ω in flat affine space. Applying this to the given structures lifted to the universal cover \tilde{M} of M shows that the affine developing map of the Thomas connection is a diffeomorphism onto a convex domain in \mathbb{R}^3 . The function F is strictly convex and solves det $\hat{\nabla}_I dF_J = 27e^{2F}\Psi$. Transferring it to the image of the affine developing map of $\hat{\nabla}$ gives on Ω a function with the same properties, and it follows (see the argument given in the remark following the statement of [23, Theorem 1 on pages 357-358]) that Ω contains no complete affine line. From the equivariance of all the

preceeding with respect to the scaling action on \hat{M} it follows that Ω is a convex cone containing no complete affine line and the projective developing map of $[\nabla]$ is a diffeomorphism onto the oriented projectivization of Ω , which by the preceeding argument is a convex domain in the projective sphere the closure of which contains no pair of antipodal points (for if it did, the cone over it, which is Ω , would contain a complete affine line). This shows that $[\nabla]$ is a convex projective structure.

Theorem 11.5. If a surface M admits a convex flat real projective structure $[\nabla]$ then it admits a unique Riemannian signature conformal structure [h] such that $([\nabla], [h])$ is an exact Einstein AH structure with negative weighted scalar curvature and complete distinguished metric.

Proof. By assumption the pull-back of \hat{M} over the universal cover of M is identified by the affine developing map of the Thomas connection of $[\nabla]$ with a convex cone $\Omega \subset \mathbb{R}^3$ containing no complete affine line. By [23, Theorem 4.4] there is a smooth function F on Ω solving det $\widehat{\nabla}_I dF_I = 27e^{2F}\Psi^2$, tending to $+\infty$ at the boundary of Ω , and such that $f_{IJ} = \widehat{\nabla}_I dF_J$ is a complete Riemannian metric on the interior of Ω . Passing to the tube domain (in \mathbb{C}^3) over Ω and applying the generalized Schwarz lemma for volume forms proved in [60, Section 1] it can be deduced that e^F has positive homogeneity -3. Define the density u on M by $\tilde{u} = e^{-F/3}$ and let $h_{ii} = -u^{-1}\mathcal{B}(u)$. Tracing through the identifications in Section 11.3 backwards, it is straightforward to check that $([\nabla], [h])$ is an Einstein AH structure with parallel negative scalar curvature and distinguished metric h_{ii} . The completeness of h_{ii} follows from the splitting (11.6) and the completeness of f_{IJ} , as in the proof of Theorem 11.4. If [g] is another conformal structure with the same properties as [h], let G be the corresponding function on Ω , which has the same properties as has F. Passing to the tube domain over Ω and applying the Schwarz lemma from [60] shows that F = G, and so g and h are homothetic (here the completeness of both g and h is essential).

Theorem 11.5 shows that an exact Einstein AH structure with negative scalar curvature and complete distinguished metric is already completely determined by its underlying (necessarily flat) projective structure, which is convex. This correspondence is evidently diffeomorphism equivariant, and so combining Theorems 7.7, 9.7 and 11.5 there results Theorem 1.1.

Proof of Theorem 1.1. The only thing that perhaps requires comment is the genus 1 case. From Theorem 7.7 it follows that an exact Einstein AH structure on a torus is Weyl if and only if it is a flat conformal structure. The analogue in this case of Theorem 9.7 follows straightforwardly from Lemma 9.2.

11.9. Remark on tractor formalism

Projective structures are the simplest parabolic geometries, and the powerful general machinery (see [17]) applicable to such geometries should be useful in further understanding Einstein AH structures. In particular, it should be possible to give substance to the analogy between Einstein AH structures and extremal Kähler metrics using the cohomological ideas of [7] (implicitly relating certain conformal and projective BGG sequences), which have yet to be developed within the general parabolic geometry framework.

It should be mentioned that S. Armstrong has in [2] proposed a notion of Einstein structures for general parabolic geometries. Here is not the place for a careful examination of how his notion relates to that of Einstein AH structures, but it is briefly indicated how to pass from the formalism used in this paper to the tractor formalism of [16]. The homogeneities of \mathbb{E} , ∇ , g, and Ψ , are, respectively 0, 0, 2, and 3. These numbers are explained by passing to the tractor bundle $\mathcal{T} \to M$, which is the rank 3 vector bundle over M the total space of which is the quotient of $T\hat{M}$ by the action of \mathbb{R}^{\times} via $r^{-1}TR_r$. Sections of \mathcal{T} correspond to homogeneity -1 vector fields on \hat{M} , from which observation it is clear that ∇ induces a connection on \mathcal{T} (the tractor connection), and g and Ψ descend, respectively, to give a metric and a volume form on \mathcal{T} . The relation between the Thomas and tractor connections is explained in a bit more detail in [31, Section 3.1] or in [18, Section 2.3]; for background on the tractor formalism see [16]. The vector field \mathbb{E} does not descend to \mathcal{T} , but its span does, giving a distinguished line subbundle of \mathcal{T} . Since a choice of a non-vanishing density on M can be identified with a section of this distinguished line subbundle, it determines a splitting of \mathcal{T} as a direct sum of T M and the trivial line bundle. In much of the literature the objects just described appear defined in terms of such a splitting; for example the exposition in [49, Sections 2 and 3] is made in this way. Probably an appropriate reformulation of the Einstein AH condition would interpret the section of $S^2(\mathcal{T}^*)$ corresponding to g as harmonic.

12. Lagrangian immersions in (para)-Kähler space forms

12.1. Let (N, g, J) be a (para)-Kähler manifold, which means that J_i^{j} is an endomorphism satisfying $J_i^{p} J_p^{j} = \epsilon \delta_i^{j}$, where ϵ is -1 in the Kähler case and +1 in the para-Kähler case; g_{ij} is a metric, respectively Riemannian or split; and the tensor $\Omega_{ij} = J_i^{p} g_{pj}$ is a symplectic form. The definitions for para-Kähler manifolds of the Ricci form, the Einstein condition, etc. are formally identical to those in the Kähler case. A (para)-Kähler manifold has **constant (para)-holomorphic sectional curvature** 4c if its curvature has the form

$$R_{ijk}^{\ l} = 2c \left(\delta_{[i}^{\ l} g_{j]k} - \epsilon J_{[i}^{\ l} \Omega_{j]k} + \epsilon \Omega_{ij} J_k^{\ l} \right).$$
(12.1)

This is equivalent to the condition that for all $X \neq 0$ there holds

$$g(R(X, JX)X, JX) = 4c\epsilon g(X, X)^2.$$

Note that if the para-Kähler structure (g, J) has constant para-holomorphic sectional curvature 4c, then the para-Kähler structure (-g, -J), which has the same

underlying symplectic structure, has constant para-holomorphic sectional curvature -4c. An immersed submanifold of a para-Kähler manifold is **space-like** if the induced metric is positive definite. While the para-Kähler structures (g, J) and (-g, -J) appear similar, their space-like submanifolds are different.

If \mathbb{V} is a vector space with dual \mathbb{V}^* , a flat para-Kähler structure (\mathbb{G}, \mathbb{J}) on $\mathbb{V} \times \mathbb{V}^*$ is constituted by the symplectic form $\Omega((u, \mu), (v, v)) = \mu(v) - v(u)$ and the para-complex structure \mathbb{J} equaling the identity on \mathbb{V}^* and minus the identity on \mathbb{V} . The map $\Psi : \mathbb{V} \times \mathbb{V}^* \to \mathfrak{gl}(n + 1, \mathbb{R})^*$ given by $\langle \Psi(u, \mu), A \rangle = \mu(Au)$ is the moment map for the action of the general linear group $GL(n + 1, \mathbb{R})$ of \mathbb{V} on $\mathbb{V} \times \mathbb{V}^*$. The level sets of non-zero level of $\langle \Psi, \operatorname{Id} \rangle$ are the pseudo-spheres of constant \mathbb{G} norm. Their images in the quotient of $\{(u, \mu) \in \mathbb{V} \times \mathbb{V}^* : \mu(u) \neq 0\}$ by the action of the center of $GL(n+1, \mathbb{R})$ are the two components $\Sigma_{\pm} = \{([u], [\mu]) \in \mathbb{P}^+(\mathbb{V}) \times \mathbb{P}^+(\mathbb{V}^*) : \pm \mu(u) < 0\}$ of the complement of the incidence correspondence in $\mathbb{P}^+(\mathbb{V}) \times \mathbb{P}^+(\mathbb{V}^*)$, where \mathbb{P}^+ denotes oriented projectivization. The flat para-Kähler structure on $\mathbb{V} \times \mathbb{V}^*$ descends to Σ_{\pm} to give the model para-Kähler structures of constant para-holomorphic sectional curvature. This is formally parallel to the construction of the Fubini-Study metric on $\mathbb{P}^n(\mathbb{C})$ by reduction of the flat Kähler structure on complex Euclidean space via the Hopf fibration and the moment map for the action of U(n).

This can be understood in a more general context as follows. A **para-Hermitian symmetric space** is an affine symmetric space G/H with an almost para-Hermitian structure such that the symmetries act as automorphisms of the almost para-Hermitian structure. The almost para-Hermitian structure of a para-Hermitian symmetric space is necessarily para-Kähler, and G acts by para-Kähler automorphisms. This and other basic facts about these spaces are due to S.Kaneyuki and collaborators in a series of papers, from which there results:

Theorem 12.1 ([40,42,43]). Let G be a connected, semisimple Lie group and $H \subset G$ a closed subgroup. The following are equivalent

- (1) G/H is a homogeneous para-Kähler manifold.
- (2) *H* is an open subgroup of a Levi subgroup of a parabolic subgroup *P* of *G* having abelian nilradical.
- (3) G/H is a G-equivariant covering space of the adjoint orbit of a hyperbolic semisimple element of \mathfrak{g} .

Up to covering para-Hermitian symmetric spaces of semisimple Lie groups are in bijection with semsimimple graded Lie algebras $\mathfrak{g} = \mathfrak{g}_{-1} \oplus \mathfrak{g}_0 \oplus \mathfrak{g}_1$ in such a way that $\mathfrak{g} = \mathfrak{lie}(G)$ and $\mathfrak{g}_0 = \mathfrak{lie}(H)$.

In the setting of Theorem 12.1, G/H is diffeomorphic to the cotangent bundle of G/P, and the symplectic form on G/H is the pull-back of the Kostant-Kirillov symplectic form pulled back from the coadjoint orbit via some fixed multiple of the Killing form. The para-Hermitian symmetric spaces are Einstein. The proof is formally parallel to the proof that Hermitian symmetric spaces are Einstein (see [47, Proposition 9.7]). The scalar curvature is determined up to a scale factor determined by the choice of invariant symplectic form. The para-Hermitian symmetric space structure on the adjoint orbit of an element of $\mathfrak{sl}(n + 1, \mathbb{R})$ generating the center of the stabilizer in $SL(n + 1, \mathbb{R})$ of any element of $\Sigma_{\pm} \subset \mathbb{P}^+(\mathbb{V}) \times \mathbb{P}^+(\mathbb{V}^*)$ has constant non-zero para-holomorphic sectional curvature. This orbit is identified with the corresponding connected component Σ_{\pm} , and its para-Hermitian structure agrees up to constant factors with the model para-Kähler structure described above. As in Theorem 12.1 the components Σ_{\pm} are diffeomorphic to the cotangent bundle $T^*\mathbb{P}^+(\mathbb{V})$.

12.2. Let $i: M \to N$ be a Lagrangian immersion in the 2*n*-dimensional (para)-Kähler manifold (N, g, J), assumed space-like in the para-Kähler case. Via J_i^{j} the normal bundle to i(M) is identified with its tangent bundle, and this gives an identification of the second fundamental form $\Pi(X, Y)$ with the completely symmetric tensor Π_{ijk} on M defined by $\Pi(X, Y, Z) = \Omega(\Pi(X, Y), Z)$ for $X, Y, Z \in \Gamma(TM)$. Let $h_{ij} = i^*(g)_{ij}$ be the induced metric, let $H^i = h^{ia} \Pi_{apq} h^{pq}$ be the vector field dual to the mean curvature one-form (which is the one-form identified with the mean curvature vector using the (para)-Kähler structure), and let $B_{ijk} = \Pi_{ijk} - \frac{3}{n+2}H_{(i}h_{jk})$ be the completely trace-free part of the second fundamental form. In the Kähler case, a lemma of P. Dazord, [27], shows that dH_{ij}^{b} is the pull-back via *i* of the Ricci form, and the same statement is true in the para-Kähler case, with a formally identical proof. It follows that if *g* is Einstein, then $dH_{ij}^{b} = 0$; in particular this is true if *g* has constant (para)-holomorphic sectional curvature 4c, the Gauß-Codazzi equations yield

$$\begin{aligned} &\mathcal{R}_{h} - 2c - \epsilon |B|_{h}^{2} + \frac{\epsilon}{4} |H|_{h}^{2} = 0, \\ &4 \operatorname{div}_{h}(B)_{ij} = \operatorname{tf}_{h}(\mathfrak{L}_{H}h)_{ij}, \quad 2D_{[i}B_{j]kl} = h_{k[i}\operatorname{div}_{h}(B)_{j]l} + h_{l[i}\operatorname{div}_{h}(B)_{j]k}, \end{aligned}$$
(12.2)

the last of which is vacuous by (3.6). Say that the immersion is **CKMC** (has **conformal Killing mean curvature**) if $\text{tf}_h(\mathfrak{L}_H h)_{ij} = 0$. By (12.2) and Lemma 3.5, the immersion is CKMC if and only if both $B^{(3,0)}$ and $H^{(1,0)}$ are holomorphic. In particular, a space-like Lagrangian immersion of a surface of genus g > 1 is CKMC if and only if it has mean curvature 0.

Suppose now that the immersion has mean curvature 0 so that $H^i = 0$, and define $\nabla = D - B_{ijp}h^{kp}$. Then $\nabla_i h_{jk} = 2B_{ijk}$, so ∇ generates with [h] an AH structure ($[\nabla], [h]$). Since $h^{pq}\nabla_i h_{pq} = 0 = 2h^{pq}\nabla_p h_{qi}$, ∇ is the aligned representative of ($[\nabla], [h]$), which is exact, and has h as a distinguished representative. From (5.8) and (12.2) it follows that the curvature R_h is

$$\mathbf{R}_{h} = \Re_{h} - |B|_{h}^{2} = 2c + (\epsilon - 1)|B|_{h}^{2}.$$
 (12.3)

In the para-Kähler case $\epsilon = 1$, so $R_h = 2c$ is a constant. Hence the weighted scalar curvature of $([\nabla], [h])$ is parallel, and $([\nabla], [h])$ is Einstein. This proves:

Theorem 12.2. On a mean curvature zero space-like Lagrangian immersion of a surface in a four dimensional para-Kähler manifold of constant para-holomorphic

sectional curvature 4c there is induced an exact Riemannian Einstein AH structure with scalar curvature 2c.

Theorem 12.2 is the n = 2 special case of [30, Theorem 8.4]. R. Hildebrand's [38, Theorem 4.3] is essentially equivalent to Theorem 12.2, although it is stated using different terminology.

Corollary 12.3.

- (1) In a 4-dimensional para-Kähler manifold of constant negative para-holomorphic sectional curvature, there is no mean curvature zero space-like Lagrangian immersion of a two sphere.
- (2) In a 4-dimensional para-Kähler manifold of constant positive para-holomorphic sectional curvature, there is no mean curvature zero space-like Lagrangian immersion of a compact orientable surface of genus greater than one.
- (3) In a flat 4-dimensional para-Kähler manifold, a mean curvature zero space-like Lagrangian immersion of a compact orientable surface is a totally geodesic Lagrangian immersion of a flat torus.
- (4) In the flat para-K\u00e4hler space V×V*, there is no mean curvature zero space-like Lagrangian immersion of a compact orientable surface.

Proof. By Theorem 7.7 there cannot be an exact Einstein AH structure with negative scalar curvature on the two-sphere, there cannot be an exact Einstein AH structure with positive scalar curvature on a compact orientable surface of genus greater than one, and an exact Einstein AH structure with vanishing scalar curvature on a compact orientable surface is necessarily that generated by a flat metric on a torus. This proves (12.3)-(12.3). As the geodesics in the flat para-Káhler structure on $\mathbb{V} \times \mathbb{V}^*$ are affine lines, which are contained in no compact subset, there can be no such immersion in $\mathbb{V} \times \mathbb{V}^*$.

The claims of Corollary 12.3 are not the strongest results of this sort possible. For example, conclusion (4) of Corollary 12.3 follows from the much stronger J. Jost and Y. L Xin's [41, Theorem 4.2], which generalizes a Bernstein type theorem of Jörgens-Calabi-Pogorelov:

Theorem 12.4 ([41]). If the image of a mean curvature zero space-like Lagrangian immersion in the flat para-Kähler space $\mathbb{V} \times \mathbb{V}^*$ is closed then it is an affine subspace.

It would be interesting to know which exact Einstein AH structures on surfaces can be realized as in Corollary 12.3 as immersed or embedded mean curvature zero Lagrangian submanifolds of para-Kähler space forms. In fact, there is a way to associate to an exact Einstein AH structure on a surface M of genus g > 1 a spacelike Lagrangian immersion of its universal cover \tilde{M} in a para-Kähler manifold of constant para-holomorphic sectional curvature. This is now sketched. It is planned to report the details elsewhere, although, since the first version of this paper appeared, Hildebrand has describe, in [38] and [39], a closely related correspondence between centro-affine hypersurfaces and Lagrangian submanifolds of para-Kähler manifolds; in particular, Theorem 4.1 and [38, Corollary 4.2] yield conclusions very similar to those of the following paragraph.

Because it is integrable, the horizontal subbundle of the cotangent bundle determined by a flat affine connection constitutes with the vertical subbundle a pair of transverse foliations which are Lagrangian with respect to the tautological symplectic structure, so determine on the cotangent bundle a pair of para-Kähler structures distinguished by the choice of the vertical or the horizontal subbundle as the +1eigensubbundle of the para-complex structure. Parallel transport by the flat affine connection $\widehat{\nabla}$ on \mathbb{V} determines an identification $T^*\mathbb{V} \simeq \mathbb{V} \oplus \mathbb{V}^*$ under which the horizontal (respectively vertical) subbundle is sent to that corresponding to $\mathbb{V} \times \{0\}$ (resp $\{0\} \times \mathbb{V}^*$). Under this identification, Ω corresponds to twice the tautological symplectic form on $T^*\mathbb{V}$, the para-complex structure \mathbb{J} corresponds to the choice of the the vertical subbundle of $T^*\mathbb{V}$ as the +1-eigensubbundle, and the graph Γ_β of a closed one-form β on \mathbb{V} is identified with a Lagrangian submanifold of $\mathbb{V} \times \mathbb{V}^*$ which is conical (preserved by positive dilations) if and only if β has homogeneity 2. The flat para-Kähler structure on the cotangent bundle determined by the choice of the vertical subbundle as the +1 eigenbundle of the para-complex structure has the property that the pull-back $\beta^*(\mathbb{G})$ via the closed one-form β of the resulting para-Kähler metric is $2\widehat{\nabla}\beta$. If this is non-degenerate, then the second fundamental form of Γ_{β} is $\widehat{\nabla}\widehat{\nabla}\beta$, and its mean curvature is the logarithmic covariant derivative of det $\widehat{\nabla}\beta$. In particular, Γ_{β} has mean curvature zero if and only if det $\widehat{\nabla}\beta$ is parallel; if β is the differential of a positive homogeneity 2 function \tilde{v} , this is equivalent to \tilde{v} solving the Monge-Ampère equation (11.7). Mean curvature zero immersed Lagrangian submanifolds of Σ_{\pm} correspond to mean curvature zero immersed conical Lagrangian submanifolds of $\mathbb{V} \times \mathbb{V}^*$. This is formally parallel to the correspondence between minimal Lagrangian immersions in complex projective space and minimal Lagrangian cones in complex Euclidean space recounted in Section 2 of [59]. The Thomas connection of the convex flat projective structure determined by the lift to \tilde{M} of the given Einstein AH structure on M is identified with the restriction to a proper open convex cone in \mathbb{V} of the standard flat affine connection ∇ on \mathbb{V} . On this cone there is, as in Section 11.4 and the proofs of Theorems 11.4 and 11.5, the positive homogeneity 2 solution \tilde{v} of the Monge-Ampère equation (11.7). The graph of the one-form $d\tilde{v}$ is a conical space-like mean curvature zero Lagrangian submanifold which covers the desired mean curvature zero Lagrangian submanifold of the para-Kähler space form. The induced Einstein AH structure coincides with the original one on \tilde{M} .

12.3. Given a background metric \tilde{h}_{ij} , a cubic holomorphic differential $B^{(3,0)}$, and a holomorphic vector field $H^{(1,0)}$, it makes sense to look for a conformal metric $h_{ij} = e^{\phi} \tilde{h}_{ij}$ such that (h, B, H) are as for the induced tensors on a CKMC Lagrangian immersion in a (para)-Kähler manifold (N, g, J). There results the equation

$$\Delta_{\tilde{h}}\phi - \mathcal{R}_{\tilde{h}} + 2ce^{\phi} - \frac{\epsilon}{4}e^{2\phi}|H|_{\tilde{h}}^2 + \epsilon e^{-2\phi}|B|_{\tilde{h}}^2 = 0.$$
(12.4)

The solution of the case of (12.4) in which $\epsilon = +1$ and c < 0 has been described for compact surfaces in Section 9. In the Kähler ($\epsilon = -1$) case, (12.4) should be interesting for both signs of c, corresponding to the complex projective plane and the complex hyperbolic plane. One wonders whether for Lagrangian immersions in a complex hyperbolic 4-manifold there is a deformation space of solutions like that for mean curvature zero surfaces in three-dimensional hyperbolic space studied by C. Taubes in [66]. After the first version of this paper had been completed there appeared the preprint [50], in which question of this sort are treated in detail for the c = -1 case of (12.4). For example, after appropriate changes of notation, of [50, Theorem 4.1] is:

Lemma 12.5 ([50]). If *M* is a compact orientable surface of genus at least two and \tilde{h} is a metric of constant scalar curvature -2 on *M* then for every cubic holomorphic differential $B^{(3,0)}$ such that there holds everywhere on *M* the bound $|B|_{\tilde{h}}^2 \leq 8/27$ there is a solution ϕ to the equation (12.4) with parameters $\epsilon = -1$ and c = -1 satisfying $0 \geq \phi \geq \log 2 - \log 3$.

Proof. Since the short proof is just like the proof of Lemma 9.4, it is convenient to give it here. Clearly 0 is a supersolution of $\mathcal{A}(\phi) = \Delta_{\tilde{h}}\phi + 2 - 2e^{\phi} - e^{-2\phi}|B|_{\tilde{h}}^2$. If c is any constant, then $e^{2c}\mathcal{A}(c) \ge -2p(e^c)$ where $p(r) = r^3 - r^2 + \frac{1}{2}\max_M |B|_{\tilde{h}}^2$. This polynomial p is non-negative at r = 0 and has at r = 2/3 a local minimum at which its value is $\frac{1}{2}\max_M |B|_{\tilde{h}}^2 - 4/27$. Hence p has a positive zero if and only if $\max_M |B|_{\tilde{h}}^2 \le 8/27$, in which case its smallest positive zero r_1 is no greater than 2/3. In this case $\mathcal{A}(\log r_1) \ge 0$, so $\log r_1$ is a negative subsolution of \mathcal{A} . As in the proof of Lemma 9.4 this suffices to show the existence of a solution to $\mathcal{A}(\phi) = 0$ satisfying the indicated bounds.

More work has to be done to construct from such a solution a minimal Lagrangian immersion in the complex hyperbolic plane, and this is a part of what is accomplished in [50]. A similar analysis in the para-Kähler case should be interesting as well.

References

- J. A. ALEDO, J. M. ESPINAR and J. A. GÁLVEZ, *The Codazzi equation for surfaces*, Adv. Math. 224 (2010), 2511–2530.
- S. ARMSTRONG, Einstein connections and involutions via parabolic geometries, J. Geom. Phys. 60 (2010), 1424–1440.
- [3] A. BANYAGA, "The Structure of Classical Diffeomorphism Groups", Mathematics and its Applications, Vol. 400, Kluwer Academic Publishers Group, Dordrecht, 1997.
- [4] Y. BENOIST, "Convexes divisibles. I", Algebraic groups and arithmetic, Tata Inst. Fund. Res., Mumbai, 2004, 339–374.
- [5] S. B. BRADLOW, Vortices in holomorphic line bundles over closed Kähler manifolds, Comm. Math. Phys. 135 (1990), 1–17.

- [6] E. CALABI, Improper affine hyperspheres of convex type and a generalization of a theorem by K. Jörgens, Michigan Math. J. 5 (1958), 105–126.
- [7] E. CALABI, On compact, Riemannian manifolds with constant curvature. I, Proc. Sympos. Pure Math., Vol. III, American Mathematical Society, Providence, R.I., 1961, 155–180.
- [8] E. CALABI, Examples of Bernstein problems for some nonlinear equations, Global Analysis (Proc. Sympos. Pure Math., Vol. XV, Berkeley, Calif., 1968), Amer. Math. Soc., Providence, R.I., 1970, 223–230.
- [9] E. CALABI, *Complete affine hyperspheres. I*, Symposia Mathematica, Vol. X (Convegno di Geometria Differenziale, INDAM, Rome, 1971), Academic Press, London, 1972, 19–38.
- [10] E. CALABI, Extremal Kähler metrics, In: "Seminar on Differential Geometry", Ann. of Math. Stud., Vol. 102, Princeton Univ. Press, Princeton, N.J., 1982, 259–290.
- [11] E. CALABI, Hypersurfaces with maximal affinely invariant area, Amer. J. Math. 104 (1982), 91–126.
- [12] E. CALABI, *Convex affine maximal surfaces*, Results Math. 13 (1988), 199–223, reprinted in: "Affine Differentialgeometrie" [(Oberwolfach, 1986), 199–223, Tech. Univ. Berlin, Berlin, 1988].
- D. M. J. CALDERBANK, Möbius structures and two-dimensional Einstein-Weyl geometry, J. Reine Angew. Math. 504 (1998), 37–53.
- [14] D. M. J. CALDERBANK, The Faraday 2-form in Einstein-Weyl geometry, Math. Scand. 89 (2001), 97–116.
- [15] D. M. J. CALDERBANK, Two dimensional Einstein-Weyl structures, Glasg. Math. J. 43 (2001), 419–424.
- [16] A. ČAP and A. R. GOVER, *Tractor calculi for parabolic geometries*, Trans. Amer. Math. Soc. 354 (2002), 1511–1548.
- [17] A. ČAP and J. SLOVÁK, "Parabolic Geometries. I. Background and General Theory", Mathematical Surveys and Monographs, Vol. 154, American Mathematical Society, Providence, RI, 2009.
- [18] A. ČAP and V. ŽÁDNÍK, Contact projective structures and chains, Geom. Dedicata 146 (2010), 67–83.
- [19] X. CHEN, P. LU and G. TIAN, A note on uniformization of Riemann surfaces by Ricci flow, Proc. Amer. Math. Soc. **134** (2006), 3391–3393.
- [20] S. Y. CHENG and S. T. YAU, Maximal space-like hypersurfaces in the Lorentz-Minkowski spaces, Ann. of Math. (2) 104 (1976), 407–419.
- [21] S. Y. CHENG and S. T. YAU, On the regularity of the Monge-Ampère equation $det(\partial^2 u/\partial x_i \partial x_i) = F(x, u)$, Comm. Pure Appl. Math. **30** (1977), 41–68.
- [22] S. Y. CHENG and S. T. YAU, On the existence of a complete Kähler metric on noncompact complex manifolds and the regularity of Fefferman's equation, Comm. Pure Appl. Math. 33 (1980), 507–544.
- [23] S. Y. CHENG and S. T. YAU, *The real Monge-Ampère equation and affine flat structures*, Proceedings of the 1980 Beijing Symposium on Differential Geometry and Differential Equations, Vol. 1, 2, 3 (Beijing, 1980) (Beijing), Science Press, 1982, 339–370.
- [24] S. Y. CHENG and S. T. YAU, Complete affine hypersurfaces. I. The completeness of affine metrics, Comm. Pure Appl. Math. 39 (1986), 839–866.
- [25] S. CHOI and W. M. GOLDMAN, The deformation spaces of convex ℝP²-structures on 2orbifolds, Amer. J. Math. 127 (2005), 1019–1102.
- [26] R. COURANT and D. HILBERT, "Methods of mathematical physics. Vol. II: Partial Differential Equations, (Vol. II by R. Courant.), Interscience Publishers (a division of John Wiley & Sons), New York-London, 1962.
- [27] P. DAZORD, Sur la géométrie des sous-fibrés et des feuilletages lagrangiens, Ann. Sci. École Norm. Sup. (4) 14 (1981), 465–480.
- [28] C. J. EARLE and J. EELLS, A fibre bundle description of Teichmüller theory, J. Differential Geom. 3 (1969), 19–43.

- [29] V. A. FATEEV, E. ONOFRI AND A. B. ZAMOLODCHIKOV, Integrable deformations of the O(3) sigma model. The sausage model, Nuclear Phys. B 406 (1993), 521–565.
- [30] D. J. F. FOX, Geometric structures modeled on affine hypersurfaces and generalizations of the Einstein Weyl and affine hypersphere equations, arXiv:0909.1897.
- [31] D. J. F. FOX, Contact projective structures, Indiana Univ. Math. J. 54 (2005), 1547– 1598.
- [32] D. J. F. FOX, Commutative, nonassociative algebras with non-degenerate, invariant Killing form, and the associated cubic polynomials, preprint, 2010.
- [33] Ó. GARCÍA-PRADA, A direct existence proof for the vortex equations over a compact Riemann surface, Bull. London Math. Soc. 26 (1994), 88–96.
- [34] P. GAUDUCHON, La 1-forme de torsion d'une variété hermitienne compacte, Math. Ann. 267 (1984), 495–518.
- [35] P. GAUDUCHON, Structures de Weyl-Einstein, espaces de twisteurs et variétés de type $S^1 \times S^3$, J. Reine Angew. Math. **469** (1995), 1–50.
- [36] W. M. GOLDMAN, Convex real projective structures on compact surfaces, J. Differential Geom. 31 (1990), 791–845.
- [37] F. HERRLICH and G. SCHMITHÜSEN, On the boundary of Teichmüller disks in Teichmüller and in Schottky space, Handbook of Teichmüller theory. Vol. I, IRMA Lect. Math. Theor. Phys., Eur. Math. Soc., Zürich 11 (2007) 293–349.
- [38] R. HILDEBRAND, The cross-ratio manifold: a model of centro-affine geometry, Int. Electron. J. Geom. 4 (2011), 32–62.
- [39] R. HILDEBRAND, Half-dimensional immersions in para-Kähler manifolds, Int. Electron. J. Geom. 4 (2011), 85–113.
- [40] Z. HOU, S. DENG, S. KANEYUKI and K. NISHIYAMA, *Dipolarizations in semisimple Lie algebras and homogeneous para-Kähler manifolds*, J. Lie Theory **9** (1999), 215–232.
- [41] J. JOST and Y. L. XIN, Some aspects of the global geometry of entire space-like submanifolds, Results Math. 40 (2001), 233–245.
- [42] S. KANEYUKI, Compactification of parahermitian symmetric spaces and its applications. II. Stratifications and automorphism groups, J. Lie Theory **13** (2003), 535–563.
- [43] S. KANEYUKI and M. KOZAI, *Paracomplex structures and affine symmetric spaces*, Tokyo J. Math. **8** (1985), 81–98.
- [44] J. L. KAZDAN and F. W. WARNER, Curvature functions for compact 2-manifolds, Ann. of Math. (2) 99 (1974), 14–47.
- [45] S. KERCKHOFF, H. MASUR and J. SMILLIE, *Ergodicity of billiard flows and quadratic differentials*, Ann. of Math. (2) **124** (1986), 293–311.
- [46] S. KOBAYASHI, The first Chern class and holomorphic symmetric tensor fields, J. Math. Soc. Japan 32 (1980), 325–329.
- [47] S. KOBAYASHI and K. NOMIZU, "Foundations of differential geometry", Vol. II, Interscience Tracts in Pure and Applied Mathematics, n. 15 Vol. II, Interscience Publishers John Wiley & Sons, Inc., New York-London-Sydney, 1969.
- [48] M. KONTSEVICH and Y. SOIBELMAN, *Homological mirror symmetry and torus fibrations*, In: "Symplectic Geometry and Mirror Symmetry" (Seoul, 2000), World Sci. Publ., River Edge, NJ, 2001, 203–263.
- [49] F. LABOURIE, Flat projective structures on surfaces and cubic holomorphic differentials, Pure Appl. Math. Q. 3 (2007), 1057–1099.
- [50] J. LOFTIN and I. MCINTOSH, Minimal Lagrangian surfaces in CH² and representations of surface groups into SU(2, 1), Geom. Dedicata 162 (2013), 67–93.
- [51] J. C. LOFTIN, Affine spheres and convex \mathbb{RP}^n -manifolds, Amer. J. Math. **123** (2001), 255–274.
- [52] J. C. LOFTIN, Riemannian metrics on locally projectively flat manifolds, Amer. J. Math. 124 (2002), 595–609.
- [53] J. C. LOFTIN, *The compactification of the moduli space of convex* ℝP² surfaces. *I*, J. Differential Geom. 68 (2004), 223–276.

- [54] J. C. LOFTIN, Flat metrics, cubic differentials and limits of projective holonomies, Geom. Dedicata 128 (2007), 97–106.
- [55] J. C. LOFTIN, Survey on affine spheres, In: "Handbook of Geometric Analysis", Vol. II, Adv. Lect. Math. (ALM), Int. Press, Somerville, MA, 13 (2010), 161–192, arXiv:0809.1186.
- [56] J. C. LOFTIN, S. T. YAU and E. ZASLOW, Affine manifolds, SYZ geometry and the "Y" vertex, J. Differential Geom. 71 (2005), 129–158.
- [57] J. C. LOFTIN, S. T. YAU and E. ZASLOW, Erratum to affine manifolds, SYZ geometry and the "Y" vertex, Loftin's web page, 2008.
- [58] H. MASUR and S. TABACHNIKOV, *Rational billiards and flat structures*, In: "Handbook of Dynamical Systems", Vol. 1A, North-Holland, Amsterdam, 2002, 1015–1089.
- [59] I. MCINTOSH, Special Lagrangian cones in C³ and primitive harmonic maps, J. London Math. Soc. (2) 67 (2003), 769–789.
- [60] N. MOK and S. T. YAU, Completeness of the Kähler-Einstein metric on bounded domains and the characterization of domains of holomorphy by curvature conditions, In: "The Mathematical Heritage" of Henri Poincaré, Part 1 (Bloomington, Ind., 1980), Proc. Sympos. Pure Math., Amer. Math. Soc., Providence, RI 39 (1983) 41–59.
- [61] M. NOGUCHI, Yang-Mills-Higgs theory on a compact Riemann surface, J. Math. Phys. 28 (1987), 2343–2346.
- [62] K. NOMIZU and F. PODESTÀ, On affine Kähler structures, Bull. Soc. Math. Belg. Sér. B 41 (1989), 275–282.
- [63] K. NOMIZU and T. SASAKI, "Affine Differential Geometry", Cambridge Tracts in Mathematics, Vol. 111, Cambridge University Press, Cambridge, 1994.
- [64] R. SCHOEN and S. T. YAU, "Lectures on Differential Geometry", Conference Proceedings and Lecture Notes in Geometry and Topology, I, International Press, Cambridge, MA, 1994.
- [65] H. SHIMA and K. YAGI, Geometry of Hessian manifolds, Differential Geom. Appl. 7 (1997), 277–290.
- [66] C. H. TAUBES, *Minimal surfaces in germs of hyperbolic 3-manifolds*, Proceedings of the Casson Fest, Geom. Topol. Monogr., Geom. Topol. Publ., Coventry 7 (2004), 69–100.
- [67] T. Y. THOMAS, "The differential invariants of generalized spaces", Chelsea Publishing Company, New York, 1934.
- [68] K. P. TOD, Compact 3-dimensional Einstein-Weyl structures, J. London Math. Soc. (2) 45 (1992), 341–351.
- [69] M. TROYANOV, Les surfaces euclidiennes à singularités coniques, Enseign. Math. (2) 32 (1986), 79–94.
- [70] M. TROYANOV, On the moduli space of singular Euclidean surfaces, In: "Handbook of Teichmüller Theory", Vol. I, IRMA Lect. Math. Theor. Phys., Eur. Math. Soc., Zürich 11 (2007), 507–540.
- [71] N. S. TRUDINGER and X.-J. WANG, *The Monge-Ampère equation and its geometric applications*, In: "Handbook of Geometric Analysis", n. 1, Adv. Lect. Math. (ALM), Vol. 7, Int. Press, Somerville, MA, 2008, 467–524.
- [72] M. VIANA, Ergodic theory of interval exchange maps, Rev. Mat. Complut. 19 (2006), 7– 100.
- [73] A. VITTER, Affine structures on compact complex manifolds, Invent. Math. 17 (1972), 231– 244.
- [74] L. VRANCKEN, Centroaffine differential geometry and its relations to horizontal submanifolds, In: "PDEs, Submanifolds and Affine Differential Geometry" (Warsaw, 2000), Banach Center Publ., Polish Acad. Sci., Warsaw 57 (2002), 21–28.
- [75] C. P. WANG, Some examples of complete hyperbolic affine 2-spheres in R³, In: "Global Differential Geometry and Global Analysis" (Berlin, 1990), Lecture Notes in Math., Vol. 1481, Springer, Berlin, 1991, 271–280.

- [76] H. WEYL, "The classical groups. Their invariants and representations", Princeton University Press, Princeton, N.J., 1939.
- [77] M. WOLF, The Teichmüller theory of harmonic maps, J. Differential Geom. 29 (1989), 449–479.

Departamento de Matemática Aplicada EUIT Industrial Universidad Politécnica de Madrid Ronda de Valencia 3 28012 Madrid, Spain daniel.fox@upm.es